INFORMATION TO USERS

This manuscript has been reproduced from the microfilm master. UMI films the text directly from the original or copy submitted. Thus, some thesis and dissertation copies are in typewriter face, while others may be from any type of computer printer.

The quality of this reproduction is dependent upon the quality of the copy submitted. Broken or indistinct print, colored or poor quality illustrations and photographs, print bleedthrough, substandard margins, and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send UMI a complete manuscript and there are missing pages, these will be noted. Also, if unauthorized copyright material had to be removed, a note will indicate the deletion.

Oversize materials (e.g., maps, drawings, charts) are reproduced by sectioning the original, beginning at the upper left-hand corner and continuing from left to right in equal sections with small overlaps.

Photographs included in the original manuscript have been reproduced xerographically in this copy. Higher quality 6" x 9" black and white photographic prints are available for any photographs or illustrations appearing in this copy for an additional charge. Contact UMI directly to order.

ProQuest Information and Learning 300 North Zeeb Road, Ann Arbor, MI 48106-1346 USA 800-521-0600

UM®

University of Alberta

The Dimensional-Reduction Anomaly

by

Patrick James Sutton



A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Department of Physics

Edmonton, Alberta

Fall 2000



National Library of Canada

Acquisitions and Bibliographic Services

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque nationale du Canada

Acquisitions et services bibliographiques

395, rue Wellington Ottawa ON K1A 0N4 Canada

Your file. Votre rélérance

Our lile Notre rélérance

The author has granted a nonexclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission. L'auteur a accordé une licence non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

0-612-59681-8

Canadä

University of Alberta

Library Release Form

NAME OF AUTHOR:	Patrick James Sutton
TITLE OF THESIS:	The Dimensional-Reduction Anomaly
Degree:	Doctor of Philosophy

YEAR THIS DEGREE GRANTED: 2000

Permission is hereby granted to the University of Alberta Library to reproduce single copies of this thesis and to lend or sell such copies for private, scholarly or scientific research purposes only.

The author reserves all other publication and other rights in association with the copyright in the thesis, and except as hereinbefore provided neither the thesis nor any substantial portion thereof may be printed or otherwise reproduced in any material form whatever without the author's prior written permission.

Patrick James Inter

Patrick James Sutton Physics Department University of Alberta Edmonton, Alberta T6G 2J1

20 September 2000

University of Alberta

Faculty of Graduate Studies and Research

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research for acceptance, a thesis entitled The Dimensional-Reduction Anomaly submitted by Patrick James Sutton in partial fulfillment of the requirements for the degree of Doctor of Philosophy.

Prof. V. P. Frolov (Supervisor)

Fare Im.

Prof. D. N. Page (Ghair) an

Prof. R. B. Mann (External)

BA Jampbell

Prof. B. A. Campbell

Prof. M. Razavy

Prof. H. Kunzle

DATE: September 8: 2000

Abstract

In *D*-dimensional spacetimes which can be foliated by *n*-dimensional homogeneous subspaces, a quantum field can be decomposed in terms of modes on the subspaces. Substitution into the field equations reduces the free quantum field in *D* dimensions to a collection of quantum fields in D - n dimensions. This dimensional reduction establishes a formal relationship whereby objects in the physical theory, such as the Green function and the effective action, can be written as sums of the corresponding objects in the dimensionally reduced theories.

In this thesis we examine dimensional reduction in the context of renormalization. Quantities of physical interest in quantum field theory, such as the expectation value of the stress tensor, are divergent and must be renormalized. Though the equivalence of the original and dimensionally-reduced theories is easily established at the bare level, we demonstrate that the divergent terms which must be subtracted to renormalize the higher- and lower-dimensional theories are not related by the mode decomposition. As a result, renormalized expectation values in D dimensions can be obtained by summing over their renormalized (D - n)-dimensional counterparts only if the

contribution of each mode is modified by an extra anomalous contribution. We call this effect the dimensional-reduction anomaly.

We explicitly calculate the dimensional-reduction anomaly in the field fluctuations and the stress tensor for several classes of spacetimes of physical interest, with particular emphasis on spherical and static spaces. In each case, the anomaly is shown to produce significant mode-by-mode corrections to renormalized expectation values in the dimensionally reduced theories. For spherical geometries we investigate the relevance of the anomaly to recent attempts to calculate the stress tensor and Hawking radiation from Schwarzschild black holes using two-dimensional dilaton-gravity models. For static spaces we find an intriguing relationship between the anomaly and a new, general approximation scheme for renormalized expectation values of quantum fields.

То

Catherine

Preface

The research presented here was performed by the author, partially in collaboration with V. P. Frolov and A. I. Zelnikov, at the University of Alberta between November 1998 and July 2000. Unless otherwise noted, all of the material in this report is original. Much of it has been published in the following articles:

- P. Sutton, "Dimensional-reduction anomaly in spherically symmetric spacetimes". Physical Review D 62 044033 (2000).
- V. Frolov, P. Sutton, and A. Zelnikov, "Dimensional-reduction anomaly". Physical Review D 61 024021 (2000).
- P. J. Sutton, "Dimensional Reduction of the Effective Action and the Multiplicative Anomaly". Proceedings of the Eighth Canadian Conference on General Relativity and Relativistic Astrophysics, pp. 257-261 (American Institute of Physics, New York, 1999).

These results have also been presented at the Second International School on Field Theory and Gravitation (April 25-28, 2000, Vitoria, Brasil) and at the Fradkin Memorial meeting (June 5-10, 2000, Moscow, Russia).

Acknowledgements

This thesis could not have come to fruition without the help of many people. In particular, I would like to thank my supervisor Valeri Frolov and my colleague Andrei Zelnikov, whose guidance and friendship have been much appreciated. I would also like to thank the many friends who have made the past five years so enjoyable, including Andrew Brown, Anthony Faust, Dave Lamb, Shaun Hendy, Bruce Campbell, Dave Sept, Warren Anderson, Jason Myatt, J. P. DeVilliers, Bahman Darian, Yuri Gusev, Ron Whalen, Kimphrey Tu, Connell McCluskey, the Irish guys and the other football hooligans, Ellen Crawford, Dave Maybury, Jason Middleton, Norm Buchanan, Kristen Beaty, Dave Shaw, Alasdair Syme, and especially Kipp Cannon.

I would also like to express my gratitude to the Natural Sciences and Engineering Research Council (NSERC) and the University of Alberta for their generous financial support of my studies.

Special thanks go to my family, particularly my parents Denis and Cathy, for their love, support, and encouragement. I could not ask for finer examples by which to live my life.

Most of all, I thank my wife Catherine. She has been my light and my love, a pillar of faith, and an unending source of support and encouragement throughout these many years. All that I have and all that I am I owe to her.

Table of Contents

A	bstra	ict		
P	refac	e		
A	ckno	wledge	ements	
Ta	able	of Con	tents	
1	Intr	oducti	ion	1
2	Qua	antum	Field Theory in Curved Spacetime and Dimensional	
	Rec	luction	1	5
	2.1	Quant	um Field Theory in Curved Spacetime	6
		2.1.1	Classical Field	6
		2.1.2	Canonical Quantization	8
		2.1.3	Vacuum States	9
		2.1.4	Green Functions	11
		2.1.5	Effective Action	12
		2.1.6	Heat Kernel	14
		2.1.7	Renormalization	15
		2.1.8	Euclidean Approach	16
	2.2	Dimer	asional Reduction	18

TABLE OF CONTENTS

		2.2.1	Spacetime Metric and Notations	19
		2.2.2	Quantum Field in D Dimensions $\ldots \ldots \ldots \ldots \ldots$	20
		2.2.3	Mode Decompositions	21
		2.2.4	Dimensional Reduction of $\langle \hat{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu\nu} \rangle$	26
		2.2.5	Dimensional Reduction of the Conservation Equation	29
		2.2.6	Induced Boundary Conditions	30
		2.2.7	Dilaton-Gravity Models	30
	2.3	The D	Dimensional-Reduction Anomaly	33
3	Sph	erical	Decompositions in Flat Space	36
	3.1	Spheri	ical Dimensional Reductions	37
	3.2	s-mod	le in Flat Space at Finite Temperature	39
		3.2.1	Formalism	40
		3.2.2	s-mode Contribution to $\langle \hat{\Phi}^2 \rangle^{eta}$	42
		3.2.3	s-mode Contribution to $\langle \hat{T}_{\mu u} angle^{eta}$	45
	3.3	The D	Dimensional-Reduction Anomaly in Flat Space	50
		3.3.1	Scalar Field at Finite Temperature	56
4	Sph	erical	Decompositions in Curved Space	59
	4.1	The E)imensional-Reduction Anomaly in $\langle \hat{\Phi}^2 \rangle$	60
		4.1.1	Structure of Divergences	60
		4.1.2	Calculation of the Anomaly	62
		4.1.3	Spherical-Reduction Anomaly in Schwarzschild	
			Spacetime	67
	4.2	The D	Dimensional-Reduction Anomaly in the Effective Action	67
	4.3	The D	Dimensional-Reduction Anomaly in the Stress Tensor	71
	4.4	Anom	alous Stress Tensor in Schwarzschild Space	77
		4.4.1	Massless Field	80

TABLE OF CONTENTS

		4.4.2	Large-Mass Case	•	88
5	5 Static Decompositions				
	5.1	Rindle	er space	•	93
	5.2	The D) imensional-Reduction Anomaly in $\langle \hat{\Phi}^2 \rangle$	•	97
		5.2.1	(1+3) Reduction	•	98
		5.2.2	(2+2) Reduction	. 1	03
	5.3	The D	imensional-Reduction Anomaly in the Effective Action	. 1	05
		5.3.1	(1+3) Reduction	. 1	05
		5.3.2	(2+2) Reduction	. 1	07
	5.4	The D	Dimensional-Reduction Anomaly in $\langle \hat{\Phi}^2 angle$ at Finite		
		Tempe	erature	. 1	08
6	Ana	lytic A	Approximations for $\langle \hat{\Phi}^2 angle$ and W	1	14
	6.1	Analy	tic Approximation for $\langle \hat{\Phi}^2 angle$. 1	14
	6.2	Analy	tic Approximation for the Effective Action	. 1	20
	6.3	Analy	tic Approximation for $\langle \hat{T}_{\mu u} angle$. 1	22
7	Dise	cussior	and Future Work	1	25
A	Qua	intum	Fields in Flat Spacetime at Finite Temperature	1	29
	A.1	Four c	limensions	. 1	30
	A.2	Two d	limensions	. 1	32
в	Sph	erical	Decompositions	1	34
2	B 1	Spheri	ical Decomposition of Curvatures	1	34
	B.1 B.2	Small	Distance Expansions	• •	37
	<u>.</u>	Onidil.	Distance Expansions	• 1	
С	Stat	tic Deo	compositions	1	40
	C.1	Small	-Distance Expansions for (1+3) Reductions	. 1	40

TABLE OF CONTENTS

C.2	Small-Distance Expansions for (2+2) Reductions			
C.3	.3 Useful Formulae			
	C.3.1	The Modified Bessel Function K_{ν}		
	C.3.2	Integrals of μ^n for the (1+3) Reduction		
	C.3.3	Formulae for the (2+2) Reduction		

Bibliography

	D dimensions	D – n dimensions	n dimensions
indices	μ, ν, \ldots	a, b, c, \ldots	i, j, k, \ldots
coordinates	X^{μ}	x^{a}	y^i
metric	$g_{\mu u}$	h_{ab}	Ω_{ij}
d'Alembertian		Δ_{h}	$\Delta_{oldsymbol{\Omega}}$
covariant derivative	$ abla_{\mu},~()_{;\mu}$	$ abla_a, abla, ()_{ a }$	∇_i , (): <i>i</i>
field objects	G, W, \ldots	$\mathcal{G}, \mathcal{W}, \ldots$	
curvatures	$^{D}R = R[g]$	R = R[h]	$R[\Omega]$

 Table 1: Conventions on Notation

Chapter 1

Introduction

In the absence of a full quantum theory of gravitation, attempts to model the behaviour of quantized matter incorporating gravitational interactions are forced to rely on the semiclassical approximation. In this approach, the gravitational field is treated as a classical background on which the quantized matter fields propagate. Quantum amplitudes may then be calculated perturbatively to the desired order.¹

Even at the semiclassical level, vacuum polarization effects in quantized matter are notoriously difficult to compute. As a result, much effort has been devoted to calculating vacuum polarization and particle production in privileged spacetimes which possess a high degree of symmetry; these include homogeneous cosmological models [1-4], and eternal black hole spacetimes [5-19]. In these systems, one can take advantage of the continuous symmetries of the geometry to simplify the calculation of field-dependent functions. By selecting a coordinate system based on the symmetries and using separation of variables, the quantum field may be decomposed in terms of modes in the symmetry directions to effec-

¹Small gravitational fluctuations about the classical background may be treated as a separate matter field of gravitons and quantized to the one-loop level, the non-renormalizability of gravity preventing a consistent extension to higher orders.

tively reduce the dimensionality of the system. For example, in a *D*-dimensional static spacetime one can decompose the quantum field in terms of Fourier time modes $e^{i\omega t}$. Substituting this Fourier transform into the field equation yields a new field equation in (D-1) dimensions with an ω -dependent potential. Quantities of interest in the physical theory can then be obtained by solving for the corresponding quantities in the dimensionally reduced theory and summing over all ω (performing the inverse Fourier transform).

This technique is standard in the solution of partial differential equations for classical fields. There is, however, a difficulty that one encounters in applying dimensional reduction to quantum field theories. Specifically, quantities of physical interest in quantum field theories, such as expectation values, typically are divergent and must be renormalized to yield a finite, meaningful result. This raises the question of how renormalization affects the relationship between the dimensionally reduced theory and the original physical theory. In the language of operators, we ask whether dimensional reduction and renormalization commute. In this report, we find that they do not. This may be understood from the fact that renormalization is a purely local operation, depending only on the short-distance or high-frequency behaviour of the theory. Mode decomposition, however, is sensitive to global characteristics such as topology, and so probes the long-distance or low-frequency behaviour. Integrating out some portion of the manifold can change the global properties of the system, a change to which the renormalization is insensitive. This leads to different results when renormalization is carried out before versus after the mode decomposition. In subsequent chapters we will see that further discrepancies between renormalization in the two theories arise due to local effects depending on the curvature and field potential. We call the resulting non-commutability of dimensional reduction and renormalization the **dimensional-reduction anomaly**.

CHAPTER 1. INTRODUCTION

The problem of dimensional reduction and renormalization may be of relevance to various issues that have appeared in the literature in recent years. One of the most notable of these is the attempt to calculate the stress tensor and Hawking radiation of Schwarzschild black holes using two-dimensional dilaton-gravity models [20-31]. In this case, a massless, minimally coupled scalar field in a four-dimensional spherically symmetric spacetime (such as that of a Schwarzschild black hole) is decomposed into spherical harmonics, reducing the field to one propagating in two dimensions. Using the conformal properties of two-dimensional spaces, the calculation of the effective action is greatly simplified. One can then obtain the contribution of the $\ell = 0$ spherical mode to the stress tensor and Hawking radiation in the original, physical spacetime. Various attempts along these lines [32-39] have been rewarded with unsettling results [40], including in some cases a negative Hawking flux, and in all cases two-dimensional stress tensors that are qualitatively different from the expected four-dimensional one near the black hole. We show that the dimensional-reduction anomaly supplies a state-independent contribution to the stress tensor which appears to correct the predictions of the dimensionally reduced theories near the event horizon, while leaving the asymptotic behaviour (including the Hawking radiation) unaffected.

A second system that we shall consider is that of a scalar field in static space at finite temperature. In this case we demonstrate that the dimensionalreduction anomaly can be used to derive a new general analytic approximation for the renormalized physical theory which is based on the high-frequency behaviour of the field. This technique can be viewed as an extension of previous approximations for conformal fields by Page, Brown, and Ottewill [8, 9] and by Frolov and Zelnikov [10], as well as the approximation of Anderson, Hiscock, and Samuel [11, 12] for fields in static spherically symmetric spacetimes.

This thesis is organized as follows. In Chapter 2 we review some of the basic formalism of free quantum fields in curved space. We then examine the dimensional reduction of a quantum field in a spacetime with a homogeneous subspace. We indicate how the dimensional-reduction anomaly arises and relate it to the more general multiplicative anomaly. Chapter 3 is devoted to explicit calculations of the dimensional-reduction anomaly for the simple case of spherical decompositions in flat space. We show how the anomaly arises and why it must be included for the dimensional reduction to yield standard results for flat spacetime. In Chapter 4 we extend our considerations to general four-dimensional spherically symmetric spaces, with particular emphasis on the Schwarzschild geometry. We calculate explicitly the anomalies in $\langle \hat{\Phi}^2 \rangle$, the effective action, and the stress tensor. Comparisons to known results for renormalized quantum fields in two and four dimensions demonstrate the significance of the anomaly to dilaton-gravity models. In Chapter 5 we examine dimensional reduction in four-dimensional static spaces and derive the corresponding anomaly. For the zero-temperature case, we demonstrate in Chapter 6 how this anomaly may be used to obtain a new approximation scheme for quantum fields in static spaces. We conclude in Chapter 7 with a summary of our results and point out some prospects for future work.

Chapter 2

Quantum Field Theory in Curved Spacetime and Dimensional Reduction

Our main subject is special properties of quantum fields propagating in curved spacetimes. In this chapter we collect some general results which will be useful later. We begin in Section 2.1 with a brief review of the quantization of a free scalar field in a gravitational background, with emphasis on such topics as Green functions, the effective action, the heat kernel, and the choice of quantum state. In Section 2.2 we examine the dimensional reduction of such a theory when the spacetime contains a homogeneous subspace, showing how it may be rewritten as a collection of fields in a lower-dimensional spacetime. In Section 2.3 we indicate how this dimensional reduction breaks down under renormalization, producing the dimensional-reduction anomaly.

2.1 Quantum Field Theory in Curved Spacetime

Let us review the general scheme for the quantization of a free field in a gravitational background. It will be sufficient for our purposes to consider real scalar fields only, though the following formalism is easily extended to more general bosonic or fermionic fields. For more general and detailed accounts see, for example, DeWitt [41, 42], Berezin [43], Birrell and Davies [44], and Wald [45].

2.1.1 Classical Field

A self-consistent description of a classical scalar field Φ propagating in a curved spacetime with metric $g_{\mu\nu}$, where the gravitational interaction of the matter is taken into account, may be formulated in terms of the actions for the scalar and gravitational fields. The classical action for Φ is taken to be

$$S = \frac{1}{2} \int dx \sqrt{|g|} \Phi F \Phi , \qquad (2.1)$$

where F is a self-adjoint operator.¹ Combining the action (2.1) with the classical Einstein-Hilbert action for general relativity,²

$$S_{\rm grav} = -\frac{1}{16\pi} \int dx \sqrt{|g|} \left[R - 2\Lambda \right] , \qquad (2.2)$$

and setting the variation of the total action $S + S_{grav}$ with respect to the metric equal to zero yields the Einstein equations,

$$R_{\mu\nu} - \frac{1}{2}g_{\mu\nu}R + \Lambda g_{\mu\nu} = 8\pi T_{\mu\nu}. \qquad (2.3)$$

¹Our F is self-adjoint if, for any smooth complex functions ψ_1 , ψ_2 having compact support in an open spacetime region of interest,

$$\int dx \sqrt{|g|} \left[\overline{\psi}_1(F\psi_2) - (\overline{F}\overline{\psi}_1)\psi_2 \right] = 0,$$

where the bar denotes the complex conjugate.

²We ignore the surface term that should be included for rigour [46].

Here $T_{\mu\nu}$ is the stress tensor for the field Φ , and is defined by

$$T_{\mu\nu} \equiv \frac{2}{\sqrt{|g|}} \frac{\delta S}{\delta g^{\mu\nu}} \,. \tag{2.4}$$

In this thesis we will consider the gravitational background as a fixed external field and ignore the back-reaction of the (quantized) matter on the geometry, focusing rather on the vacuum polarization of the scalar field.

For a fixed background geometry the field equation satisfied by Φ is found by setting to zero the variation of the action (2.1) with respect to the field, and is

$$F\Phi = 0. (2.5)$$

The self-adjointness of the field operator F can be used to show that there exists an invariant "inner product" for solutions of the field equation which is bilinear in the fields and independent of the Cauchy surface on which it is evaluated. Specifically, if Φ_1 , Φ_2 are any two complex solutions of (2.5) and Σ is a complete Cauchy hypersurface, one can show that

$$(\Phi_1, \Phi_2) \equiv -i \int_{\Sigma} d\Sigma^{\mu} \,\overline{\Phi}_1 \,\overset{\leftrightarrow}{f}_{\mu} \Phi_2$$

is invariant under smooth deformations and displacements of Σ . (If the space is noncompact, one assumes that the solutions fall off sufficiently rapidly at infinity.) Here $\stackrel{\leftrightarrow}{f}_{\mu}$ is related to F via

$$\int_{\Omega} dx \sqrt{|g|} \left[\overline{\psi}_1(F\psi_2) - (\overline{F\psi}_1)\psi_2 \right] = \int_{\partial\Omega} d\Sigma^{\mu} \overline{\psi}_1 \overset{\leftrightarrow}{f}_{\mu} \psi_2 \,,$$

where Ω is any compact region of spacetime with smooth boundary $\partial\Omega$, $d\Sigma^{\mu}$ is the outward-directed surface element of $\partial\Omega$, and ψ_1 , ψ_2 are any smooth complex functions defined on an open region containing Ω . Using this inner product, one can introduce a complete, orthonormal set of solutions $\{\Phi_i, \overline{\Phi}_i\}$ (the bar denoting the complex conjugate) such that

$$(\Phi_i, \Phi_j) = -(\overline{\Phi}_i, \overline{\Phi}_j) = \delta_{ij}, \qquad (\overline{\Phi}_i, \Phi_j) = (\Phi_i, \overline{\Phi}_j) = 0.$$
(2.6)

Since the set $\{\Phi_i, \overline{\Phi}_i\}$ is complete, we can decompose the field Φ as

$$\Phi = \sum_{i} \left[a_{i} \Phi_{i} + \overline{a}_{i} \overline{\Phi}_{i} \right] , \qquad (2.7)$$

where a_i represents the constant amplitude of the field in the mode Φ_i . Note that the set $\{\Phi_i, \overline{\Phi}_i\}$ is not unique; this will be of some importance later.

2.1.2 Canonical Quantization

Assuming that the spacetime is globally hyperbolic, so that it may be foliated by spacelike Cauchy hypersurfaces, we can define a canonical momentum Π conjugate to Φ via

$$\Pi \equiv \frac{\delta S}{\delta \Phi_{,0}} \,. \tag{2.8}$$

Here the x^0 coordinate enumerates the hypersurfaces.

A standard procedure for quantizing the theory (2.1) is to elevate Φ , Π to operators and impose on each slice the standard canonical commutation relations

$$\left[\hat{\Phi}(x^0,\mathbf{x}),\hat{\Phi}(x^0,\mathbf{x}')\right] = 0, \qquad (2.9)$$

$$\left[\hat{\Pi}(x^{0}, \mathbf{x}), \hat{\Pi}(x^{0}, \mathbf{x}')\right] = 0, \qquad (2.10)$$

$$\left|\hat{\Phi}(x^0,\mathbf{x}),\hat{\Pi}(x^0,\mathbf{x}')\right| = i\,\delta(\mathbf{x}-\mathbf{x}')\,,\tag{2.11}$$

where the \mathbf{x} are the coordinates on the hypersurfaces.

In the decomposition (2.7) the a_i , \bar{a}_i become operators:

$$\hat{\Phi} = \sum_{i} \left[\hat{a}_{i} \Phi_{i} + \hat{a}_{i}^{\dagger} \overline{\Phi}_{i} \right] , \qquad (2.12)$$

where the dagger (†) denotes the hermitian conjugate. The commutation relations (2.9)–(2.11) then imply that the \hat{a}_i , \hat{a}_i^{\dagger} act respectively as the annihilation and creation operators for quanta in the mode Φ_i , as they satisfy the commutation relations

$$[\hat{a}_i, \hat{a}_j] = [\hat{a}_i^{\dagger}, \hat{a}_j^{\dagger}] = 0, \qquad [\hat{a}_i, \hat{a}_j^{\dagger}] = \delta_{ij}.$$
 (2.13)

The free quantum field thus decomposes to a collection of independent quantum harmonic oscillators, and we can construct a Fock basis for the Hilbert space just as one would do for the quantum oscillator. In particular, the vacuum state $|0\rangle$ is defined as that state annihilated by all \hat{a}_i :

$$\hat{a}_i |0\rangle = 0. \tag{2.14}$$

Multiparticle states can then be built up by repeated applications of the creation operator \hat{a}_{i}^{\dagger} .

2.1.3 Vacuum States

It should be noted that the physical nature of the vacuum state (2.14) depends on the choice of basis modes $\{\Phi_i, \overline{\Phi}_i\}$. One of the most interesting features of quantum field theory in curved spacetime is that in general there is no set of modes which is "preferred" by the geometry. As a result, there is no unique vacuum state.

To see the consequences of this, consider two complete, orthonormal sets of states,

$$\{u_i, \overline{u}_i\}, \qquad \{v_i, \overline{v}_i\}, \qquad (2.15)$$

with associated creation and annihilation operators

$$\{\hat{a}_i, \hat{a}_i^{\dagger}\}, \qquad \{\hat{b}_i, \hat{b}_i^{\dagger}\}, \qquad (2.16)$$

and vacua $|0\rangle_u$, $|0\rangle_v$ such that

$$\hat{a}_i |0\rangle_u = 0, \qquad \hat{b}_i |0\rangle_v = 0.$$
 (2.17)

Since both sets are complete, we must be able to write the $\{v_i, \overline{v}_i\}$ as linear combinations of the $\{u_i, \overline{u}_i\}$. For example,

$$v_i = \sum_{k} [\alpha_{ik} u_k + \beta_{ik} \overline{u}_k]. \qquad (2.18)$$

This is known as a Bogoliubov transformation. The orthonormality and completeness of both sets of modes dictates that the matrices α and β must satisfy

$$\alpha \alpha^{\dagger} - \beta \beta^{\dagger} = I \,, \tag{2.19}$$

$$\overline{\alpha}\beta^{\dagger} - \overline{\beta}\alpha^{\dagger} = 0, \qquad (2.20)$$

where I is the identity matrix.

The operators $\{\hat{b}_i, \hat{b}_i^{\dagger}\}$ can also be written in terms of the $\{\hat{a}_i, \hat{a}_i^{\dagger}\}$. For example,

$$\hat{b}_{i} = \sum_{k} [\overline{\alpha}_{ik} \hat{a}_{k} - \overline{\beta}_{ik} \hat{a}_{k}^{\dagger}]. \qquad (2.21)$$

If $\beta \neq 0$, the annihilation and creation operators are mixed in the Bogoliubov transformation. As a result, the vacua $|0\rangle_u$ and $|0\rangle_v$ are inequivalent. In particular, the number of particles in the mode v_i contained in the $|0\rangle_u$ vacuum is

$$_{u}\langle 0|\hat{b}_{i}^{\dagger}\hat{b}_{i}|0\rangle_{u} = \sum_{k} |\beta_{ki}|^{2}.$$
 (2.22)

If the gravitational field vanishes (or becomes static) at very early or very late times, the vacuum state in these asymptotic regions can be chosen naturally. To do so, one identifies the basis modes Φ_i with positive-frequency solutions $(\Phi \propto e^{-i\omega t}, \omega > 0)$. For such a definition the corresponding vacuum state is the state of lowest possible energy. For $x^0 \to -\infty$ it is known as the "in" vacuum and denoted by $|0; in\rangle$, while for $x^0 \to +\infty$ it is the "out" vacuum $|0; out\rangle$. In the presence of both "in" and "out" asymptotic regions one has at least two generically different privileged vacuum states. If at intermediate times the gravitational field is non-trivial, a given "in" mode Φ_i^{in} will typically scatter into a linear combination of the late-time modes $\{\Phi_j^{out}, \overline{\Phi}_j^{out}\}$. In particular, if the decomposition of Φ_i^{in} into "out" modes contains any of the $\overline{\Phi}_j^{out}$ (i.e., if $\beta \neq 0$) then the vacuum states $|0; in\rangle$ and $|0; out\rangle$ are not equivalent. In this case a field which begins in the state $|0; in\rangle$ at early times will contain "out" particles at late times. The interpretation is that the nontrivial gravitational field has produced particles of the quantum field.

2.1.4 Green Functions

The various Green functions associated with the quantized theory may be defined as expectation values or matrix elements of products of the field operators for a given quantum state. Of particular interest to us are the Hadamard Green function $G^{(1)}(x, x')$ and the Feynman Green function $G_F(x, x')$ for some vacuum state $|0\rangle$:

$$G^{(1)}(x,x') \equiv \langle 0|\{\hat{\Phi}(x),\hat{\Phi}(x')\}|0\rangle, \qquad (2.23)$$

$$G_F(x,x') \equiv i \langle 0 | T(\hat{\Phi}(x), \hat{\Phi}(x')) | 0 \rangle. \qquad (2.24)$$

Here $\{,\}$ denotes the anticommutator and T() the time-ordered product,

$$T(\hat{\Phi}(x),\hat{\Phi}(x')) = \theta(x,x')\hat{\Phi}(x)\hat{\Phi}(x') + \theta(x',x)\hat{\Phi}(x')\hat{\Phi}(x),$$

where $\theta(x, x') = 1$ if x lies to the future of a spacelike hypersurface through x', and vanishes otherwise. The choice of hypersurface is arbitrary, as the field operators commute for spacelike separations.

Using the commutation relations (2.9)-(2.11), one can show that the Green functions obey the differential equations

$$FG^{(1)}(x,x') = 0,$$
 (2.25)

$$FG_F(x,x') = -\delta(x,x') \equiv -\frac{\delta(x-x')}{\sqrt{|g|}}.$$
 (2.26)

Note that the boundary conditions obeyed by the Green functions are contained implicitly by the choice of modes $\{\Phi_i, \overline{\Phi}_i\}$ used to define the vacuum state.

For a given mode set $\{\Phi_i, \overline{\Phi}_i\}$, the Green functions can be found by substituting the mode decomposition (2.12) into (2.23) or (2.24). For example, for the Hadamard function we find

$$G^{(1)}(x,x') = \sum_{i} \left[\Phi_i(x) \overline{\Phi}_i(x') + \overline{\Phi}_i(x) \Phi_i(x') \right].$$
(2.27)

The stress tensor for the quantized theory may also be expressed as a sum over modes by taking the quadratic-in- Φ expression resulting from the differentiation of the action with respect to the metric in (2.4), making the replacement $\Phi \rightarrow \hat{\Phi}$, and taking the expectation value. For the vacuum state we obtain

$$\langle 0|\hat{T}_{\mu\nu}|0\rangle = \sum_{i} T_{\mu\nu}(\Phi_{i}, \overline{\Phi}_{i}), \qquad (2.28)$$

where $T_{\mu\nu}(\Phi_i, \overline{\Phi}_i)$ is the expression resulting from (2.4); it is symmetric in x, x'and in $\Phi_i, \overline{\Phi}_i$. Equivalently, we will see in Section 2.2 that the stress tensor can be written as the coincidence limit $x' \to x$ of a differential operator acting on the Hadamard Green function $G^{(1)}(x, x')$. Using the mode-decomposed form (2.27) for $G^{(1)}(x, x')$ then yields (2.28).

2.1.5 Effective Action

We will be interested in calculating the expectation value of the stress tensor for quantized scalar fields. While the mode-sum technique is physically transparent, the required mode set is typically extremely difficult to find. An alternative approach which is both elegant and enlightening is to define a quantized version of the action (2.1), called the effective action and denoted W, from which the expectation value of the stress tensor for a given state can be obtained by functional differentiation in analogy to (2.4):

$$\langle \hat{T}_{\mu\nu} \rangle \equiv \frac{2}{\sqrt{|g|}} \frac{\delta W}{\delta g^{\mu\nu}} \,. \tag{2.29}$$

Using the 'in-out' formalism, it is not difficult to show that W may be obtained from the vacuum-to-vacuum transition amplitude (0; out|0; in). With hindsight we define

$$e^{iW} \equiv \langle 0; out|0; in \rangle = \int \mathcal{D}[\Phi] e^{iS[\Phi]},$$
 (2.30)

where we use the path integral representation for (0; out|0; in) (see, e.g., [44, 47]). Under a change δS in the classical action, the corresponding variation in W is given by

$$\delta W = e^{-iW} \int \mathcal{D}[\Phi] \, \delta S \, e^{iS[\phi]}$$
(2.31)

$$= \frac{\langle 0; \operatorname{out} | \delta S | 0; \operatorname{in} \rangle}{\langle 0; \operatorname{out} | 0; \operatorname{in} \rangle}.$$
(2.32)

Equation (2.32) is a statement of the well-known "Schwinger variational principle" [48]. For the special case where the variation δS is due to a variation $\delta g^{\mu\nu}$ in the metric, we have by definition of the stress tensor

$$\delta S = \frac{1}{2} \int dx \sqrt{|g|} \,\delta g^{\mu\nu} \,T_{\mu\nu} \,. \tag{2.33}$$

Substituting (2.33) into (2.32) it clearly follows that

$$\frac{\langle 0; \operatorname{out} | \hat{T}_{\mu\nu} | 0; \operatorname{in} \rangle}{\langle 0; \operatorname{out} | 0; \operatorname{in} \rangle} = \frac{2}{\sqrt{|g|}} \frac{\delta W}{\delta g^{\mu\nu}}.$$
(2.34)

Other expectation values such as $\langle 0; in | \hat{T}_{\mu\nu} | 0; in \rangle$ differ from (2.34) by finite, welldefined amounts [42].

The next task is to find a convenient method for calculating W, which is typically a difficult problem. Returning to (2.30) and inserting the expression (2.1) for the classical field action, we obtain a Gaussian path integral for the free field. This evaluates to [44, 47]

$$e^{iW} \propto [\det(F)]^{-\frac{1}{2}},$$
 (2.35)

where we treat the field operator F formally as a matrix. From this we determine

$$W = \frac{i}{2} \ln \det(F) = \frac{i}{2} \operatorname{Tr} \ln(F) .$$
 (2.36)

In writing (2.36) we have dropped a metric-independent additive constant, as it will not contribute to the stress tensor.

2.1.6 Heat Kernel

The result (2.36) for the effective action can be reformulated in a more convenient manner through the introduction of a quantity known as the heat kernel. Let us formally consider G_F as an operator on the space of vectors $|x\rangle$ normalized by

$$\langle x|x'\rangle = \delta(x,x'),$$
 (2.37)

such that

$$G_F(x, x') = \langle x | G_F | x' \rangle.$$
(2.38)

The differential equation for G_F then becomes $G_F = -F^{-1}$, and hence one can write

$$G_F = -F^{-1} = i \int_0^\infty ds \, e^{iFs} \,, \tag{2.39}$$

$$\ln(F) = -\int_0^\infty \frac{ds}{s} e^{iFs}, \qquad (2.40)$$

where in the second equation we have discarded a metric-independent infinite constant. For rigour, we should add a small positive imaginary part to F to ensure that the integrals over s converge in the $s \to \infty$ limit; for a massive field this is equivalent to setting $m^2 \to m^2 - i\epsilon$. This procedure also assures that (2.39) evaluates to the Feynman Green function rather than one of the other Green functions [49].

Taking matrix elements, we find

$$G_F(x,x') = i \int_0^\infty ds \, K(x,x'|s) ,$$
 (2.41)

$$W = -\frac{i}{2} \int_0^\infty \frac{ds}{s} \int dx \sqrt{|g|} K(x, x|s) , \qquad (2.42)$$

where the function $K(x, x'|s) = \langle x|e^{iFs}|x'\rangle$ is known as the heat kernel for the operator F. It is clearly a solution of the Schrödinger-like equation

$$\left(i\frac{\partial}{\partial s} + F\right)K(x, x'|s) = 0 \tag{2.43}$$

with initial condition

$$K(x, x'|s=0) = \delta(x, x').$$
(2.44)

These results can also be obtained through alternative means which avoid the formal constructions (2.37)-(2.40) [50].

The heat kernel formulation has several virtues. It provides a unified means of studying both the Green function and the effective action. From a conceptual standpoint, it is interesting because it gives a quantum-mechanical solution³ to a field theory problem. Most important for our purposes, the heat kernel formulation is convenient for renormalizing quantum field theories in curved spacetime, which we consider next.

2.1.7 Renormalization

One of the principle features of quantum field theory is that quantities of physical interest contain infinities. For example, the expectation value of the square of the field operator, which is a measure of the quantum fluctuations in the field, can be obtained from the coincidence limit of the Hadamard Green function (2.23):

$$\langle \hat{\Phi}^2(x) \rangle = \lim_{x' \to x} \frac{1}{2} G^{(1)}(x, x') .$$
 (2.45)

However, this Green function diverges in the coincidence limit, so that $\langle \hat{\Phi}^2(x) \rangle$ is not well-defined. Similar infinities plague the stress tensor and the effective action. The procedure by which these infinities are eliminated to yield finite, physically meaningful results is known as renormalization.

The heat kernel formalism just discussed provides a powerful method for renormalizing quantum field theories in curved spacetimes. In (2.41)-(2.42), the divergences in both the Green function and the effective action come from

³Note that the heat kernel is formally identical to the amplitude for a quantum-mechanical particle to propagate from x' to x in "time" s under the "Hamiltonian" -F.

the $s \rightarrow 0$ limit of the s integral (the $s \rightarrow \infty$ limit is well-behaved due to the $i\epsilon$ prescription). The advantage of the heat kernel formulation is that the small-s behaviour of the heat kernel is known for arbitrary curved spaces of any dimension, in the form of the Schwinger-DeWitt expansion [41, 42, 51]. We postpone a detailed examination of this expansion until Section 4.1; suffice it to say that the divergences in the effective action (Green function) in *D*-dimensional spacetime are contained in the first N_D terms ($N_D - 1$ terms) of the Schwinger-DeWitt expansion of the heat kernel, where

$$N_D = \begin{cases} \frac{D}{2} + 1 & \text{for } D \text{ even,} \\ \\ \frac{D+1}{2} & \text{for } D \text{ odd.} \end{cases}$$
(2.46)

Moreover, the coefficients in the Schwinger-DeWitt expansion are purely local functions of the background fields appearing in the classical gravitational and matter actions: the curvature, the field potential, and their covariant derivatives. (This is because the divergences come about in the $x' \rightarrow x$ limit, and so must be independent of boundary conditions and the global nature of the spacetime.) Because of their purely local form, the divergences can be absorbed into the classical actions (2.1), (2.2) by redefinitions of the Newtonian gravitational constant, the cosmological constant, the coupling in the potential, and the other parameters in these actions. From a practical standpoint, that means that we can renormalize a given expectation value by simply subtracting the contribution from the first few terms of the Schwinger-DeWitt expansion.

2.1.8 Euclidean Approach

Until now we have been considering quantum field theory on Lorentzian manifolds; i.e., on spacetimes with indefinite metric. An alternative approach which is frequently used for quantum field theory calculations is to define and compute all quantities of interest on a Euclidean manifold; i.e., on a space with positivedefinite metric [52]. For static spaces, this is equivalent to moving from real to imaginary time via the 'Wick rotation' $t \rightarrow -it$. Results for the physical spacetime can be obtained by analytic continuation from Euclidean space at the end of the calculation.

The Euclidean approach has several advantages. Foremost among these is that the differential operator F, which is hyperbolic in Lorentzian manifolds, becomes elliptic in Euclidean manifolds. For a wide class of Euclidean spaces and operators, its inverse (the Green function) is then well-defined and unique [52]. This Green function G obeys the differential equation

$$FG(X, X') = \delta^D(X, X') \equiv \frac{\delta^D(X - X')}{\sqrt{g}}, \qquad (2.47)$$

and vanishes as the separation distance goes to infinity in noncompact manifolds.

One can also obtain the Euclidean Green function from the heat kernel, which is so named because in Euclidean signature it obeys the heat equation

$$\left(-\frac{\partial}{\partial s}+F\right)K(x,x'|s)=0$$
(2.48)

with initial condition

$$K(x, x'|s=0) = \delta(x, x').$$
(2.49)

The heat equation is obtained from (2.43) by the replacement $s \rightarrow -is$, $t \rightarrow -it$. Similarly, the Euclidean Green function and effective action are related to the Euclidean heat kernel via

$$G(x, x') = \int_0^\infty ds \, K(x, x'|s) \,, \tag{2.50}$$

$$W = -\frac{1}{2} \int_0^\infty \frac{ds}{s} \int dx \sqrt{g} K(x, x|s) \,. \tag{2.51}$$

Note that for massive fields these integrals are well-defined in the $s \to \infty$ limit because the Euclidean heat kernel is exponentially damped for large s; no $i\epsilon$ prescription is needed.

Because of these properties, it is often more convenient to work in Euclidean space and continue the final results back to Lorentzian spacetime. For a static space, the inverse Wick rotation gives

Lorentzian Euclidean

$$G_F = i G |_{t \to it}$$
, (2.52)
 $W = i W |_{t \to it}$.

Note that the Green function that results from the continuation of this G back to real time is the Feynman function G_F [49, 52].

One other useful characteristic of Euclidean spaces is that they are very convenient for considering quantum fields at nonzero temperatures [53]. In particular, one can show that expectation values for a quantum field at temperature $t = \beta^{-1}$ can be obtained by treating the field as propagating on a manifold which is periodic in imaginary time with period β [54, 55]. From this it follows that the Euclidean Green function and the Lorentzian Hadamard function are both periodic in Euclidean time (antiperiodic for fermions) [56, 57]. These properties will be useful when we consider quantum fields at finite temperatures.

One drawback of the Euclidean approach is that many spacetimes of interest are not sections of a complex manifold which also contains a unique Euclidean section. However, this formulation is applicable to the spacetimes we shall be most interested in: flat spacetime, Schwarzschild spacetime, and general static spacetimes. In future chapters our calculations of the dimensional-reduction anomaly will be done for Euclidean manifolds.

2.2 Dimensional Reduction

In this section we consider the dimensional reduction of a scalar field in a formal manner. We demonstrate how field objects such as the effective action and stress tensor may be related to the corresponding objects in a dimensionally reduced theory. In the following section we will indicate how this relationship breaks down under renormalization to give rise to the dimensional-reduction anomaly. Henceforth we work in Euclidean space unless stated otherwise.

2.2.1 Spacetime Metric and Notations

Consider a D-dimensional space with a line element of the form

$$ds^{2} = g_{\mu\nu}(X^{\tau})dX^{\mu}dX^{\nu} = h_{ab}(x^{c})dx^{a}dx^{b} + e^{-4\phi(x^{c})/n}\Omega_{ij}(y^{k})dy^{i}dy^{j}.$$
 (2.53)

Here $X^{\alpha} = (x^{a}, y^{i})$, Ω_{ij} is the metric of an *n*-dimensional homogeneous space called the *internal space*, and h_{ab} is an arbitrary (D - n)-dimensional Euclidean metric. The function ϕ is known as the *dilaton*, and sets the scale of the internal space. The normalization of the dilaton field is a question of convenience; we set it by requiring that \sqrt{g} for the metric (2.53) be proportional to $e^{-2\phi}$ for any number of internal dimensions *n*. Well-known examples of metrics of the form (2.53) are those of spherical spacetimes, and metrics connected with a dimensional reduction in Kaluza-Klein theories.

At this point some conventions on notation are in order. We shall need to be able to distinguish quantities like Green functions defined in different dimensions. "Ordinary" letters such as G, W are used for the original D-dimensional theory, while calligraphic letters such as \mathcal{G} , W refer to dimensionally reduced quantities. All curvatures will be with respect to h unless explicitly labelled otherwise; for example, R = R[h] and ${}^{D}R = R[g]$. As for differential operators, we shall understand \Box and $()_{;\mu}$, ∇_{μ} to represent the d'Alembertian and covariant derivative using the metric g, while Δ_h , $()_{|a}$, ∇_a are the d'Alembertian and covariant derivative using the metric h. We also use the shorthand $(\nabla U)^2 \equiv h^{ab}U_{|a}U_{|b}$, $\nabla U \cdot \nabla S \equiv h^{ab}U_{|a}S_{|b}$. For the dilaton ϕ we shall understand ϕ_a , ϕ_{ab} , etc. to denote multiple (D - n)-dimensional covariant derivatives of ϕ calculated using the metric h. In the rare cases where we need to take derivatives in the internal homogeneous space, we use the notation Δ_{Ω} , ()_{:i}. These conventions are repeated in Table 1, which is located at the beginning of the thesis for easy reference.

2.2.2 Quantum Field in D Dimensions

Let $\tilde{\Phi}$ be a scalar field propagating in the spacetime (2.53) and described by the classical action

$$S = \frac{1}{2} \int dX \sqrt{g} \left[g^{\alpha\beta} \Phi_{;\alpha} \Phi_{;\beta} + (m^2 + V) \Phi^2 \right] , \qquad (2.54)$$

variation of which with respect to Φ yields the field equation

$$F\Phi \equiv \left(\Box - m^2 - V\right)\Phi = 0. \qquad (2.55)$$

Note that we explicitly separate the mass term m^2 from the potential V. The latter may contain an interaction with the curvature, $\xi R[g]$, for a non-minimally coupled field, but is not fixed at the moment. We only assume that when calculated on the background (2.53), the potential V is independent of the y^i coordinates.

The stress tensor for this theory is obtained by varying the action (2.54) with respect to the metric $g_{\mu\nu}$ as in (2.4). Since we need to know the behaviour of the potential under variations of the metric to compute the stress tensor, we calculate $T_{\mu\nu}$ for the physically interesting case $V = \xi R[g]$. We find

$$T_{\mu\nu} \equiv \frac{2}{\sqrt{g}} \frac{\delta S}{\delta g^{\mu\nu}}$$
(2.56)
= $(1 - 2\xi) \Phi_{;\mu} \Phi_{;\nu} - 2\xi \Phi_{;\mu\nu} \Phi + \xi R_{\mu\nu} [g] \Phi^2$
+ $(2\xi - \frac{1}{2}) g_{\mu\nu} \left[g^{\alpha\beta} \Phi_{;\alpha} \Phi_{;\beta} + (m^2 + \xi R[g]) \Phi^2 \right],$ (2.57)

where we have used the field equation (2.55) to remove $\Box \Phi$ terms. Note that this
classical stress tensor is traceless for conformally invariant fields, i.e., for $m^2 = 0$, $\xi = \frac{1}{4} \frac{D-2}{D-1}$.

In the quantum theory, the field Φ is elevated to an operator. Quantities of physical interest, such as $\langle \hat{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu\nu} \rangle$, can be written in terms of the Euclidean Green function,

$$G(X, X') = \langle \hat{\Phi}(X) \hat{\Phi}(X') \rangle, \qquad (2.58)$$

which satisfies

$$FG(X, X') = -\delta(X, X') \equiv -\frac{\delta(X - X')}{\sqrt{g}}.$$
 (2.59)

Clearly,

$$\langle \hat{\Phi}^2(X) \rangle = \lim_{X' \to X} G(X, X') \,. \tag{2.60}$$

Using (2.57), the expectation value of the quantum stress tensor can be written in terms of a differential operator acting on G(X, X') as follows:

$$\langle \hat{T}_{\mu\nu} \rangle = \lim_{X' \to X} D_{\mu\nu} G(X, X') , \qquad (2.61)$$

where for $V = \xi R[g]$

$$D_{\mu\nu} = \left(\frac{1}{2} - \xi\right) \left(g^{\alpha'}_{\mu} \nabla_{\alpha'} \nabla_{\nu} + g^{\alpha'}_{\nu} \nabla_{\mu} \nabla_{\alpha'}\right) - \xi \left(\nabla_{\mu} \nabla_{\nu} + g^{\alpha'}_{\mu} g^{\beta'}_{\nu} \nabla_{\alpha'} \nabla_{\beta'}\right) + \left(2\xi - \frac{1}{2}\right) g_{\mu\nu} \left(g^{\alpha\beta'} \nabla_{\alpha} \nabla_{\beta'} + m^2 + \xi R[g]\right) + \xi R_{\mu\nu}[g].$$
(2.62)

Here $g_{\mu}^{\alpha'}$ is the bivector of parallel transport, and ∇_{α} ($\nabla_{\alpha'}$) is the covariant derivative at X (X') using the metric g. If the Green function has been 'renormalized' so that it and its first two derivatives are finite as $X' \to X$ then one may set $g_{\mu}^{\alpha'} \to g_{\mu}^{\alpha}$.

2.2.3 Mode Decompositions

Let us now consider what happens when we take advantage of the homogeneity of the internal space $\Omega_{ij}dy^i dy^j$. Using the line element (2.53), the operator \Box becomes

$$\Box = \Delta_h - 2\,\nabla\phi\cdot\nabla + \mathrm{e}^{4\phi/n}\Delta_\Omega\,,\tag{2.63}$$

where Δ_h , Δ_{Ω} are the d'Alembertians corresponding to the metrics h_{ab} , Ω_{ij} respectively, and ∇ is understood to denote the covariant derivative with respect to the metric h_{ab} .

Considerable simplification of the problem in spacetime (2.53) is connected with the fact that for a wide class of homogeneous metrics the eigenvalue problem

$$\Delta_{\Omega} Y(y) = -\lambda Y(y) \tag{2.64}$$

is well-studied [58]. We denote by $Y_{\lambda \varrho}$ the harmonics or eigenfunctions of (2.64), and use a collective index ϱ to distinguish between different solutions of (2.64) for the same λ . These modes may be chosen to obey standard orthonormality and completeness conditions,

$$\int dy \sqrt{\Omega} Y_{\lambda \varrho}(y) \overline{Y}_{\lambda' \varrho'}(y) = \delta_{\lambda \lambda'} \delta_{\varrho \varrho'}, \qquad (2.65)$$

$$\sum_{\lambda \varrho} Y_{\lambda \varrho}(y) \overline{Y}_{\lambda \varrho}(y') = \delta(y, y') \equiv \frac{\delta^n (y - y')}{\sqrt{\Omega}} .$$
 (2.66)

We write summation over indices assuming that the spectrum is discrete (or equivalently, that the internal space is compact). For a continuous spectrum one must replace summation by integration over the spectrum. In what follows we shall assume that this rule is automatically applied.

Owing to (2.65), (2.66), the field Φ can be decomposed in terms of the harmonics $Y_{\lambda \varrho}(y)$:

$$\Phi(X) = e^{\phi(x)} \sum_{\lambda, \varrho} c_{\lambda \varrho} \varphi_{\lambda}(x) Y_{\lambda \varrho}(y) , \qquad (2.67)$$

where $c_{\lambda \rho}$ are constants. Substituting this mode decomposition into the classical action (2.54) and using (2.64), (2.65) gives

$$S = \frac{1}{2} \int dx \sqrt{h} \int dy \sqrt{\Omega} e^{-2\phi} \left[e^{4\phi/n} \Omega^{ij} \Phi_{;i} \overline{\Phi}_{;j} + h^{ab} \Phi_{;a} \overline{\Phi}_{;b} + (m^2 + V) \Phi \overline{\Phi} \right]$$

$$= \sum_{\lambda,\varrho} \sum_{\lambda',\varrho'} c_{\lambda\varrho} c_{\lambda'\varrho'} \frac{1}{2} \int dx \sqrt{h} \int dy \sqrt{\Omega} \left[e^{4\phi/n} \varphi_{\lambda} \varphi_{\lambda'} \Omega^{ij} Y_{\lambda\varrho;i} \overline{Y}_{\lambda'\varrho';j} + h^{ab} (\phi_a \varphi_{\lambda} + \varphi_{\lambda|a}) (\phi_b \varphi_{\lambda'} + \varphi_{\lambda'|b}) Y_{\lambda\varrho} \overline{Y}_{\lambda'\varrho'} + (m^2 + V) \varphi_{\lambda} \varphi_{\lambda'} Y_{\lambda\varrho} \overline{Y}_{\lambda'\varrho'} \right]$$

$$= \sum_{\lambda,\varrho} |c_{\lambda\varrho}|^2 \frac{1}{2} \int dx \sqrt{h} \left[h^{ab} \varphi_{\lambda|a} \varphi_{\lambda|b} + (m^2 + V + \phi^a \phi_a - \phi_a^a + \lambda e^{4\phi/n}) \varphi_{\lambda}^2 \right].$$
(2.68)

We use complex notation in (2.68) in case the modes $Y_{\lambda \rho}$ are complex.⁴

The classical action S for the D-dimensional theory can thus be written as the sum of (D-n)-dimensional actions S_{λ} ,

$$S = \sum_{\lambda, \varrho} |c_{\lambda \varrho}|^2 S_{\lambda}, \qquad (2.69)$$

where

$$S_{\lambda} = \frac{1}{2} \int dx \sqrt{h} \left[h^{ab} \nabla_{a} \varphi_{\lambda} \nabla_{b} \varphi_{\lambda} + (m^{2} + V_{\lambda}) \varphi_{\lambda}^{2} \right], \qquad (2.70)$$

and the 'induced potential' V_{λ} is

$$V_{\lambda} \equiv V - \Delta_h \phi + (\nabla \phi)^2 + \lambda e^{4\phi/n} . \qquad (2.71)$$

In other words, by expanding the field in modes we effectively reduce the original D-dimensional system to a similar system in (D-n)-dimensional space with an effective potential V_{λ} depending on the dilaton field ϕ .

An important feature of this dimensional reduction is that the (D - n)dimensional action S_{λ} is independent of ϱ . As a result, for a given λ each ϱ makes an equal contribution to the classical action (2.68) up to the choice of the constants $c_{\lambda\varrho}$. In the quantum theory, the $c_{\lambda\varrho}$ will be replaced by creation and annihilation operators, and each ϱ contributes equally to the quantum action and other expectation values. The total action due to all modes ϱ for a given λ

⁴One can always choose real $Y_{\lambda \varrho}$. In practice, however, it is often more convenient to work with complex modes, such as the Fourier modes $e^{\pm i\omega t}$ or the spherical harmonics $Y_{\ell m}$.

will then be simply the action for any one ρ multiplied by the degeneracy of the eigenvalue λ .

The (D-n)-dimensional field $\varphi_{\lambda}(x)$ obeys the equation

$$\mathcal{F}_{\lambda} \varphi_{\lambda} \equiv \left(\Delta_{h} - m^{2} - V_{\lambda} \right) \varphi_{\lambda} = 0, \qquad (2.72)$$

which may be derived from the variation of the reduced action (2.70) or by direct substitution of the mode decomposition (2.67) into the *D*-dimensional field equation (2.55) and using (2.63). The special case $V = \xi R[g]$ is of particular interest. Substituting for R[g] in terms of R[h], $R[\Omega]$, and the dilaton (see, for example, Appendix B.1), we have

$$V_{\lambda} = \xi R[h] + (1 - 4\frac{n+1}{n}\xi)(\nabla\phi)^2 + (4\xi - 1)\Delta_h\phi + (\lambda + \xi R[\Omega])e^{4\phi/n}.$$
 (2.73)

Note that $R[\Omega]$ is a constant, since Ω_{ij} describes a homogeneous space.

The stress tensor $\mathcal{T}_{ab}^{(\lambda)}$ for our dimensionally reduced theory is obtained from the variation of (2.70) with respect to the (D-n)-dimensional metric h^{ab} . For the potential (2.73), we find

$$\mathcal{T}_{ab}^{(\lambda)} \equiv \frac{2}{\sqrt{h}} \frac{\delta S_{\lambda}}{\delta h^{ab}}$$

$$= (1 - 2\xi) \nabla_{a} \varphi_{\lambda} \nabla_{b} \varphi_{\lambda} - 2\xi \varphi_{\lambda} \nabla_{a} \nabla_{b} \varphi_{\lambda} + (2\xi - \frac{1}{2}) h_{ab} (\nabla \varphi_{\lambda})^{2} + \xi R_{ab} [h] \varphi_{\lambda}^{2}$$

$$+ (2\xi - \frac{1}{2}) h_{ab} \left[m^{2} + \xi R[h] + (4\xi - 1) \Delta_{h} \phi + (1 - 4\frac{n+1}{n}\xi) (\nabla \phi)^{2}$$

$$+ e^{4\phi/n} (\xi R[\Omega] + \lambda) \right] \varphi_{\lambda}^{2} + (1 - 4\frac{n+1}{n}\xi) \phi_{a} \phi_{b} \varphi_{\lambda}^{2}$$

$$+ (1 - 4\xi) \varphi_{\lambda} \left[\phi_{a} \nabla_{b} + \phi_{b} \nabla_{a} - \frac{1}{2} h_{ab} \Delta_{h} \phi - h_{ab} \nabla \phi \cdot \nabla \right] \varphi_{\lambda} , \qquad (2.75)$$

where we use the field equation (2.72) to remove $\Delta_h \varphi_\lambda$ terms. It is interesting that this dimensionally reduced theory is only conformally invariant for D - n = 2(i.e., when the reduction is to a two-dimensional theory), $\lambda = 0$, m = 0, and $\xi = 0$. In this case the classical stress tensor (2.75) will be traceless, while the trace of its quantum counterpart will be determined entirely by the conformal trace anomaly. This useful property, combined with the fact that any two-dimensional space is conformally related to flat space, is at the heart of most efforts involving dilaton gravity models (see Section 2.2.7).

Strictly speaking, the dimensionally reduced stress tensor does not have any components in the y^i sector. It is easily demonstrated, however, that one can write the *D*-dimensional T_i^i in terms of the functional derivative of the action (2.54) with respect to the dilaton. Defining the effective pressure $P \equiv T_i^i/n$, one finds

$$P \equiv \frac{1}{n} T_i^{\ i} = \frac{1}{2\sqrt{g}} \frac{\delta S}{\delta \phi} \,. \tag{2.76}$$

For the dimensionally reduced theory we thus define analogously

$$\mathcal{P}^{(\lambda)} \equiv \frac{1}{2\sqrt{h}} \frac{\delta S_{\lambda}}{\delta \phi}$$

$$= (2\xi - \frac{1}{2})(\nabla \varphi_{\lambda})^{2} + (2\xi - \frac{1}{2}) \left[m^{2} + \xi R[h] + (4\xi - 1)\Delta_{h}\phi + (1 - 4\frac{n+1}{n}\xi)(\nabla \phi)^{2} + e^{4\phi/n}(\xi R[\Omega] + \lambda) \right] \varphi_{\lambda}^{2}$$

$$- (1 - 4\frac{n+1}{n}\xi) \varphi_{\lambda} \left[\nabla \phi \cdot \nabla + \frac{1}{2}\Delta_{h}\phi \right] \varphi_{\lambda}$$

$$+ \frac{1}{n} e^{4\phi/n} (\xi R[\Omega] + \lambda) \varphi_{\lambda}^{2}.$$

$$(2.77)$$

For the quantized theory, the Euclidean Green function is defined by

$$\mathcal{G}_{\lambda}(x,x') = \langle \hat{\varphi}_{\lambda}(x)\hat{\varphi}_{\lambda}(x') \rangle,$$
 (2.79)

and satisfies

$$\mathcal{F}_{\lambda}\mathcal{G}_{\lambda}(x,x') = -\delta(x,x') \equiv -\frac{\delta(x-x')}{\sqrt{h}}.$$
 (2.80)

In analogy to (2.60) we have

$$\langle \hat{\varphi}_{\lambda}^{2}(x) \rangle = \lim_{x' \to x} \mathcal{G}_{\lambda}(x, x').$$
 (2.81)

Replacing φ_{λ} by its operator equivalent and taking the expectation value of both sides of (2.75), one finds that the quantum stress tensor may be calculated using

$$\langle \hat{\mathcal{T}}_{ab}^{(\lambda)} \rangle = \lim_{x' \to x} \mathcal{D}_{ab}^{(\lambda)} \mathcal{G}_{\lambda}(x, x') ,$$
 (2.82)

where

$$\mathcal{D}_{ab}^{(\lambda)} \equiv \left(\frac{1}{2} - \xi\right) \left(h_{a}^{c'} \nabla_{c'} \nabla_{b} + h_{b}^{c'} \nabla_{a} \nabla_{c'}\right) - \xi \left(\nabla_{a} \nabla_{b} + h_{a}^{c'} h_{b}^{d'} \nabla_{c'} \nabla_{d'}\right) + \left(2\xi - \frac{1}{2}\right) h_{ab} h^{cd'} \nabla_{c} \nabla_{d'} + \xi R_{ab} [h] + \left(2\xi - \frac{1}{2}\right) h_{ab} \left[m^{2} + \xi R[h] + 4\xi \Delta_{h} \phi + \left(1 - 4\frac{n+1}{n}\xi\right) (\nabla \phi)^{2} + e^{4\phi/n} (\xi R[\Omega] + \lambda)\right] + \left(1 - 4\frac{n+1}{n}\xi\right) \phi_{a} \phi_{b} + \left(2\xi - \frac{1}{2}\right) h_{ab} \phi^{c} \left(\nabla_{c} + h_{c}^{d'} \nabla_{d'}\right) - \left(2\xi - \frac{1}{2}\right) \left[\phi_{a} (\nabla_{b} + h_{b}^{d'} \nabla_{d'}) + \phi_{b} (\nabla_{a} + h_{a}^{d'} \nabla_{d'})\right].$$
(2.83)

Similarly, the effective pressure $\mathcal{P}^{(\lambda)}$ may be calculated using

$$\langle \hat{\mathcal{P}}^{(\lambda)} \rangle = \lim_{x' \to x} \mathcal{D}_{\mathcal{P}}^{(\lambda)} \mathcal{G}_{\lambda}(x, x'),$$
 (2.84)

where

$$\mathcal{D}_{\mathcal{P}}^{(\lambda)} \equiv (2\xi - \frac{1}{2})h^{cd'}\nabla_{c}\nabla_{d'} - \frac{1}{2}(1 - 4\frac{n+1}{n}\xi)\left[\phi^{c}\left(\nabla_{c} + h^{d'}_{c}\nabla_{d'}\right) + \Delta_{h}\phi\right] \\ + (2\xi - \frac{1}{2})h_{ab}\left[m^{2} + \xi R[h] + (4\xi - 1)\Delta_{h}\phi + (1 - 4\frac{n+1}{n}\xi)(\nabla\phi)^{2} + e^{4\phi/n}(\xi R[\Omega] + \lambda)\right] + \frac{1}{n}e^{4\phi/n}(\xi R[\Omega] + \lambda).$$
(2.85)

2.2.4 Dimensional Reduction of $\langle \hat{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu\nu} \rangle$

Mode-decomposition relations such as (2.67) for the field and (2.69) for the action can be extended to expectation values in the quantized theories. For example, by considering the differential equations (2.59), (2.80) for the Green functions and using (2.63) and the properties (2.64) and (2.66) of the $Y_{\lambda \varrho}$, one can easily verify that G and \mathcal{G}_{λ} are related by

$$G(X, X') = \sum_{\lambda, \varrho} e^{\phi(x) + \phi(x')} Y_{\lambda \varrho}(y) \overline{Y}_{\lambda \varrho}(y') \mathcal{G}_{\lambda}(x, x').$$
(2.86)

Let us assume for convenience that the homogeneous space is compact (similar results will hold for the noncompact case). Then one can show that

$$\lim_{y' \to y} \sum_{\varrho} Y_{\lambda \varrho}(y) \overline{Y}_{\lambda \varrho}(y') = \frac{\mathcal{N}_{\lambda}}{V_{\Omega}}, \qquad (2.87)$$

where V_{Ω} is the volume of the homogeneous space, and \mathcal{N}_{λ} is the degeneracy⁵ of the eigenvalue λ :

$$\mathcal{N}_{\lambda} = \sum_{\varrho} \ . \tag{2.88}$$

Taking the coincidence limit of both sides of (2.86) then yields

$$\langle \hat{\Phi}^2 \rangle = \sum_{\lambda} \frac{\mathcal{N}_{\lambda}}{V_{\Omega}} e^{2\phi} \langle \hat{\varphi}_{\lambda}^2 \rangle.$$
 (2.89)

Similarly, inserting the decomposition (2.86) into the expression (2.61) for the D-dimensional stress tensor, applying the operator D_{ab} , and comparing to (2.82), (2.83), one finds

$$\langle \hat{T}_{ab} \rangle = \lim_{X' \to X} D_{ab} \sum_{\lambda, \varrho} e^{\phi(x) + \phi(x')} Y_{\lambda \varrho}(y) \overline{Y}_{\lambda \varrho}(y') \mathcal{G}_{\lambda}(x, x')$$

$$= \lim_{x' \to x} \sum_{\lambda, \varrho} e^{\phi(x) + \phi(x')} Y_{\lambda \varrho}(y) \overline{Y}_{\lambda \varrho}(y) \mathcal{D}_{ab}^{(\lambda)} \mathcal{G}_{\lambda}(x, x')$$

$$= \sum_{\lambda} \frac{\mathcal{N}_{\lambda}}{V_{\Omega}} e^{2\phi} \langle \hat{T}_{ab}^{(\lambda)} \rangle.$$

$$(2.90)$$

This demonstrates formally that the x^a sector of the *D*-dimensional stress tensor can be obtained as the sum of the stress tensors for the dimensionally reduced

⁵For discrete spectra \mathcal{N}_{λ} is the dimension of the space of eigenfunctions $Y_{\lambda \varrho}$ with fixed λ . It is finite [60].

theories. A similar procedure can be carried out for the components of the stress tensor in the y^i sector (the pressures), yielding an analogous result:

$$\langle \hat{P} \rangle = \sum_{\lambda} \frac{\mathcal{N}_{\lambda}}{V_{\Omega}} e^{2\phi} \langle \hat{\mathcal{P}}^{(\lambda)} \rangle.$$
 (2.91)

Another elegant means to obtain these dimensional-reduction results is by employing the heat kernel. In Section 2.1 we were introduced to the heat kernel as a standard approach to renormalization in curved space. For a field satisfying the differential equation (2.55) in Euclidean space, the heat kernel K(X, X'|s)is a solution of of the heat equation (2.48) with initial condition (2.49); from it the Euclidean Green function and effective action may be obtained using (2.50), (2.51). Analogous formulae also hold for the dimensionally reduced theory with operator \mathcal{F}_{λ} , heat kernel $\mathcal{K}_{\lambda}(x, x'|s)$, Green function $\mathcal{G}_{\lambda}(x, x')$, and effective action \mathcal{W}_{λ} .

Using the field equations (2.55), (2.72) and the properties (2.65), (2.66) of the $Y_{\lambda \varrho}$, one sees that the heat kernels K(X, X'|s), $\mathcal{K}_{\lambda}(x, x'|s)$ obey a modedecomposition relation exactly analogous to that for the Green functions, (2.86):

$$K(X, X'|s) = \sum_{\lambda, \varrho} e^{\phi(x) + \phi(x')} Y_{\lambda \varrho}(y) \overline{Y}_{\lambda \varrho}(y') \mathcal{K}_{\lambda}(x, x'|s) .$$
(2.92)

This relation provides an easy way to produce the mode-decomposition relations for the (unrenormalized) Green function and the effective action. For example, in the compact case, integrating both sides of (2.92) over s as in (2.42) yields the mode decomposition of the quantized effective action,

$$W = \sum_{\lambda} \mathcal{N}_{\lambda} \mathcal{W}_{\lambda} , \qquad (2.93)$$

in analogy to the result (2.69) for the classical actions. The mode decompositions of the stress tensor and pressure can then be derived from (2.93) by functional differentiation. For noncompact internal spaces, the decompositions (2.86) and (2.92) for the Green function and heat kernel still hold, but $V_{\Omega} \to \infty$ and (2.87) is no longer valid. We must therefore go back to (2.89)-(2.91) and replace $\mathcal{N}_{\lambda}/V_{\Omega} \to \infty$ by $\sum_{\varrho} Y_{\lambda\varrho}(y) \overline{Y}_{\lambda\varrho}(y)$, which is easily shown to be *y*-independent. We will see explicit examples of this case in Chapters 5 and 6.

2.2.5 Dimensional Reduction of the Conservation Equation

The D-dimensional action S is invariant under the variation

$$\delta g_{\mu\nu} = \eta_{\mu;\nu} + \eta_{\nu;\mu},$$

$$\delta \Phi = \eta^{\nu} \Phi_{\nu}, \qquad (2.94)$$

where η is an arbitrary smooth vector field. This invariance implies that the *D*-dimensional stress tensor obeys the conservation equation

$$T_{\mu\nu}^{\ ;\nu} = 0. \tag{2.95}$$

This invariance survives quantum corrections (see Section 2.3), and so applies to the quantum stress tensor as well:

$$\langle \hat{T}_{\mu\nu} \rangle^{;\nu} = 0. \qquad (2.96)$$

Substituting the mode-decomposed form for $\langle \hat{T}_{\mu\nu} \rangle$ yields the corresponding (non)conservation equation for the dimensionally reduced stress tensor,

$$\nabla^{b} \langle \hat{\mathcal{T}}_{ab}^{(\lambda)} \rangle = -2\phi_{a} \langle \hat{\mathcal{P}}^{(\lambda)} \rangle .$$
(2.97)

The same result could also be derived directly in (D - n) dimensions by noting the invariance of S_{λ} under the transformation

$$\delta h_{ab} = \eta_{a|b} + \eta_{b|a},$$

$$\delta \varphi_{\lambda} = \eta^{a} \varphi_{\lambda | a},$$

$$\delta \Phi = \eta^{a} \Phi_{a}. \qquad (2.98)$$

This invariance also survives quantum corrections.

2.2.6 Induced Boundary Conditions

A final point regarding general dimensional reductions which will be very important for us later concerns global properties of the space and induced boundary conditions.

Let us assume that there exist points $x^a = x_0$ in the space (2.53) such that $e^{-\phi(x_0)} = 0$; i.e., at which the volume of the internal space vanishes. If the *D*-dimensional manifold is regular at these points, then for a regular quantum state the (renormalized) expectation values of the field in that state will also be finite there. Considering the mode decomposition formulae (2.67) and (2.86), it is clear that the dimensionally reduced field $\hat{\varphi}_{\lambda}$ and Green function \mathcal{G}_{λ} must vanish at these points. We call this effect "induced boundary conditions". It will be very important in future chapters.

Note also that the induced potential (2.71) diverges at x_0 if $\lambda \neq 0$.

2.2.7 Dilaton-Gravity Models

Classical Background

One of the interesting examples of dimensional reduction appearing in the literature in recent years is the spherical reduction of Einsteinian gravity in $D \ge 4$ dimensions to two dimensions [20]. As an example, consider the total (matter plus gravitational) action for a minimally coupled massless scalar field in a four-dimensional spacetime with zero cosmological constant:

$$S_{\rm grav} + S = -\frac{1}{2} \int d^4 X \sqrt{|g|} \left[\frac{1}{8\pi} R[g] + (\nabla \Phi)^2 \right] .$$
 (2.99)

Under the assumption that both the spacetime and field configuration are spherically symmetric (i.e., keeping only the $\ell = 0$ mode of the field in an expansion in spherical harmonics), this action reduces to

$$S_{\text{grav}} + S = -\frac{1}{2} \int d^2 x \sqrt{|h|} \left[\frac{1}{2} e^{-2\phi} \left(R[h] + 4\Delta_h \phi - 6(\nabla \phi)^2 \right) + \frac{1}{2} R[\Omega] + (\nabla \varphi_{\ell=0})^2 + \left((\nabla \phi)^2 - \Delta_h \phi \right) \varphi_{\ell=0}^2 \right], \quad (2.100)$$

where we write

$$\Phi(x) = \frac{e^{\phi(x)}\varphi_{\ell=0}(x)}{\sqrt{4\pi}},$$
 (2.101)

and $R[\Omega] = 2$ is the curvature of the unit two-sphere.

The first line of (2.100) describes two-dimensional gravity interacting with a dilaton background field ϕ . This theory can be generalized from Einsteinian gravity in $D \ge 4$ dimensions to include models inspired by string theory [21] and other sources [22-28]; the resulting class of systems is known as *dilaton gravity*. All classical solutions for dilaton gravity containing one event horizon and one singularity in the absence of matter are known [22, 25, 26], and their global properties and other characteristics have been studied [29-31]. The second line of (2.100) contains the classical action for a massless, minimally coupled two-dimensional scalar field propagating in the dilaton-dependent potential $(\nabla \phi)^2 - \Delta_h \phi$. This corresponds to the *s* mode (the $\ell = 0$ harmonic) of the four-dimensional scalar field Φ . The quantization of this sector of the theory has been the subject of much interest in recent years, and we will discuss it momentarily.

Quantum Fields: Vacuum Polarization and Particle Creation

The interest in dilaton-gravity models stems from the fact that semiclassical [32– 39] and quantum-gravitational [61–63] calculations seem to be much easier to perform in two dimensions than in higher-dimensional spacetimes. For example, the Riemann tensor in two dimensions has only one independent component, and two-dimensional spaces are conformally flat. As well, for conformally invariant fields in two dimensions the quantum stress tensor is largely fixed by knowledge of its trace. These properties make it possible to perform some vacuum polarization and other calculations for arbitrary metrics, making the study of the back-reaction of quantized matter on the gravitational background possible [64]. This leads to the hope of being able to use dilaton-gravity models to obtain information on the quantum creation and evaporation of genuine four-dimensional Schwarzschild black holes, including insight into the information-loss puzzle and black-hole thermodynamics [65].

If dilaton-gravity models are to shed light on the nature of black-hole evaporation and quantum gravity, it is clear that effects which are already known must be reproduced by them. The most interesting of these is Hawking radiation [66– 68]. The first attempt to calculate the Hawking radiation in the $\ell = 0$ mode for a four-dimensional black hole using a two-dimensional model was made by Mukhanov, Wipf, and Zelnikov [32]. They used the conformal anomaly in two dimensions to calculate the effective action for a massless scalar field coupled minimally to gravity but nonminimally to the dilaton background. This technique leaves undetermined a conformally invariant term. The authors developed a perturbative technique to compute this conformally invariant part but were unable to calculate it completely. Various attempts since then by a number of other authors [33-39] have led to different results for the Hawking radiation and stress tensor for dilaton-gravity, and disagreement on the correct approach to computing them. (Some authors [36] have stated that the Hawking radiation should be derived by integrating the (non)-conservation equation (2.97) for the two-dimensional stress tensor, while others [37] maintain that the effective action formalism is sufficient.) In addition to questions concerning their asymptotic behaviour, comparison of the stress tensors obtained from dilaton gravity to the four-dimensional stress tensor for a massless scalar field near the event horizon also shows qualitatively different behaviour. These problems call into question the usefulness of dilaton-gravity models for understanding the physics of fourdimensional black holes, and at present there does not appear to be a consensus on the issue of the quantum effective action for dilaton gravity.

Part of this thesis (Chapter 4) is devoted to an examination of this issue, and the connection between spherically reduced quantum fields and physics in four dimensions.

2.3 The Dimensional-Reduction Anomaly

In the preceding section we saw that in a space (2.53), the main objects which characterize a free quantum field, such as the Green function, the square of the field operator, and the stress tensor, can all be written as formal sums of the corresponding objects in dimensionally reduced theories. These arguments have been applied to the field theory at the classical or bare level, and have commonly been assumed to hold for the renormalized theories as well. A central point of this thesis is to demonstrate that relationships like (2.89)-(2.91) in fact break down under renormalization.

The need for renormalization is apparent when one considers, for example, the expressions (2.60) and (2.81) for the expectation value of the square of the field operator. The Green function is divergent in the coincidence limit. As a result, our mode-decomposition expressions like (2.89) are purely formal; both sides contain divergences which must be removed to yield physically meaningful results. This raises the question of whether the renormalization in D dimensions is equivalent to renormalization in (D - n) dimensions. Specifically, is the sum over λ of the divergent terms in (D - n) dimensions equal to the divergent terms in D dimensions? As discussed in Section 2.1, the divergences in the effective action W in Ddimensional spacetime can be removed by subtracting the first N_D terms of the Schwinger-DeWitt expansion of the heat kernel K for the operator F, where N_D is given by (2.46). By contrast, the dimensionally reduced action W_{λ} is renormalized by subtracting the first N_{D-n} terms of the Schwinger-DeWitt expansion of the heat kernel \mathcal{K}_{λ} for the operator \mathcal{F}_{λ} . In subsequent chapters we will calculate the "divergent" parts of these heat kernels for various spaces and demonstrate explicitly that they do not satisfy a mode-decomposition relation like (2.92). As a result, the renormalization procedure destroys the formal representation (2.93) for the effective actions, so that after renormalization one gets

$$W_{\rm ren} = \sum_{\lambda} \mathcal{N}_{\lambda} \left[\mathcal{W}_{\lambda | \rm ren} + \Delta \mathcal{W}_{\lambda} \right] \,. \tag{2.102}$$

We call the additional contribution ΔW_{λ} the dimensional-reduction anomaly. Similar anomalies occur in the other expectation values of physical interest:

$$G_{\rm ren}(X,X') = \sum_{\lambda,\varrho} e^{\phi(x) + \phi(x')} Y_{\lambda\varrho}(y) \overline{Y}_{\lambda\varrho}(y') \left[\mathcal{G}_{\lambda|\rm ren}(x,x') + \Delta \mathcal{G}_{\lambda}(x,x') \right] , \quad (2.103)$$

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = \sum_{\lambda} \frac{N_{\lambda}}{V_{\Omega}} e^{2\phi} \left[\langle \hat{\varphi}_{\lambda}^2 \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_{\lambda}^2 \rangle \right] , \qquad (2.104)$$

$$\langle \hat{T}_{ab} \rangle_{\rm ren} = \sum_{\lambda} \frac{N_{\lambda}}{V_{\Omega}} e^{2\phi} \left[\langle \hat{T}_{ab}^{(\lambda)} \rangle_{\rm ren} + \Delta \langle \hat{T}_{ab}^{(\lambda)} \rangle \right] , \qquad (2.105)$$

$$\langle \hat{P} \rangle_{\rm ren} = \sum_{\lambda} \frac{\mathcal{N}_{\lambda}}{V_{\Omega}} e^{2\phi} \left[\langle \hat{\mathcal{P}}^{(\lambda)} \rangle_{\rm ren} + \Delta \langle \hat{\mathcal{P}}^{(\lambda)} \rangle \right] \,. \tag{2.106}$$

In each case the anomaly is found as the difference between the renormalization terms for the (D-n)-dimensional theory and the mode-decomposed renormalization terms from D dimensions. For example, using the orthonormality condition (2.65) for the $Y_{\lambda \varrho}$, one can show that for compact internal spaces the anomaly in the Green function is

$$\Delta \mathcal{G}_{\lambda}(x,x') = \mathcal{G}_{\lambda|\operatorname{div}}(x,x') - \mathrm{e}^{-(\phi(x) + \phi(x'))} \int dy \sqrt{\Omega} \int dy' \sqrt{\Omega'} \, \overline{Y}_{\lambda\varrho}(y) \, Y_{\lambda\varrho}(y') \, G_{\operatorname{div}}(X,X') \, dy' \,$$

where the subscript 'div' denotes the divergent part obtained from the Schwinger-DeWitt expansion. In subsequent chapters it will be demonstrated that this anomaly is generally nonvanishing.

One might observe that there exists a relationship between the dimensionalreduction anomaly and the so-called *multiplicative anomaly* [69-71]. Formally, one can write

$$F = \prod_{\lambda,\varrho} \mathcal{F}_{\lambda} , \qquad (2.108)$$

$$-\frac{1}{2}\log \det F = -\frac{1}{2}\sum_{\lambda} \mathcal{N}_{\lambda} \log \det \mathcal{F}_{\lambda}. \qquad (2.109)$$

The latter relation is nothing but (2.93) for the quantum action. The violation of the formal relation (2.109) for products of operators after renormalization is known as the multiplicative anomaly.

In the following chapters we discuss special examples of the dimensionalreduction anomaly. For simplicity, we restrict ourselves to the physically interesting case where the number of spacetime dimensions D is 4, and the number of dimensions of the "internal" homogeneous space is 1 or 2. Unless specified otherwise, we work with manifolds of Euclidean signature. When we do consider Lorentzian spacetimes, our metric signature is always (- + + +). We use the conventions of Misner, Thorne, and Wheeler [72] for the definition of the curvature.

Chapter 3

Spherical Decompositions in Flat Space

We begin our study of the dimensional-reduction anomaly with an examination of one of the simplest and most familiar examples of dimensional reduction: the decomposition of a scalar field in flat space into spherical harmonics. This surprisingly nontrivial example illustrates most aspects of the dimensional-reduction anomaly, and allows their direct physical interpretation. At the same time, it is simple enough for exact results to be obtained, and will serve as a check on later calculations for the anomaly in curved spherically symmetric spaces.

We will be particularly interested in the role of the s mode (the $\ell = 0$ spherical harmonic) for quantum fields at nonzero temperatures. Accordingly, we begin in Section 3.2 with a direct calculation of the s-mode contribution to $\langle \hat{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu\nu} \rangle$ at finite temperature in four dimensions. We then consider in Section 3.3 how to solve the same problem via dimensional reduction to a two-dimensional theory. We will demonstrate that renormalization of the dimensionally reduced theory fails to yield the correct expectation values, and explain why. We will then show how the dimensional-reduction anomaly corrects the predictions of the dimensionally reduced theory to reproduce the expected results for four dimensions.

3.1 Spherical Dimensional Reductions

In section 2.2 we examined in a formal manner the dimensional reduction of a scalar field without specifying the nature of the mode decomposition. As preparation for Sections 3.2 and 3.3 and for Chapter 4, we rewrite the more important formulae from Section 2.2 for the special case of spherical decompositions in a four-dimensional space.

Choosing standard¹ angular coordinates (θ, η) , the line element for a fourdimensional spherically symmetric space may be written as

$$ds^{2} = g_{\mu\nu}(X^{\tau})dX^{\mu}dX^{\nu} = h_{ab}(x^{c})dx^{a}dx^{b} + \rho^{2}e^{-2\phi(x^{c})}\left(d\theta^{2} + \sin^{2}\theta \,d\eta^{2}\right) \,. \quad (3.1)$$

Here ρ is a constant with the dimensions of length. The radius of a two-sphere of fixed x^a is given by $r = \rho e^{-\phi(x^a)}$. In the general case h_{ab} is an arbitrary two-dimensional metric; in this chapter we consider only flat two-dimensional space.

Consider a massive scalar field propagating on the space (3.1) and obeying the field equation (2.55), where the potential V is also spherically symmetric. A natural procedure is to decompose the field into spherical harmonics $Y_{\ell m}(\theta, \eta)$,

$$\Phi(X) = \sum_{\ell=0}^{\infty} \sum_{m=-\ell}^{\ell} c_{\ell m} \varphi_{\ell}(x^{a}) \frac{Y_{\ell m}(\theta, \eta)}{r}, \qquad (3.2)$$

where the $c_{\ell m}$ are constants. Using the orthogonality and completeness of the $Y_{\ell m}(\theta, \eta)$ on the two-sphere, one easily verifies that the dimensional-reduction

¹We denote the azimuthal coordinate by η rather than ϕ to avoid confusion with the dilaton.

formulae from the previous chapter can be carried over directly for this decomposition using D = 4, n = 2, $R[\Omega] = 2$, and making the substitutions

$$Y_{\lambda \varrho}(y) \to Y_{\ell m}(\theta, \eta),$$
 (3.3)

$$\lambda \to \ell(\ell+1) \,, \tag{3.4}$$

$$\mathcal{N}_{\lambda} \to 2\ell + 1 \,, \tag{3.5}$$

$$V_{\Omega} \to 4\pi \,, \tag{3.6}$$

$$\sum_{\lambda} \to \sum_{\ell=0}^{\infty} . \tag{3.7}$$

In addition,

$$\sum_{m=-\ell}^{\ell} Y_{\ell m}(\theta,\eta) \overline{Y}_{\ell m}(\theta',\eta') = \frac{2\ell+1}{4\pi} P_{\ell}(\cos\lambda) .$$
(3.8)

Here P_{ℓ} is a Legendre polynomial and we now use λ to represent the angular separation of the points X, X', given implicitly by

$$\cos \lambda = \cos \theta \, \cos \theta' + \sin \theta \, \sin \theta' \, \cos(\eta - \eta') \,. \tag{3.9}$$

Following the analysis of Section 2.2, we see that φ_{ℓ} behaves as a field propagating in the two-dimensional space with metric h_{ab} . It satisfies the field equation (2.72) with induced potential

$$V_{\ell} = V + \frac{\ell(\ell+1)}{r^2} - \Delta\phi + (\nabla\phi)^2, \qquad (3.10)$$

and obeys Dirichlet boundary conditions wherever r = 0 (provided the manifold is regular there). The corresponding two-dimensional Green functions \mathcal{G}_{ℓ} are related to the four-dimensional G via

$$G(X, X') = \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r r'} P_{\ell}(\cos \lambda) \mathcal{G}_{\ell}(x, x'), \qquad (3.11)$$

while the renormalized quantities (2.102)-(2.106) take the following explicit form:

$$G_{\rm ren}(X,X') = \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi \tau \tau'} P_{\ell}(\cos\lambda) \left[\mathcal{G}_{\ell|\rm ren}(x,x') + \Delta \mathcal{G}_{\ell}(x,x') \right] , \qquad (3.12)$$

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r^2} \left[\langle \hat{\varphi}_{\ell}^2 \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_{\ell}^2 \rangle \right] , \qquad (3.13)$$

$$W_{\rm ren} = \sum_{\ell=0}^{\infty} (2\ell+1) \left[\mathcal{W}_{\ell|\rm ren} + \Delta \mathcal{W}_{\ell} \right] , \qquad (3.14)$$

$$\langle \hat{T}_{\mu\nu} \rangle_{\rm ren} = \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r^2} \left[\langle \hat{\mathcal{T}}_{\mu\nu}^{(\ell)} \rangle_{\rm ren} + \Delta \langle \hat{\mathcal{T}}_{\mu\nu}^{(\ell)} \rangle \right] \,. \tag{3.15}$$

$$\langle \hat{P} \rangle_{\rm ren} = \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r^2} \left[\langle \hat{\mathcal{P}}^{(\ell)} \rangle_{\rm ren} + \Delta \langle \hat{\mathcal{P}}^{(\ell)} \rangle \right] \,. \tag{3.16}$$

In each case the anomaly is the difference between the divergent subtraction terms for the dimensionally reduced theory and the mode-decomposed subtraction terms from four dimensions. For example, in analogy to (2.107) we find

$$\Delta \mathcal{G}_{\ell}(x,x') = \mathcal{G}_{\ell|\operatorname{div}}(x,x') - 2\pi r r' \int_{-1}^{1} d(\cos\lambda) P_{\ell}(\cos\lambda) G_{\operatorname{div}}(X,X') \,. \tag{3.17}$$

The remainder of this chapter is devoted to explicit calculations of these anomalies and elucidation of their importance for dimensional reductions in flat space. General spherically symmetric geometries are considered in Chapter 4.

3.2 s-mode in Flat Space at Finite Temperature

To better understand the role played by individual angular modes in spherical decompositions, we calculate the contribution made by the *s* modes to the square of the field operator and the stress tensor of a scalar field in flat fourdimensional spacetime at finite temperature. These follow the calculations of Balbinot, Fabbri, and Frolov for $\langle \hat{T}_{\mu\nu} \rangle$ in the massless case [73]. We use the metric signature (- + + +).

3.2.1 Formalism

In Section 2.2 we saw explicitly how the expectation values of the square of the field operator and the stress tensor for a scalar field in Euclidean space can be obtained from the Euclidean Green function. Formulae (2.60)-(2.62) hold in Lorentzian geometries as well with the replacement $G \rightarrow \frac{1}{2}G^{(1)}$, where $G^{(1)}$ is the Hadamard function for the quantum state of interest. Thus,

$$\langle \hat{\Phi}^2 \rangle = \lim_{X' \to X} \frac{1}{2} G^{(1)}(X, X'),$$
 (3.18)

$$\langle \hat{T}_{\mu\nu} \rangle = \lim_{X' \to X} \frac{1}{2} (\frac{1}{2} - \xi) \left(G^{(1)}_{\mu'\nu} + G^{(1)}_{\mu\nu'} \right) - \frac{1}{2} \xi \left(G^{(1)}_{\mu\nu} + G^{(1)}_{\mu'\nu'} \right) + \frac{1}{2} \xi R_{\mu\nu}[g] G^{(1)}$$

$$+ (\xi - \frac{1}{4}) g_{\mu\nu} \left[g^{\rho\sigma'} G^{(1)}_{\rho\sigma'} + (m^2 + \xi R[g]) G^{(1)} \right],$$

$$(3.19)$$

where $G_{\mu\nu'}^{(1)} \equiv g_{\nu}^{\rho'} \nabla_{\rho'} \nabla_{\mu} G^{(1)}(X, X')$, etc. Let us decompose the field operator $\hat{\Phi}$ in terms of a complete orthonormal set of modes $\{\Phi_{k\ell m}, \overline{\Phi}_{k\ell m}\},$

$$\hat{\Phi}(X) = \sum_{\ell=0}^{\infty} \sum_{m=-\ell}^{\ell} \int dk \left[\hat{a}_{k\ell m} \Phi_{k\ell m}(X) + \hat{a}^{\dagger}_{k\ell m} \overline{\Phi}_{k\ell m}(X) \right], \qquad (3.20)$$

where

$$\Phi_{k\ell m} = \varphi_{k\ell}(t,r) \, \frac{Y_{\ell m}(\theta,\eta)}{r} \,. \tag{3.21}$$

Under this spherical decomposition the four-dimensional field equation

$$\left(\Box - m^2\right)\Phi_{k\ell m} = 0 \tag{3.22}$$

reduces to

$$\left(-\partial_t^2 + \partial_r^2 - m^2 - \frac{\ell(\ell+1)}{r^2}\right)\varphi_{k\ell} = 0.$$
(3.23)

In particular, for the $\ell = 0$ mode the φ_{k0} are just plane waves. However, to account for the fact that $r \in [0, \infty)$, we must impose Dirichlet boundary conditions

on the $\varphi_{k\ell}$ at r = 0. The correctly normalized² s modes are then

$$\Phi_{k00} = \frac{-i}{2\pi r \sqrt{\omega_k}} e^{-i\omega_k t} \sin(kr) , \qquad (3.24)$$

where $k \in (0,\infty)$, and $\omega_k \equiv \sqrt{k^2 + m^2}$ is the energy eigenvalue of the mode $\Phi_{k\ell m}$.

It can be shown that choosing the time dependence of the modes such that $\Phi_{k\ell m} \propto e^{-i\omega_k t}$ guarantees that the vacuum state associated with these modes will be the ordinary Minkowski (zero-temperature) vacuum. The zero-temperature Hadamard function is then given by (2.27):

$$G^{(1)}_{\beta \to \infty}(X, X') = \sum_{\ell=0}^{\infty} \sum_{m=-\ell}^{\ell} \int_{0}^{\infty} dk \left(\Phi_{k\ell m} \overline{\Phi}'_{k\ell m} + \overline{\Phi}_{k\ell m} \Phi'_{k\ell m} \right) , \qquad (3.25)$$

with $\Phi' \equiv \Phi(X')$. Meanwhile, the Hadamard function at temperature $T = \beta^{-1}$ can be shown to be [44]

$$G_{\beta}^{(1)}(X,X') = \sum_{\ell=0}^{\infty} \sum_{m=-\ell}^{\ell} \int_{0}^{\infty} dk \left(\Phi_{k\ell m} \overline{\Phi}'_{k\ell m} + \overline{\Phi}_{k\ell m} \Phi'_{k\ell m} \right) \frac{1 + e^{-\beta \omega_{k}}}{1 - e^{-\beta \omega_{k}}}.$$
 (3.26)

In flat space with vanishing potential, a free quantum field is renormalized by subtracting from the bare quantity the corresponding result for the Minkowski vacuum state. Taking the difference of (3.26) and (3.25), one finds

$$\langle \hat{\Phi}^2 \rangle^\beta = \sum_{\ell=0}^\infty \left(2\ell + 1 \right) f^{(\ell)}, \qquad \langle \hat{T}_{\mu\nu} \rangle^\beta = \sum_{\ell=0}^\infty \left(2\ell + 1 \right) t^{(\ell)}_{\mu\nu}, \qquad (3.27)$$

where the contributions to $\langle \hat{\Phi}^2 \rangle^{\beta}$ and $\langle \hat{T}_{\mu\nu} \rangle^{\beta}$ due to modes of fixed ℓ with m = 0are

$$f^{(\ell)} = \int_0^\infty dk \, \frac{2}{\mathrm{e}^{\beta\omega_k} - 1} \, \Phi_{k\ell 0} \, \overline{\Phi}_{k\ell 0} \,, \qquad (3.28)$$

²The normalized *s*-modes are those satisfying

$$t_{\mu\nu}^{(\ell)} = \int_{0}^{\infty} dk \, \frac{2}{\mathrm{e}^{\beta\omega_{k}} - 1} \left[\left(\frac{1}{2} - \xi \right) \left(\partial_{\mu} \Phi_{k\ell 0} \partial_{\nu} \overline{\Phi}_{k\ell 0} + \partial_{\nu} \Phi_{k\ell 0} \partial_{\mu} \overline{\Phi}_{k\ell 0} \right) \right. \\ \left. + \left(2\xi - \frac{1}{2} \right) g_{\mu\nu} \left(g^{\rho\sigma} \partial_{\rho} \Phi_{k\ell 0} \partial_{\sigma} \overline{\Phi}_{k\ell 0} + m^{2} \Phi_{k\ell 0} \overline{\Phi}_{k\ell 0} \right) \right. \\ \left. - \xi \left(\Phi_{k\ell 0} (\nabla_{\mu} \nabla_{\nu} \overline{\Phi}_{k\ell 0}) + (\nabla_{\mu} \nabla_{\nu} \Phi_{k\ell 0}) \overline{\Phi}_{k\ell 0} \right) \right] .$$
(3.29)

Since each value of the azimuthal number m for a given ℓ gives the same contribution to $\langle \tilde{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu\nu} \rangle$, (3.28) and (3.29) are written for the m = 0 mode alone, and the degeneracy $\mathcal{N}_{\ell} = 2\ell + 1$ is accounted for in (3.27).

Since in the following we deal solely with the s mode, we drop the superscript (ℓ) and understand f, $t_{\mu\nu}$ as referring to the $\ell = 0$ mode only.

3.2.2 s-mode Contribution to $\langle \hat{\Phi}^2 \rangle^{\beta}$

Computing (3.28) for the $\ell = 0$ modes (3.24) we find

$$f = \frac{2}{(2\pi r)^2} \int_0^\infty dk \, \frac{1}{e^{\beta \sqrt{k^2 + m^2}} - 1} \, \frac{\sin^2(kr)}{\sqrt{k^2 + m^2}} \,. \tag{3.30}$$

This can be expressed in terms of the known integral

$$\int_0^\infty dk \, \frac{\mathrm{e}^{-\beta\sqrt{k^2+m^2}}}{\sqrt{k^2+m^2}} \cos(2kr) = K_0(m\sqrt{\beta^2+4r^2}) \tag{3.31}$$

by rewriting the β -dependent factor in (3.30) as a geometric series in $e^{-\beta\omega_k}$. The result is

$$f = \frac{1}{(2\pi r)^2} \sum_{n=1}^{\infty} \left[K_0(nm\beta) - K_0(m\sqrt{(n\beta)^2 + 4r^2}) \right] .$$
(3.32)

Massless limit

Taking the limit $m \to 0$ in (3.32) reduces the K_0 Bessel function to a logarithm:

$$f_{m=0} = \frac{1}{(2\pi r)^2} \frac{1}{2} \sum_{n=1}^{\infty} \ln \left[1 + \left(\frac{2r}{n\beta}\right)^2 \right]$$

= $\frac{1}{(2\pi r)^2} \frac{1}{2} \ln \frac{\sinh (2\pi r/\beta)}{(2\pi r/\beta)}.$ (3.33)

The same result could also be obtained by taking the $m \to 0$ limit in (3.30) before evaluating the integral. As expected, $f_{m=0}$ is finite at r = 0,

$$f_{m=0} \stackrel{r=0}{=} \frac{1}{12\beta^2}, \qquad (3.34)$$

while for large r it decreases as

$$f_{m=0} \stackrel{r \to \infty}{=} \frac{1}{4\pi\beta r} \,. \tag{3.35}$$

Low-temperature limit of massive theory

For $m\beta \gg 1$, the n > 1 terms in (3.32) are exponentially damped with respect to the n = 1 term, so we may neglect them to obtain

$$f_{m\beta\gg1} = \frac{1}{(2\pi r)^2} \left[K_0(m\beta) - K_0(m\sqrt{\beta^2 + 4r^2}) \right] .$$
 (3.36)

Here the limiting values for large and small r are

$$f_{m\beta\gg1} \stackrel{r=0}{=} \frac{1}{2\pi^2} \frac{m}{\beta} K_1(m\beta) , \qquad (3.37)$$

$$f_{m\beta\gg1} \stackrel{r\to\infty}{=} \frac{K_0(m\beta)}{(2\pi r)^2}.$$
 (3.38)

High-temperature limit of massive theory

The evaluation of the sum (3.32) in the previous cases has relied upon the asymptotic behaviour of the Bessel function K_0 for large and small values of its argument. However, in the high temperature limit, $m\beta \ll 1$, these arguments are neither always large nor always small, making the sums difficult to evaluate for general r. In spite of this, for very large and small values of mr we can still obtain f by making use of the sums

$$\sum_{n=1}^{\infty} \frac{K_1(nb)}{nb} = \frac{\pi^2}{6b^2} - \frac{\pi}{2b} - \frac{1}{4}\left(\gamma + \ln\frac{b}{4\pi}\right) + \frac{1}{8} + O(b^2), \quad (3.39)$$

$$\sum_{n=1}^{\infty} K_0(nb) = \frac{\pi}{2b} + \frac{1}{2} \left(\gamma + \ln \frac{b}{4\pi} \right) + O(b^2), \qquad (3.40)$$

$$\sum_{n=1}^{\infty} \left[\frac{K_0(nb)}{(nb)^2} + 2\frac{K_1(nb)}{(nb)^3} \right] = \frac{\pi^4}{45b^4} + \frac{\pi^2}{12b^2} - \frac{\pi}{6b} + \frac{1}{16} \left(\gamma + \ln \frac{b}{4\pi} \right) - \frac{3}{64} + O(b^2)$$
(3.41)

(see, e.g., 5.9.1.4 of [74]).

For $r \to 0$ a Taylor series expansion for $K_0(m\sqrt{(n\beta)^2 + 4r^2})$ yields

$$f_{m\beta\ll 1} \stackrel{r=0}{=} \frac{m^2}{2\pi^2} \sum_{n=1}^{\infty} \frac{K_1(nm\beta)}{nm\beta}$$
$$= \frac{1}{12\beta^2} - \frac{m}{4\pi\beta} - \frac{m^2}{8\pi^2} \left(\gamma + \ln\frac{m\beta}{4\pi}\right) + \frac{m^2}{16\pi^2} + O(\beta^2) \,. \tag{3.42}$$

Note that taking the $m \to 0$ limit returns the expected result (3.34) for a massless field.

For $mr \gg 1$, we can ignore the $K_0(m\sqrt{(n\beta)^2 + 4r^2})$ terms as exponentially damped with respect to the $K_0(nm\beta)$. We then have

$$f_{m\beta\ll 1} \stackrel{r\to\infty}{=} \frac{1}{(2\pi r)^2} \sum_{n=1}^{\infty} K_0(nm\beta)$$
$$= \frac{1}{4\pi r^2} \left[\frac{1}{2m\beta} + \frac{1}{2\pi} \left(\gamma + \ln \frac{m\beta}{4\pi} \right) + O(\beta^2) \right]. \quad (3.43)$$

Note that the $m \to 0$ limit is ill-defined. This is not surprising, since the assumption $mr \gg 1$ used to derive (3.43) excludes m = 0.

It should be noted that our results (3.34), (3.37), and (3.42) for the s-mode contribution to $\langle \hat{\Phi}^2 \rangle$ at r = 0 exactly reproduce the value obtained by summing over all modes [see (A.5), (A.7), (A.8) in Appendix A]. This occurs because as $r \to 0$, $\Phi_{klm} \propto r^{\ell}$, so only the s-mode makes a nonvanishing contribution to $\langle \hat{\Phi}^2 \rangle$ at the origin. Meanwhile, the $r \to \infty$ limit taken in (3.38), (3.43) equals $(4\pi r^2)^{-1}$ times the corresponding results for $\langle \hat{\Phi}^2 \rangle^{\beta}$ in two dimensions [see (A.17), (A.18)]. Similar phenomena will be seen with the stress tensor.

44

3.2.3 s-mode Contribution to $\langle \hat{\mathbf{T}}_{\mu\nu} \rangle^{\beta}$

Using (3.29), one finds that the stress tensor due to the *s*-modes has only diagonal components,

$$t_{\mu}^{\nu} = \frac{1}{(2\pi r)^2} \int_0^\infty dk \, \frac{2}{e^{\beta\omega_k} - 1} \frac{1}{\omega_k} \left[\frac{1}{2} k^2 T_{\mu}^{(1)\nu} + \frac{1}{2} k^2 \cos(2kr) T_{\mu}^{(2)\nu} + \frac{k}{2r} \sin(2kr) T_{\mu}^{(3)\nu} + \frac{1}{4r^2} (1 - \cos(2kr)) T_{\mu}^{(4)\nu} + \frac{m^2}{2} (1 - \cos(2kr)) T_{\mu}^{(5)\nu} \right], \qquad (3.44)$$

where the constant matrices $T^{(i)\,\nu}_{\ \mu}$ are

It is easy to verify that all of the integrals in (3.44) can be written as derivatives with respect to r of the integral (3.31) after expanding the β -dependent part as a geometric series in $e^{-\beta\omega_k}$. Defining $\mu_n \equiv \sqrt{(n\beta)^2 + 4r^2}$, we find

$$(2\pi r)^{2} t_{\mu}^{\nu} = \sum_{n=1}^{\infty} \frac{m}{n\beta} K_{1}(nm\beta) T_{\mu}^{(1)\nu} + \sum_{n=1}^{\infty} \left[\frac{m}{\mu_{n}} \left(1 - \frac{8r^{2}}{\mu_{n}^{2}} \right) K_{1}(m\mu_{n}) - \frac{4m^{2}r^{2}}{\mu_{n}^{2}} K_{0}(m\mu_{n}) \right] T_{\mu}^{(2)\nu} + \sum_{n=1}^{\infty} \frac{2m}{\mu_{n}} K_{1}(m\mu_{n}) T_{\mu}^{(3)\nu} + \frac{1}{2r^{2}} \sum_{n=1}^{\infty} \left[K_{0}(nm\beta) - K_{0}(m\mu_{n}) \right] T_{\mu}^{(4)\nu} + m^{2} \sum_{n=1}^{\infty} \left[K_{0}(nm\beta) - K_{0}(m\mu_{n}) \right] T_{\mu}^{(5)\nu}.$$
(3.50)

These sums can be evaluated for the massless, low-temperature, and hightemperature limits just as was done for the s-mode contribution to $\langle \hat{\Phi}^2 \rangle^{\beta}$. Moreover, it is easily verified that this stress tensor is conserved for all ξ at all r, and that it is traceless for $\xi = \frac{1}{6}$ and m = 0.

Massless limit

Taking $m \to 0$ in (3.50) replaces the Bessel functions by their small-argument limits. Using the sums

$$\sum_{n=1}^{\infty} [n^2 + x^2]^{-1} = \frac{\pi}{2x} \left[\coth(\pi x) - \frac{1}{\pi x} \right], \qquad (3.51)$$

$$\sum_{n=1}^{\infty} \ln\left[1 + \left(\frac{x}{n}\right)^2\right] = \ln\frac{\sinh\pi x}{\pi x}, \qquad (3.52)$$

one finds

$$(t_{\mu}^{\nu})_{m=0} = \frac{1}{4\pi r^2} \frac{\pi}{6\beta^2} T^{(1)\nu}_{\mu} + \left(\frac{1}{32\pi^2 r^4} - \frac{1}{8\beta^2 r^2 \sinh^2(2\pi r/\beta)}\right) T^{(2)\nu}_{\mu} + \left(\frac{1}{8\pi\beta r^3} \coth(2\pi r/\beta) - \frac{1}{16\pi^2 r^4}\right) T^{(3)\nu}_{\mu} + \frac{1}{16\pi^2 r^4} \ln\left\{\frac{\sinh(2\pi r/\beta)}{2\pi r/\beta}\right\} T^{(4)\nu}_{\mu}.$$
(3.53)

The same result is obtained by taking the $m \rightarrow 0$ limit before evaluating the integral over k in (3.44).

This stress tensor is finite at r = 0:

$$(t_{\mu}^{\nu})_{m=0} \stackrel{r=0}{=} -\frac{\pi^2}{180\beta^4} \left(6T^{(2)\nu}_{\ \mu} + 4T^{(3)\nu}_{\ \mu} + T^{(4)\nu}_{\ \mu} \right) .$$
 (3.54)

In particular, for $\xi = \frac{1}{6}$,

$$(t_{\mu}{}^{\nu})_{m=0} \stackrel{r=0}{=} \frac{\pi^2}{36\beta^4} \begin{pmatrix} -1 & 0 & 0 & 0 \\ 0 & \frac{1}{3} & 0 & 0 \\ 0 & 0 & \frac{1}{3} & 0 \\ 0 & 0 & 0 & \frac{1}{3} \end{pmatrix} .$$
 (3.55)

Comparing to (A.11), we conclude that at the origin $\frac{5}{6}$ of the total stress tensor (i.e., that obtained by summing over all ℓ) is due to the *s* mode. The difference is made up by the $\ell = 1$ mode, which can make a nonvanishing contribution at the origin due to the fact that the expression (3.29) for the stress tensor contains up to two derivatives per term quadratic in the field.

Finally, at large r we have

Note that the leading order term is just $(4\pi r^2)^{-1}$ times the stress tensor for a massless scalar field in two dimensions, (A.19).

Low-temperature limit of massive theory

For low temperatures, $m\beta \gg 1$, the arguments $nm\beta$, $m\sqrt{(n\beta)^2 + 4r^2}$ of the Bessel functions in (3.50) are large for all n and so the Bessel functions are exponentially damped. The leading contribution to the sums then comes from the n = 1 term alone, as we saw earlier for $\langle \hat{\Phi}^2 \rangle^{\beta}$. Recalling that $\mu_1 = \sqrt{\beta^2 + 4r^2}$, we find

$$(2\pi r)^{2} (t_{\mu}^{\nu})_{m\beta \gg 1} = \frac{m}{\beta} K_{1}(m\beta) T_{\mu}^{(1)\nu} + \left[\frac{m}{\mu_{1}} \left(1 - \frac{8r^{2}}{\mu_{1}^{2}} \right) K_{1}(m\mu_{1}) - \frac{4m^{2}r^{2}}{\mu_{1}^{2}} K_{0}(m\mu_{1}) \right] T_{\mu}^{(2)\nu} + \frac{2m}{\mu_{1}} K_{1}(m\mu_{1}) T_{\mu}^{(3)\nu} + \frac{1}{2r^{2}} \left[K_{0}(m\beta) - K_{0}(m\mu_{1}) \right] T_{\mu}^{(4)\nu} + m^{2} \left[K_{0}(m\beta) - K_{0}(m\mu_{1}) \right] T_{\mu}^{(5)\nu}.$$
(3.57)

As in the massless case, the stress tensor is finite at r = 0:

$$(t_{\mu}^{\nu})_{m\beta\gg1} \stackrel{r=0}{=} - \left(6T^{(2)\nu}_{\mu} + 4T^{(3)\nu}_{\mu} + T^{(4)\nu}_{\mu} \right) \frac{1}{(2\pi)^2} \left[\frac{m^2}{\beta^2} K_0(m\beta) + \frac{2m}{\beta^3} K_1(m\beta) \right]$$
$$+ T^{(5)\nu}_{\mu} \frac{1}{(2\pi)^2} \frac{2m^3}{\beta} K_1(m\beta) .$$
(3.58)

In particular, for $\xi = \frac{1}{6}$,

$$(t_{\mu}^{\nu})_{m\beta\gg1} \stackrel{r\to 0}{=} 5 \begin{pmatrix} -1 & 0 & 0 & 0 \\ 0 & \frac{1}{3} & 0 & 0 \\ 0 & 0 & \frac{1}{3} & 0 \\ 0 & 0 & 0 & \frac{1}{3} \end{pmatrix} \frac{1}{(2\pi)^2} \left[\frac{m^2}{\beta^2} K_0(m\beta) + \frac{2m}{\beta^3} K_1(m\beta) \right]$$

Let us compare this result to the total stress tensor (due to all ℓ) for a massive field at low temperature, (A.14). We see that the second term in (A.14), which vanishes in the massless case, is due entirely to the s mode, while $\frac{5}{6}$ of the first term is due to the s mode.

Meanwhile, as $r \to \infty$ we find

$$(t_{\mu}^{\nu})_{m\beta\gg1} \stackrel{r\to\infty}{=} \frac{m^2}{(2\pi r)^2} \left[K_0(m\beta) \ T^{(5)\nu}_{\ \mu} - \frac{K_1(m\beta)}{m\beta} \ T^{(1)\nu}_{\ \mu} \right] + O\left(\frac{1}{r^4}\right) .$$
 (3.60)

By comparing to (A.20) we see that the leading-order terms are just $(4\pi r^2)^{-1}$ times the low-temperature stress tensor for a scalar field in two dimensions.

High-temperature limit of massive theory

For $m\beta \ll 1$ we can evaluate the stress tensor for $r \to 0$ and $r \to \infty$ only.

Let us consider the case $r \rightarrow 0$ first. By expanding the r-dependent Bessel functions in (3.50) for small r and making use of the sums (3.39)–(3.41) one finds

$$(t_{\mu}^{\nu})_{m\beta \ll 1} \stackrel{r \equiv 0}{=} - \left(6T_{\mu}^{(2)\nu} + 4T_{\mu}^{(3)\nu} + T_{\mu}^{(4)\nu} \right) \left[\frac{\pi^2}{180\beta^4} + \frac{m^2}{48\beta^2} - \frac{m^3}{24\pi\beta} \right] + \frac{m^4}{64\pi^2} \left(\gamma + \ln\frac{b}{4\pi} \right) - \frac{3m^4}{256\pi^2} \right] + T_{\mu}^{(5)\nu} \left[\frac{m^2}{12\beta^2} - \frac{m^3}{4\pi\beta} - \frac{m^4}{8\pi^2} \left(\gamma + \ln\frac{b}{4\pi} \right) + \frac{m^4}{16\pi^2} \right].$$
(3.61)

Note that taking the $m \to 0$ limit returns the massless result (3.54). Also, if we take $\xi = \frac{1}{6}$ and compare to the total stress tensor for a high-temperature field, (A.15), we see that the second term in (A.15) is entirely due to the *s* mode, while $\frac{5}{6}$ of the first term is due to the *s* mode.

For $mr \gg 1$, all of the Bessel functions $K_j(m\mu_n)$ will be exponentially small compared to the $K_j(mn\beta)$, and may be ignored. We then find

$$(t_{\mu}^{\nu})_{m\beta\ll 1} \stackrel{mr\to\infty}{=} \frac{1}{(2\pi r)^2} \left[\frac{\pi^2}{6\beta^2} - \frac{\pi m}{2\beta} - \frac{m^2}{4} \left(\gamma + \ln\frac{m\beta}{4\pi} \right) + \frac{m^2}{8} \right] T^{(1)\nu}_{\mu} + \frac{1}{(2\pi r)^2} \left[\frac{\pi m}{2\beta} + \frac{m^2}{2} \left(\gamma + \ln\frac{m\beta}{4\pi} \right) \right] T^{(5)\nu}_{\mu} + O\left(\frac{1}{r^4}\right) .$$

$$(3.62)$$

Again, the leading-order terms are just $(4\pi r^2)^{-1}$ times the corresponding result for a scalar field in two dimensions, (A.21). As well, for $m \to 0$, this leading part reproduces the $1/r^2$ part of the stress tensor from the massless case, (3.56).

3.3 The Dimensional-Reduction Anomaly in Flat Space

In the previous section, we calculated directly the s-mode contribution to the expectation values $\langle \hat{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu\nu} \rangle$ at finite temperature in four dimensions. In this section we attempt to solve the same problem using dimensional reduction. We show that naïve dimensional reduction fails to produce the correct expectation values for flat space. We then examine the dimensionally reduced theory in detail and show how the anomaly corrects its predictions to yield expectation values in agreement with those of the previous section.

Consider a quantum field Φ in four-dimensional flat space. For convenience, we will calculate the anomaly in Euclidean space and continue the final results back to Lorentzian signature. We assume that the field satisfies the field equation (2.55), where the potential V vanishes inside the region of interest and is spherically symmetric outside. In this case, the Green function for a given state is renormalized by subtracting the Green function for the Euclidean vacuum. In four dimensions the latter is³

$$G_{\rm div}(X, X') = \frac{m}{4\pi^2 \sqrt{2\sigma}} K_1(m\sqrt{2\sigma}),$$
 (3.63)

where σ is one-half the square of the geodesic distance between X and X', and K_1 is a modified Bessel function. In spherical coordinates $X^{\mu} = (t, r, \theta, \eta)$ the line element is

$$ds^{2} = dt^{2} + dr^{2} + r^{2} \left(d\theta^{2} + \sin^{2}\theta d\eta^{2} \right) , \qquad (3.64)$$

and

$$2\sigma = (t - t')^{2} + (r - r')^{2} + 2rr'(1 - \cos \lambda) , \qquad (3.65)$$

where λ , the angle between X and X', is given by (3.9).

If there exists a boundary Σ surrounding the region \mathcal{M} under consideration and the field obeys a non-trivial boundary condition at Σ , or equation (2.55) includes a non-vanishing potential V which vanishes in \mathcal{M} , then the Green function G(X, X') will differ from G_{div} . The renormalized Green function is then

$$G_{\rm ren}(X, X') = G(X, X') - G_{\rm div}(X, X').$$
(3.66)

It is evident that $\langle \hat{\Phi}^2 \rangle_{\text{ren}}$ and $\langle \hat{T}_{\mu\nu} \rangle_{\text{ren}}$ vanish in the absence of the external field V and boundaries.

If before renormalizing we first decompose the field $\hat{\Phi}$ into spherical harmonics as in (3.2), we will be left with an effective field $\hat{\varphi}_{\ell}$ propagating on the twodimensional space with line element $ds^2 = dt^2 + dr^2$, where $t \in (-\infty, \infty)$ and $r \in [0, \infty)$. The geodesic distance function σ is simply $\frac{1}{2}[(t-t')^2 + (r-r')^2]$, and from (3.10) the induced potential is

$$V_{\ell} = \frac{\ell(\ell+1)}{r^2}, \qquad (3.67)$$

 $^{{}^{3}}$ Eq. (3.63) may be obtained using the Schwinger-DeWitt expansion, as described in Section 4.1.

as $\Delta \phi - (\nabla \phi)^2 = 0$ for the line element (3.64). Though this induced potential is nonvanishing, it does not influence the divergences in two dimensions, as we shall see later. As a result, the theory is renormalized by subtracting the twodimensional vacuum Green function⁴

$$\mathcal{G}_{\ell|\text{div}}(x,x') = \frac{1}{2\pi} K_0(m\sqrt{(t-t')^2 + (r-r')^2}).$$
(3.68)

By this point one might have noticed a curious feature of the dimensional reduction. For the original theory in four dimensions, one typically expects the renormalized stress tensor and field fluctuations to be finite everywhere (such as for a finite-temperature state). In particular, for the Euclidean vacuum state, all expectation values should vanish identically. By (3.2) and (3.11), finiteness at r = 0 in four dimensions implies that the dimensionally reduced field and Green function must vanish there. On general grounds, however, one expects that this boundary condition will produce infinite vacuum polarization which causes the dimensionally reduced stress tensor to diverge as $r \to 0$ [75]. Thus, if one attempts to calculate $\langle \hat{T}_{\mu\nu} \rangle_{\rm ren}$ by naïvely summing over the $\langle \hat{T}_{\mu\nu}^{(\ell)} \rangle_{\rm ren}$, the resulting four-dimensional stress tensor will diverge⁵ at r = 0, even for the vacuum state! A key result of this section is that the dimensional-reduction anomaly cancels these divergences in the two-dimensional theory to yield finite results in four dimensions.

The dimensional-reduction anomaly arises because of the inequivalence of the divergent subtraction terms for the higher- and lower-dimensional theories. To

$$\sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r^2} \Delta \langle \hat{\varphi}_{\ell}^2 \rangle, \qquad \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r^2} \langle \hat{\varphi}_{\ell}^2 \rangle_{\rm ren}$$

are divergent. Similar effects were noticed in [6, 11, 12].

⁴Eq. (3.68) can be obtained using the Schwinger-DeWitt expansion in Section 4.1.

⁵In fact, one can show that the sum over ℓ will diverge at all r. For example, the anomaly $\Delta \langle \hat{\varphi}_{\ell}^2 \rangle$ varies as ℓ^0 as $\ell \to \infty$, and hence sums like

illustrate this, we decompose G_{div} into spherical harmonics and compare to $\mathcal{G}_{\ell|div}$. Defining the mode decomposition by

$$G_{\rm div}(X, X') = \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r r'} P_{\ell}(\cos\lambda) G_{\rm div|\ell}(x, x')$$
(3.69)

in accordance with (3.11), we have

$$G_{\operatorname{div}|\ell}(x,x') = 2\pi r r' \int_{-1}^{1} d(\cos\lambda) P_{\ell}(\cos\lambda) G_{\operatorname{div}}(X,X') \,. \tag{3.70}$$

Inserting (3.63) into (3.70) and using the well-known integral representation for K_{ν} ,

$$\int_0^\infty dx \, x^{-1-\nu} \exp\left\{-x - \frac{\alpha^2}{4x}\right\} = 2\left(\frac{2}{\alpha}\right)^\nu K_\nu(\alpha) \tag{3.71}$$

(see Appendix C.3.1), the integral

$$\int_{-1}^{1} dz \ P_{\ell}(z) \ \mathrm{e}^{p(1-z)} = (-1)^{\ell} \mathrm{e}^{p} \sqrt{\frac{2\pi}{p}} I_{\ell+1/2}(p) , \qquad (3.72)$$

where $I_{\ell+1/2}$ is a modified Bessel function (see e.g. [74], vol.2, eq.2.17.5.2), and the representation

$$I_{\ell+1/2}(p) = \frac{1}{\sqrt{2\pi p}} \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} \frac{1}{(2p)^k} \left[(-1)^k \mathrm{e}^p - (-1)^\ell \mathrm{e}^{-p} \right] , \qquad (3.73)$$

(see, for example, 8.467 of [76]), we obtain

$$G_{div|\ell}(x,x') = \frac{1}{2\pi} \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} \left[(-1)^k \frac{[2\sigma^-]^{k/2}}{(2mrr')^k} K_k(m\sqrt{2\sigma^-}) - (-1)^{\ell} \frac{[2\sigma^+]^{k/2}}{(2mrr')^k} K_k(m\sqrt{2\sigma^+}) \right], \qquad (3.74)$$

where

$$\sigma^{\pm} \equiv \frac{1}{2} [(t - t')^2 + (r \pm r')^2]. \qquad (3.75)$$

Let us compare this result for the mode-decomposed subtraction terms from four dimensions with the subtraction term for the two-dimensional theory, (3.68). While $\mathcal{G}_{\ell|\text{div}}$ is the free-field Green function in two dimensions, it is not difficult to verify that $G_{\text{div}|\ell}$ is the Green function for a field propagating in the centrifugal barrier potential (3.67) and obeying Dirichlet boundary conditions⁶ at r = 0. The anomaly in the Green function, given by

$$\Delta \mathcal{G}_{\ell}(x, x') = \mathcal{G}_{\ell|\operatorname{div}}(x, x') - \mathcal{G}_{\operatorname{div}|\ell}(x, x'), \qquad (3.76)$$

is then simply the negative of the renormalized Green function for a field propagating in the centrifugal barrier potential (3.67) in two dimensions and obeying Dirichlet boundary conditions at r = 0. The infinite vacuum polarization effects in the effective two-dimensional theory due to the boundary at r = 0 will thus be exactly canceled by the dimensional-reduction anomaly. In particular, for the four-dimensional vacuum state the anomaly exactly cancels the renormalized two-dimensional Green function, yielding vanishing expectation values in four dimensions. Failure to include the anomaly will leave one with divergent results for four-dimensional expectation values, even for the vacuum state.

The anomaly in the Green function can be used to calculate the anomalies in $\langle \hat{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu\nu} \rangle$. For $\langle \hat{\Phi}^2 \rangle$ we find

$$\Delta \langle \hat{\varphi}_{\ell}^2 \rangle = \lim_{x' \to x} \left[\mathcal{G}_{\ell | \operatorname{div}}(x, x') - G_{\operatorname{div}|\ell}(x, x') \right]$$
(3.77)

$$= \frac{1}{4\pi} \sum_{k=1}^{\ell} \frac{(\ell+k)!}{(\ell-k)!} \frac{1}{k} \frac{(-1)^{k+1}}{(mr)^{2k}} + \frac{(-1)^{\ell}}{2\pi} \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} \frac{K_k(2mr)}{(mr)^k}.$$
 (3.78)

For example, for the first three modes the anomalies are

$$\Delta \langle \hat{\varphi}_{\ell=0}^2 \rangle = \frac{1}{2\pi} K_0(2mr) , \qquad (3.79)$$

$$\Delta \langle \hat{\varphi}_{\ell=1}^2 \rangle = \frac{1}{2\pi} \left[\frac{1}{(mr)^2} - K_0(2mr) - \frac{2}{(mr)} K_1(2mr) \right], \qquad (3.80)$$

$$\Delta \langle \hat{\varphi}_{\ell=2}^2 \rangle = \frac{1}{2\pi} \left[\frac{3}{(mr)^2} - \frac{6}{(mr)^4} + K_0(2mr) + \frac{6}{(mr)} K_1(2mr) \right]$$

⁶Since (3.72) is finite for p = 0, $G_{div|\ell}$ obeys Dirichlet boundary conditions at r = 0, r' = 0 due to (3.70).

$$+\frac{12}{(mr)^2}K_2(2mr)\bigg].$$
 (3.81)

Note that the anomaly diverges logarithmically as $r \to 0$, and vanishes as $r \to \infty$. The divergence at r = 0 precisely cancels the corresponding divergence in the renormalized two-dimensional theory due to the boundary condition.

The expressions for the anomaly in the stress tensor are more complicated, so we show the results for the $\ell = 0$ mode only. Substituting (3.76) into (2.82), (2.83) we obtain

$$\begin{aligned} \Delta \langle \hat{\mathcal{T}}_{t}^{(0)t} \rangle &= \frac{m^{2}}{2\pi} \left[(4\xi - 1)K_{0}(2mr) + (12\xi - 2)\frac{K_{1}(2mr)}{(2mr)} + (8\xi - 2)\frac{K_{0}(2mr)}{(2mr)^{2}} \right], \end{aligned}$$
(3.82)

$$\Delta \langle \hat{\mathcal{T}}^{(0)r}_{r} \rangle = \frac{m^2}{2\pi} \left[(2 - 8\xi) \frac{K_1(2mr)}{(2mr)} + (2 - 16\xi) \frac{K_0(2mr)}{(2mr)^2} \right], \quad (3.83)$$

$$\Delta \langle \hat{\mathcal{P}}^{(0)} \rangle = \frac{m^2}{2\pi} \left[(4\xi - 1)K_0(2mr) + (16\xi - 3)\frac{K_1(2mr)}{(2mr)} + (16\xi - 2)\frac{K_0(2mr)}{(2mr)^2} \right].$$
(3.84)

In the $m \rightarrow 0$ limit the anomalous stress tensor becomes

$$\Delta \langle \hat{\mathcal{T}}_{t}^{(0)t} \rangle = \frac{1}{4\pi r^{2}} \left[(1 - 4\xi) \{ \gamma + \ln mr \} + (6\xi - 1) \right], \qquad (3.85)$$

$$\Delta \langle \hat{\mathcal{T}}_{r}^{(0)r} \rangle = \frac{1}{4\pi r^{2}} \left[(8\xi - 1) \{ \gamma + \ln mr \} + (1 - 4\xi) \right], \qquad (3.86)$$

$$\Delta \langle \hat{\mathcal{P}}^{(0)} \rangle = \frac{1}{8\pi r^2} \left[(2 - 16\xi) \left\{ \gamma + \ln mr \right\} + (16\xi - 3) \right], \qquad (3.87)$$

where the *m* in the logarithms must now be considered an arbitrary parameter. Note that the anomalous stress tensor diverges as $\ln(r)/r^2$ as $r \to 0$, and vanishes as $r \to \infty$. As was the case for $\Delta \langle \hat{\varphi}_{\ell}^2 \rangle$, the divergence at r = 0 precisely cancels the corresponding divergence in the renormalized two-dimensional theory due to the boundary condition, leaving a finite stress tensor in four dimensions.

3.3.1 Scalar Field at Finite Temperature

Let us return to the spherical decomposition of a scalar field at temperature $T = \beta^{-1}$ in four-dimensional flat space, and use this case as an illustration of the role of the dimensional-reduction anomaly. From the preceding discussion, the dimensional reduction of a free scalar field with vanishing potential produces a dimensionally reduced field $\hat{\varphi}_{\ell}$ propagating in the centrifugal barrier potential $\ell(\ell+1)/r^2$ and satisfying Dirichlet boundary conditions at r = 0. The bare zero-temperature Green function for this theory is just (3.74). The finite-temperature Green function is most conveniently obtained by recalling that in Euclidean space the temperature dependence can be realized by requiring the field to be periodic in (imaginary) time with period β . Thus, the finite-temperature Green function is not convenient from the zero-temperature function (3.74) by the replacement $(t - t') \rightarrow (t - t' + n\beta)$ and summing over all integers n:

$$\mathcal{G}_{\ell}^{\beta}(x,x') = \sum_{n=-\infty}^{\infty} \frac{1}{2\pi} \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} \left[(-1)^{k} \frac{[2\sigma_{n}^{-}]^{k/2}}{(2mrr')^{k}} K_{k}(m\sqrt{2\sigma_{n}^{-}}) - (-1)^{\ell} \frac{[2\sigma_{n}^{+}]^{k/2}}{(2mrr')^{k}} K_{k}(m\sqrt{2\sigma_{n}^{+}}) \right], \qquad (3.88)$$

where

$$\sigma_n^{\pm} \equiv \frac{1}{2} [(t - t' + n\beta)^2 + (r \pm r')^2]. \qquad (3.89)$$

The zero-temperature result is the n = 0 term, $G_{div|\ell}(x, x')$. The renormalized value of $\langle \hat{\varphi}_{\ell}^2 \rangle$ for the dimensionally reduced field is then

$$\begin{aligned} \langle \hat{\varphi}_{\ell}^{2} \rangle_{ren}^{\beta} &= \lim_{x' \to x} \left[\mathcal{G}_{\ell}^{\beta}(x, x') - \mathcal{G}_{\ell | div}(x, x') \right] \\ &= 2 \sum_{n=1}^{\infty} \frac{1}{2\pi} \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k! (\ell-k)!} \left[(-1)^{k} \left(\frac{n\beta}{2mr^{2}} \right)^{k} K_{k}(nm\beta) \right. \\ &\left. - (-1)^{\ell} \left(\frac{\sqrt{(n\beta)^{2} + (2r)^{2}}}{2mr^{2}} \right)^{k} K_{k}(m\sqrt{(n\beta)^{2} + (2r)^{2}}) \right] \\ &\left. + \lim_{x' \to x} \left[G_{div|\ell}(x, x') - \mathcal{G}_{\ell | div}(x, x') \right] \end{aligned}$$
(3.90)
Note that we have separated the n = 0 term and grouped it with the twodimensional renormalization term, $\mathcal{G}_{\ell|\text{div}}$. This last line is precisely the negative of the anomaly $\Delta \langle \varphi_{\ell}^2 \rangle$, (3.77). As we have seen, this last part will produce a contribution to $\langle \varphi_{\ell}^2 \rangle$ which diverges as $r \to 0$. However, in calculating the renormalized $\langle \hat{\Phi}^2 \rangle^{\beta}$ in four dimensions this term is precisely cancelled by the dimensional-reduction anomaly:

$$\langle \hat{\Phi}^2 \rangle_{\text{ren}}^{\beta} = \sum_{\ell=0}^{\infty} \frac{2\ell+1}{4\pi r^2} \left[\langle \varphi_{\ell}^2 \rangle_{\text{ren}}^{\beta} + \Delta \langle \varphi_{\ell}^2 \rangle \right]$$

$$= \sum_{\ell=0}^{\infty} \frac{2\ell+1}{4\pi r^2} 2 \sum_{n=1}^{\infty} \frac{1}{2\pi} \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} \left[(-1)^k \left(\frac{n\beta}{2mr^2} \right)^k K_k(nm\beta) - (-1)^{\ell} \left(\frac{\sqrt{(n\beta)^2 + (2r)^2}}{2mr^2} \right)^k K_k(m\sqrt{(n\beta)^2 + (2r)^2}) \right].$$

$$(3.92)$$

The remaining terms are the finite, temperature-dependent contributions to $\langle \hat{\Phi}^2 \rangle_{ren}^{\beta}$ in four dimensions. For example, the $\ell = 0$ mode alone makes a contribution of

$$f^{(\ell=0)} = \frac{1}{4\pi^2 r^2} \sum_{n=1}^{\infty} \left[K_0(nm\beta) - K_k(m\sqrt{(n\beta)^2 + (2r)^2}) \right].$$
(3.94)

This is precisely the result (3.32) obtained in the previous section by working directly in four dimensions.

To summarize, we have demonstrated that expectation values such as $\langle \hat{\Phi}^2 \rangle_{ren}$ and $\langle \hat{T}_{\mu\nu} \rangle_{ren}$ can be expressed as sums of the corresponding renormalized quantities from the two-dimensional theory using relations like (3.12)–(3.16). However, to obtain the correct result we see that each term in the decomposition must be modified by adding a state-independent⁷ anomaly term, such as (3.78), (3.82)– (3.84). Failure to account for these extra terms would result, for example, in nonzero expectation values in the Minkowski state. We conclude that one can

⁷See also the discussion at the end of Chapter 5.

model correctly quantum fields using lower-dimensional theories, but only if the dimensional-reduction anomaly is accounted for.

Chapter 4

Spherical Decompositions in Curved Space

In the previous chapter we demonstrated the existence of the dimensionalreduction anomaly using the simple example of spherical decompositions in flat space. We saw that the anomaly has to be accounted for if the dimensionally reduced theory is to yield correct expectation values for the physical theory in four dimensions. In this chapter we extend our analysis to the more general case of spherical decompositions in *curved* spherically symmetric spaces. These include the very interesting example of the Schwarzschild black hole, which we shall consider at some length.

We begin in Section 4.1 with a detailed derivation of the dimensional-reduction anomaly in $\langle \hat{\Phi}^2 \rangle$. The techniques developed there are then applied in Section 4.2 to the calculation of the anomaly in the effective action. In Section 4.3 we derive the anomalous stress tensor for a general spherical decomposition, and in Section 4.4 apply these results to the interesting case of the *s*-mode in Schwarzschild spacetime.

4.1 The Dimensional-Reduction Anomaly in $\langle \hat{\Phi}^2 angle$

While our chief aim is the study of the dimensional-reduction anomaly in the effective action, in order to illustrate more clearly the effect of curvature on local reduction anomalies we begin with the simpler case of the anomaly for $\langle \hat{\Phi}^2 \rangle_{ren}$.

4.1.1 Structure of Divergences

We take as our system a massive scalar field with arbitrary coupling to the fourdimensional scalar curvature, described by the field equation (2.55) with

$$V = \xi^4 R[g] \,. \tag{4.1}$$

We assume that the space of interest is Euclidean and described by the line element (3.1).

As we saw in Section 3.1, under the dimensional reduction (3.2) the quantum field $\hat{\Phi}$ reduces to a collection of effective fields $\hat{\varphi}_{\ell}$ in the two-dimensional space with metric h_{ab} . These dimensionally reduced fields propagate in the induced potential (3.10),

$$V_{\ell} = \xi^{4}R[g] + \frac{\ell(\ell+1)}{r^{2}} - \Delta\phi + (\nabla\phi)^{2}$$

= $\xi R[h] + [\ell(\ell+1) + 2\xi] \rho^{-2}e^{2\phi} + (4\xi - 1)\Delta_{h}\phi + (1 - 6\xi)(\nabla\phi)^{2}, \quad (4.2)$

where $r = \rho e^{-\phi}$, and vanish at any regular points of the manifold where r = 0. We wish to compute the anomaly associated with renormalizing this dimensionally reduced theory versus (2.55).

Since the dimensional-reduction anomaly is connected with the divergences, let us discuss first the general structure of these divergences. The most convenient way for our purposes is to use the so-called Schwinger-DeWitt proper-time formalism [51], [41], [42]. In this approach, one can write the Green function G(X, X') and effective action W in the form

$$G(X, X') = \int_0^\infty ds \, K(X, X'|s) \,, \tag{4.3}$$

$$W = -\frac{1}{2} \int_0^\infty \frac{ds}{s} \int d^4 X \sqrt{g} \, K(X, X|s) \,, \tag{4.4}$$

where K(X, X'|s) is the heat kernel for the operator F, defined by (2.48), (2.49). Analogous formulae also hold for the dimensionally reduced theory with operator \mathcal{F}_{ℓ} , heat kernel $\mathcal{K}_{\ell}(x, x'|s)$, Green function $\mathcal{G}_{\ell}(x, x')$ and effective action \mathcal{W}_{ℓ} .

The divergences in both the Green function and the effective action come from the $s \rightarrow 0$ limit of the s integral. The advantage of the heat kernel formulation in curved space is that the small-s behaviour of the heat kernel is known for arbitrary curved spaces of any dimension, in the form of the Schwinger-DeWitt expansion [41, 42, 51]. In four dimensions this is

$$K(X, X'|s) = \frac{D^{\frac{1}{2}}(X, X')}{(4\pi s)^2} \exp\left\{-m^2 s - \frac{\sigma(X, X')}{2s}\right\} \sum_{i=0}^{\infty} a_i(X, X') s^i.$$
(4.5)

Here again $\sigma(X, X')$ is one-half of the square of the geodesic distance between the points X and X' (sometimes referred to as the Synge world function [77]), while D(X, X') is the Van Vleck-Morette determinant [78, 79],

$$D(X, X') = \frac{1}{\sqrt{g(X)}\sqrt{g(X')}} \det \left[-\frac{\partial}{\partial X^{\mu}} \frac{\partial}{\partial X'^{\nu}} \sigma(X, X') \right].$$
(4.6)

The a_n are the Schwinger-DeWitt coefficients for the operator F of (2.55) with potential (4.1). In the coincidence limit $X' \to X$ the first few of these are [41]

$$a_0^{\Box-\xi^4 R} = 1, (4.7)$$

$$a_1^{\Box-\xi^4 R} = \left(\frac{1}{6} - \xi\right)^4 R, \qquad (4.8)$$

$$a_{2}^{\Box-\xi^{4}R} = \frac{1}{180} \left[{}^{4}R_{\alpha\beta\gamma\delta} {}^{4}R^{\alpha\beta\gamma\delta} - {}^{4}R_{\alpha\beta} {}^{4}R^{\alpha\beta} + \Box^{4}R \right] + \frac{1}{6} \left(\frac{1}{6} - \xi \right) \Box^{4}R + \frac{1}{2} \left(\frac{1}{6} - \xi \right)^{2} ({}^{4}R)^{2}.$$
(4.9)

Note that, as promised, the a_n are purely local functions of the curvature, the potential, and their covariant derivatives. They are independent of the field mass, as this is accounted for separately in (4.5).

For the two-dimensional operator \mathcal{F}_{ℓ} of (2.72) with induced potential (4.2) we shall need the Schwinger-DeWitt expansion of the heat kernel only in the coincidence limit. This is

$$\mathcal{K}_{\ell}(x,x|s) = \frac{1}{4\pi s} \exp\left\{-m^2 s\right\} \left[1 + s\left(\frac{1}{6}R - V_{\ell}\right) + \cdots\right].$$
 (4.10)

Considering (4.3), (4.4), it is clear that in four dimensions the divergences in G(W) arise from the first two¹ (three) terms in the Schwinger-DeWitt expansion for K, while in two dimensions we need consider only the first term (first two terms) in \mathcal{K}_{ℓ} . The anomaly in $\langle \hat{\Phi}^2 \rangle$ or W can then be calculated by mode-decomposing the appropriate terms from K, comparing to the heat kernel \mathcal{K}_{ℓ} for the dimensionally reduced theory, and finally integrating the difference over s according to (4.3) or (4.4).

4.1.2 Calculation of the Anomaly

Let us begin with the anomaly in $\langle \hat{\Phi}^2 \rangle$. The divergent part of the Green function in four dimensions is given by

$$G_{\rm div}(X,X') = \int_0^\infty ds \, K_{\rm div}(X,X'|s) \,, \tag{4.11}$$

where K_{div} consists of the first two terms of (4.5):

$$K_{\rm div}(X, X'|s) = \frac{1}{(4\pi s)^2} \exp\left\{-m^2 s - \frac{\sigma}{2s}\right\} \left[\Re_0^{\Box - \xi^4 R}(X, X') + s\Re_1^{\Box - \xi^4 R}(X, X')\right] \,. \tag{4.12}$$

¹Comparing to (3.63), one sees that the integral representation used for G_{div} in Section 3.3 was just the heat kernel representation (4.3) with (4.5)–(4.8).

Here we use the convenient notation

$$\Re_n^{\Box-\xi^{4R}}(X,X') \equiv D^{\frac{1}{2}}(X,X') \ a_n^{\Box-\xi^{4R}}(X,X') \ . \tag{4.13}$$

In principle, the anomaly in $\langle \hat{\Phi}^2 \rangle$ is straightforward to calculate. We modedecompose K_{div} in terms of Legendre polynomials in the usual manner:

$$K_{\rm div}(X,X'|s) = \sum_{\ell=0}^{\infty} \frac{(2\ell+1)}{4\pi r r'} P_{\ell}(\cos\lambda) K_{\rm div|\ell}(x,x'|s); \qquad (4.14)$$

$$K_{div|\ell}(x, x'|s) = 2\pi r r' \int_{-1}^{1} d(\cos \lambda) P_{\ell}(\cos \lambda) K_{div}(X, X'|s). \quad (4.15)$$

The anomaly in $\langle \hat{\Phi}^2 \rangle$ is then just the coincidence limit of the difference between the subtraction terms in two dimensions and those mode-decomposed from four dimensions, integrated over s:

$$\Delta\langle\hat{\varphi}^2(x)\rangle = \int_0^\infty ds \left[\mathcal{K}_{\ell|\operatorname{div}}(x,x|s) - K_{\operatorname{div}|\ell}(x,x|s)\right] \,. \tag{4.16}$$

We encounter a difficulty, however, when we try to perform the mode decomposition. For a general space, σ and the $a_n^{\Box-\xi^{4}R}$ are known only for infinitesimal separations² of X and X', while evaluation of the mode-decomposition integral (4.15) requires knowing σ and the $a_n^{\Box-\xi^{4}R}$ for finite separations of X, X' on the two-sphere. We proceed by determining an approximate K_{div} for finite separation based on the following criteria:

- 1. Our approximate K_{div} must reduce to the known value in the flat-space limit.
- 2. Our approximate K_{div} must respect the periodicity of the two-spheres (i.e., it must be periodic in the angular separation λ with period 2π).

 $^{^{2}}$ In terms of momentum integrals, finite separations correspond to the low-frequency regime, where the renormalization terms are not fixed by the divergences in the theory.

The first criterion may be met in the following manner. We split the points X and X' in the angular direction only. Using the well-known short-distance expansions obtained in [80,81], σ and the $\Re_n^{\Box-\xi^{4}R}$ may be expanded in powers of the angular separation λ , which is equivalent to expanding in powers of the curvature. These expansions are then substituted into K_{div} and, assuming small curvatures, truncated at first order in the curvature for $\Delta \langle \hat{\varphi}_{\ell}^2 \rangle$ and at second order for ΔW_{ℓ} . To take account of the periodicity of the two-spheres, the resulting expansions in powers of λ^2 are then converted into expansions in $(1 - \cos \lambda)$. Defining $z = \cos \lambda$, we have

$$\lambda^{2} = 2(1-z) + \frac{1}{3}(1-z)^{2} + \frac{4}{45}(1-z)^{3} + \cdots .$$
 (4.17)

We then substitute (4.17) for each λ^2 , truncating at the lowest order in (1 - z) which will yield the correct flat-space limit. This replacement of λ^2 by a finite series in (1 - z) means that our expansions are only modified for large angular separations, where the renormalization terms are inherently ambiguous. Our choice simply corresponds to a natural extension of the flat-space heat kernel which respects the periodicity of the two-spheres for large angular separations. The details of this procedure are presented in Appendix B.2.

Following this procedure, one finds that to first order in the curvature

$$2\sigma = 2r^{2}(1-z) + \frac{r^{2}}{3}[1-r^{2}(\nabla\phi)^{2}](1-z)^{2}, \qquad (4.18)$$

$$\Re_0^{\Box-\xi^4 R} = 1 + \frac{1}{6} {}^4 R_{\theta\theta} (1-z), \qquad (4.19)$$

$$\Re_1^{\Box-\xi^{4}R} = \left(\frac{1}{6}-\xi\right)^{4}R.$$
 (4.20)

Inserting these expansions into (4.5) yields our approximation for the "divergent" part of the four-dimensional heat kernel,

$$K_{\text{div}}(X, X'|s) = \frac{1}{(4\pi s)^2} \exp\left\{-m^2 s - \frac{r^2}{2s}(1-z)\right\} \left[1 + s\left(\frac{1}{6} - \xi\right) {}^4R + \frac{1}{6} {}^4R_{\theta\theta}(1-z) - \frac{r^2}{12s} [1 - r^2(\nabla\phi)^2](1-z)^2\right]. \quad (4.21)$$

The mode decomposition (4.15) of this K_{div} then boils down to evaluating the integral

$$J_{\ell n} \equiv -p \int_{-1}^{1} dz \, P_{\ell}(z) \, \mathrm{e}^{p(1-z)} \, (1-z)^{n} \,, \qquad (4.22)$$

where $p \equiv -r^2/2s$ is a dimensionless parameter and n is an integer. The integrals for $n \neq 0$ can be obtained from the n = 0 result (3.72), (3.73) used in the flatspace case by differentiating with respect to p, yielding

$$J_{\ell n} = \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} \left[\frac{(-1)^{k}2^{n}}{(-2p)^{n+k}} \frac{(k+n)!}{k!} - (-1)^{\ell} e^{2p} \sum_{\alpha=0}^{n} \frac{2^{n}}{(-2p)^{\alpha+k}} \frac{(k+\alpha)!n!}{k!\alpha!(n-\alpha)!} \right]$$
(4.23)

The mode-decomposed heat kernel subtraction terms for a general fourdimensional spherically symmetric spacetime are then

$$K_{\text{div}|\ell}(x,x|s) = \frac{e^{-m^2s}}{4\pi s} \left[J_{\ell 0} + s\left(\frac{1}{6} - \xi\right)^4 R J_{\ell 0} + \frac{1}{6} A_{\theta\theta} J_{\ell 1} - \frac{r^2}{12s} [1 - r^2 (\nabla \phi)^2] J_{\ell 2} \right].$$
(4.24)

The first term in (4.24) is the mode decomposition for flat space, while the other terms carry the contributions due to the curvature. Meanwhile, the various parts of the $J_{\ell n}$ fulfill several roles. First, the $k \neq 0$ terms in (4.23) are associated with the centrifugal potential $\ell(\ell + 1)/r^2$ induced by the the mode decomposition. This potential is ignored in the renormalization in two dimensions, since only the first (potential-independent) term in the Schwinger-DeWitt expansion of the heat kernel contributes to the divergences of the two-dimensional Green function. Second, the terms in (4.23) proportional to $e^{2p} = e^{-r^2/s}$ enforce the Dirichlet boundary condition at r = 0 (if this point belongs to the four-dimensional manifold), which is required if the four-dimensional subtraction term is to be finite there [see (4.14)].

These results are to be compared with the subtraction term in two dimensions, which consists of the first term of (4.10):

$$\mathcal{K}_{\ell|\operatorname{div}}(x,x|s) = \frac{\mathrm{e}^{-m^2 s}}{4\pi s} \,. \tag{4.25}$$

In contrast to $K_{div|\ell}$, $\mathcal{K}_{\ell|div}$ is independent of both the position, the two-metric h_{ab} , and the mode number ℓ . It matches just the first term in the k = 0 contribution to the flat space part of $K_{div|\ell}$.

The anomaly in the heat kernel is found by subtracting (4.24) from (4.25). It is clear that part of this anomaly is due to the different number of terms in the Schwinger-DeWitt expansion used to renormalize the four- and two-dimensional theories. One might hope that the anomaly could be eliminated by keeping the second term in the Schwinger-DeWitt expansion in two dimensions, (4.10). It is easily verified, however, that doing so only cancels the first term in the k = 1contribution to the flat-space part of $K_{div|\ell}$, and the first term in the k = 0contribution to the curvature-dependent parts of $K_{div|\ell}$. Furthermore, there is no hope that renormalization, a purely local procedure, could cancel the terms responsible for the induced boundary condition at r = 0, which is essentially nonlocal.

The anomaly in $\langle \hat{\Phi}^2 \rangle$ is now obtained by integrating the difference of $\mathcal{K}_{\ell|div}$ and $K_{div|\ell}$ as in (4.16). We find

$$\Delta \langle \hat{\varphi}_{\ell}^{2} \rangle = \frac{1}{2\pi} \left[-I[m^{2}sJ_{\ell 0}] - \frac{1}{m^{2}} \left(\frac{1}{6} - \xi \right)^{4} R I[(m^{2}s)^{2}J_{\ell 0}] - \frac{1}{6} {}^{4} R_{\theta \theta} I[m^{2}sJ_{\ell 1}] + \frac{(mr)^{2}}{12} [1 - r^{2}(\nabla \phi)^{2}] I[J_{\ell 2}] \right], \qquad (4.26)$$

where

$$I[(m^{2}s)^{t}J_{\ell n}] \equiv \sum_{k=2-t-n}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} 2^{n} \left[\frac{1}{2} \frac{(-1)^{k}}{(mr)^{2n+2k}} \frac{(k+n)!}{k!} (k+t+n-2)! \right] - (-1)^{\ell} \sum_{k=0}^{\ell} \frac{(\ell+k)!}{k!(\ell-k)!} 2^{n} \left[\sum_{\alpha=0}^{n} \frac{(k+\alpha)!n!}{k!\alpha!(n-\alpha)!} \frac{K_{k+t+\alpha-1}(2mr)}{(mr)^{k-t+\alpha+1}} \right].$$

$$(4.27)$$

(The $I[(m^2s)^t J_{\ell n}]$ result from integrating terms of the form $(m^2s)^{t-2} J_{\ell n}$ over s.) Note that the anomaly in $\langle \hat{\Phi}^2 \rangle$ generally diverges at any point x^a such that

 $r(x^a) = 0$, while for asymptotically flat spaces it vanishes as $r \to \infty$. From the discussion of Section 3.3, this behaviour is to be expected.

4.1.3 Spherical-Reduction Anomaly in Schwarzschild Spacetime

As an example, let us consider a quantum field in Euclidean Schwarzschild space, with line element

$$ds^{2} = \left(1 - \frac{2M}{r}\right)dt^{2} + \left(1 - \frac{2M}{r}\right)^{-1}dr^{2} + r^{2}\left(d\theta^{2} + \sin^{2}\theta d\eta^{2}\right), \quad (4.28)$$

where M is the black-hole mass. For this geometry ${}^{4}\!R = 0$, ${}^{4}\!R_{\mu\nu} = 0$, and

$$[1 - r^2 (\nabla \phi)^2] = \frac{2M}{r} \,. \tag{4.29}$$

Figure 4.1 shows plots of $K_{div|\ell=0}(x, x|s)$ for fixed s and various values of M/\sqrt{s} . Note that large values of M cause the mode-decomposed subtraction terms to become negative.

The anomaly in $\langle \hat{\Phi}^2 \rangle$ for the $\ell = 0$ mode is

$$\Delta \langle \hat{\varphi}_{\ell=0}^2 \rangle = \frac{K_0(2mr)}{2\pi} + \frac{M}{3\pi r} \left[\frac{1}{(mr)^2} - \frac{2}{mr} K_1(2mr) - 2K_0(2mr) - mr K_1(2mr) \right] . \tag{4.30}$$

Plots of $\Delta \langle \hat{\varphi}_{\ell=0}^2 \rangle$ for various values of mM are shown in Figure 4.2. Note that in the $M \to 0$ limit we recover the exact anomaly in flat space, (3.79).

4.2 The Dimensional-Reduction Anomaly in the Effective Action

In the previous section we calculated the dimensional-reduction anomaly in $\langle \tilde{\Phi}^2 \rangle$ for a general four-dimensional spherically symmetric space. We now use the same



Figure 4.1: $K_{\text{div}|\ell=0}(x, x|s)$ in Schwarzschild space for fixed s and $M/\sqrt{s} = 0, 2, 4, 8$ from top to bottom. The common factor $e^{-m^2s}/(4\pi s)$ has been removed. The two-dimensional subtraction term $\mathcal{K}_{\ell|\text{div}}$ would be a horizontal line at 1 on this plot.

procedure to determine the anomaly in the effective action, denoted by ΔW_{ℓ} in (3.14). Functional differentiation of ΔW_{ℓ} with respect to the metric h_{ab} and the dilaton ϕ will then give the corresponding anomalies in the stress tensor and pressure in (3.15), (3.16).

The divergences in the four-dimensional effective action (4.4) come from the first three terms of the heat kernel expansion (4.5):

$$K_{\rm div}(X, X'|s) = \frac{1}{(4\pi s)^2} \exp\left\{-m^2 s - \frac{\sigma}{2s}\right\} \left[\Re_0^{\Box - \xi^4 R} + s \Re_1^{\Box - \xi^4 R} + s^2 \Re_2^{\Box - \xi^4 R}\right] .$$
(4.31)

As in the previous section, we split the points X, X' in the angular direction



Figure 4.2: $\Delta \langle \hat{\varphi}_{\ell=0}^2 \rangle$ in Schwarzschild space for mM = 0.1, 1, 10, from top to bottom.

only. Then one can write

$$2\sigma = 2r^2 \left[(1-z) + u(1-z)^2 + v(1-z)^3 + \cdots \right], \qquad (4.32)$$

$$\Re_{n}^{\Box-\xi^{4}R} = \Re_{n(0)}^{\Box-\xi^{4}R} + \Re_{n(1)}^{\Box-\xi^{4}R}(1-z) + \Re_{n(2)}^{\Box-\xi^{4}R}(1-z)^{2} + \cdots, \quad (4.33)$$

where $z = \cos \lambda$. From the calculations for the anomaly in $\langle \hat{\Phi}^2 \rangle$ we have seen that $u = \frac{1}{6} [1 - r^2 (\nabla \phi)^2]$, $\Re_{0(0)}^{\Box - \xi^{4}R} = 1$, $\Re_{0(1)}^{\Box - \xi^{4}R} = \frac{1}{6} {}^{4}R_{\theta\theta}$, and $\Re_{1(0)}^{\Box - \xi^{4}R} = (\frac{1}{6} - \xi) {}^{4}R$. The other $\Re_{n(k)}^{\Box - \xi^{4}R}$ and v are found in Appendix B.2. Inserting these expansions into (4.31) and truncating at second order in the curvature, we find

$$\begin{split} K_{\rm div}(X,X'|s) &= \frac{1}{(4\pi s)^2} \exp\left\{-m^2 s - \frac{r^2}{2s}(1-z)\right\} \left[1 + s \Re_{1(0)}^{\Box-\xi^4 R} + s^2 \Re_{2(0)}^{\Box-\xi^4 R} + \left(\Re_{0(1)}^{\Box-\xi^4 R} + s \Re_{1(1)}^{\Box-\xi^4 R}\right)(1-z) + \left(\Re_{0(2)}^{\Box-\xi^4 R} - \frac{r^2 u}{2s} - \frac{r^2 u}{2} \Re_{1(0)}^{\Box-\xi^4 R}\right)(1-z)^2 \end{split}$$

+
$$\left(-\frac{r^2 u}{2s} \Re_{0(1)}^{\Box-\xi^4 R} - \frac{r^2 v}{2s}\right) (1-z)^3 + \frac{r^4 u^2}{8s^2} (1-z)^4 \left].$$
 (4.34)

The decomposition of the heat kernel subtraction terms (4.34) is done in the same manner as in the previous section. Employing the definition (4.15) of the spherical decomposition and using the functions $J_{\ell n}$ of (4.22), (4.23), we obtain

$$\begin{split} K_{\text{div}|\ell}(x,x|s) &= \lim_{x' \to x} 2\pi r^2 \int_{-1}^{1} d(\cos \lambda) \, P_{\ell}(\cos \lambda) \, K_{\text{div}}(X,X'|s) \\ &= \frac{e^{-m^2 s}}{4\pi s} \left[\left(1 + s \Re_{1(0)}^{\Box - \xi^{4}R} + s^2 \Re_{2(0)}^{\Box - \xi^{4}R} \right) J_{\ell 0} \right. \\ &+ \left(\Re_{0(1)}^{\Box - \xi^{4}R} + s \Re_{1(1)}^{\Box - \xi^{4}R} \right) J_{\ell 1} \\ &+ \left(\Re_{0(2)}^{\Box - \xi^{4}R} - \frac{r^2 u}{2s} - \frac{r^2 u}{2} \Re_{1(0)}^{\Box - \xi^{4}R} \right) J_{\ell 2} \\ &+ \left(-\frac{r^2 v}{2s} - \frac{r^2 u}{2s} \Re_{0(1)}^{\Box - \xi^{4}R} \right) J_{\ell 3} + \frac{r^4 u^2}{8s^2} J_{\ell 4} \right] . (4.35) \end{split}$$

Meanwhile, the divergences in the effective action for the two-dimensional theory with the potential V_{ℓ} of (4.2) arise from the first two terms of (4.10):

$$\mathcal{K}_{\ell|\operatorname{div}}(x,x|s) = \frac{\mathrm{e}^{-m^2 s}}{4\pi s} \left[1 + s \left(\frac{1}{6} R - V_{\ell} \right) \right] \,. \tag{4.36}$$

The anomaly in the effective action is found by integrating the difference of (4.35), (4.36) over s as in (4.4):

$$\Delta \mathcal{W}_{\ell} = -\frac{1}{2} \int d^2 x \sqrt{h} \int_{0}^{\infty} \frac{ds}{s} \left[\mathcal{K}_{\ell \mid \text{div}}(x, x \mid s) - K_{\text{div}\mid\ell}(x, x \mid s) \right]$$

$$= \frac{m^2}{4\pi} \int d^2 x \sqrt{h} \left[I[J_{\ell 0}] + \frac{1}{m^2} \Re_{1(0)}^{\Box - \xi^{4}R} I[m^2 s J_{\ell 0}] + \frac{1}{m^4} \Re_{2(0)}^{\Box - \xi^{4}R} I[m^4 s^2 J_{\ell 0}] \right]$$

$$+ \Re_{0(1)}^{\Box - \xi^{4}R} I[J_{\ell 1}] + \frac{1}{m^2} \Re_{1(1)}^{\Box - \xi^{4}R} I[m^2 s J_{\ell 1}] + (\Re_{0(2)}^{\Box - \xi^{4}R} - \frac{r^2 u}{2} \Re_{1(0)}^{\Box - \xi^{4}R}) I[J_{\ell 2}]$$

$$- \frac{(mr)^2 u}{2} I[\frac{1}{m^2 s} J_{\ell 2}] - \frac{(mr)^2}{2} (u \Re_{0(1)}^{\Box - \xi^{4}R} + v) I[\frac{1}{m^2 s} J_{\ell 3}]$$

$$+ \frac{(mr)^4 u^2}{8} I[\frac{1}{m^4 s^2} J_{\ell 4}] \right]. \qquad (4.37)$$

The $I[(m^2s)^t J_{\ell n}]$ are given by (4.27). Using (4.37) and the values of u, v, and the $\Re_{n(k)}^{\Box-\xi^{4}R}$ given in Appendix B.2, one can compute the anomalous contribution to the stress tensor. This is done in the next section.

4.3 The Dimensional-Reduction Anomaly in the Stress Tensor

The stress-tensor induced by a given effective action W in a space with metric h_{ab} may be calculated using

$$\langle \hat{\mathcal{T}}_{ab} \rangle = \frac{2}{\sqrt{h}} \frac{\delta \mathcal{W}}{\delta h^{ab}} \,. \tag{4.38}$$

In the previous section we found the anomaly in the effective action for the decomposition of a general four-dimensional spherically symmetric space, (4.37). We now calculate the anomalous contribution to the stress tensor due to this action. We do this by substituting the values of u, v, and the $\Re_{n(k)}^{\Box-\xi}{}^{4R}$ given in Appendix B.2 into (4.37), using integration by parts to cast the action into a form suitable for performing the variations, and then calculating $\Delta \langle \hat{T}_{ab}^{(\ell)} \rangle$ using (4.38). The anomaly in the pressure, $\Delta \langle \hat{\mathcal{P}}^{(\ell)} \rangle$, is most easily computed once $\Delta \langle \hat{T}_{ab}^{(\ell)} \rangle$ is known by using the dimensionally reduced conservation equation (2.97), which implies

$$\langle \hat{\mathcal{P}}^{(\ell)} \rangle = -\frac{1}{2(\nabla \phi)^2} \phi^a \nabla^b \langle \hat{\mathcal{T}}^{(\ell)}_{ab} \rangle .$$
(4.39)

This procedure is straightforward but tedious, so we skip most of the details.

There are only a limited number of invariants dependent on h^{ab} (such as $\Delta_h \phi$, $(\nabla \phi)^2$, and R[h]) which enter ΔW_ℓ . For this reason it will be convenient to define the following functions which result from the variation of these objects. Assuming S and N are scalar functions, and that $\delta N = 0$ (but not δS) under a variation δh^{ab} , one can show that

$$\int d^2x \sqrt{h} \,\delta([\nabla \phi]^2) S = \int d^2x \sqrt{h} \,\delta h^{ab} F^0_{ab}(S) \,, \tag{4.40}$$

$$\int d^2x \sqrt{h} \,\delta([\nabla \phi]^2) S = \int d^2x \sqrt{h} \,\delta h^{ab} F^0_{ab}(S) \,, \qquad (4.41)$$

$$\int d^2x \sqrt{h} \,\delta(\Delta_h N) S = \int d^2x \sqrt{h} \,\delta h^{ab} F^1_{ab}(N,S) \,, \qquad (4.42)$$

$$\int d^2x \sqrt{h} \,\delta(R)S = \int d^2x \sqrt{h} \,\delta h^{ab} \,F^2_{ab}(R,S)\,,$$

where

$$F^0_{ab}(S) = \phi_a \phi_b S , \qquad (4.43)$$

$$F_{ab}^{1}(N,S) = \frac{1}{2} \left[-N_{a}\partial_{b} - N_{b}\partial_{a} + h_{ab}\Delta_{h}N + h_{ab}\nabla N \cdot \nabla \right] S, \quad (4.44)$$

$$F_{ab}^{2}(R,S) = \frac{1}{2} \left[-\nabla_{a} \nabla_{b} + h_{ab} \Delta_{h} + \frac{1}{2} h_{ab} R \right] S.$$
(4.45)

The other variations which we shall require can be written in terms of these functions. For example,

$$\int d^2x \sqrt{h} \,\delta(\,^4R)S = \int d^2x \sqrt{h} \,\delta h^{ab} \left[F^2_{ab}(R,S) + 4F^1_{ab}(\phi,S) - 6F^0_{ab}(S)\right] \,, \quad (4.46)$$

$$\int d^{2}x \sqrt{h} \,\delta(\Delta_{h}{}^{4}R)S = \int d^{2}x \sqrt{h} \,\delta h^{ab} \left[F_{ab}^{2}(R,\Delta_{h}S) + F_{ab}^{1}({}^{4}R,S) + 4F_{ab}^{1}(\phi,\Delta_{h}S) - 6F_{ab}^{0}(\Delta_{h}S)\right].$$
(4.47)

We also have

$$\int d^2x \delta(\sqrt{h}) S = \int d^2x \sqrt{h} \,\delta h^{ab} \left[-\frac{1}{2} h_{ab} S \right] \,. \tag{4.48}$$

For convenience, we divide our effective action (4.37) up into 9 separate parts,

$$\Delta \mathcal{W}_{\ell} = \frac{m^2}{4\pi} \sum_{i=1}^{1X} W_i \,. \tag{4.49}$$

These parts and their associated stress tensors for the case $V = \xi^4 R$ are listed below. The second form shown for each W_i is obtained using integration by parts and is used for the calculation of the $\langle T_{ab} \rangle_i$. Note that each W_i is constructed using a different $I[(m^2s)^t J_{\ell n}]$. For convenience, we write the $I[(m^2s)^t J_{\ell n}]$ as simply I on all but the first line of each result.

$$W_{\rm t} = \int d^2x \sqrt{h} I[J_{\ell 0}], \qquad (4.50)$$

$$\langle T_{ab} \rangle_{\mathfrak{l}} = -h_{ab}I, \qquad (4.51)$$

$$W_{II} = \int d^2x \sqrt{h} \frac{1}{m^2} \Re_{1(0)}^{\Box - \xi^{4R}} I[m^2 s J_{\ell 0}]$$

= $\frac{\left(\frac{1}{6} - \xi\right)}{m^2} \int d^2x \sqrt{h} {}^{4R} I,$ (4.52)

$$\langle T_{ab} \rangle_{II} = \frac{2\left(\frac{1}{6} - \xi\right)}{m^2} \left[-\frac{1}{2} h_{ab} \,^4 R \, I - 6 F^0_{ab}(I) + 4 F^1_{ab}(\phi, I) + F^2_{ab}(R, I) \right] \,, \, (4.53)$$

$$W_{III} = \int d^{2}x \sqrt{h} \frac{1}{m^{4}} \Re_{2(0)}^{\Box - \xi^{4}R} I[m^{4}s^{2}J_{\ell 0}]$$

$$= \frac{\left(\frac{1}{6} - \xi\right)^{2}}{2m^{4}} \int d^{2}x \sqrt{h} ({}^{4}R)^{2}I$$

$$+ \frac{\left(\frac{1}{6} - \xi\right)}{6m^{4}} \int d^{2}x \sqrt{h} \left[\Delta_{h}{}^{4}R I + 2{}^{4}R \Delta_{h}\phi I + 2{}^{4}R \nabla \phi \cdot \nabla I\right]$$

$$+ \frac{1}{180m^{4}} \int d^{2}x \sqrt{h} \left[\left(\Delta_{h}{}^{4}R + \frac{1}{2}R^{2} + 10(\Delta_{h}\phi)^{2} + \frac{2}{r^{4}}\right) I$$

$$+ \left(2R + 12\Delta_{h}\phi - 8(\nabla\phi)^{2} + \frac{4}{r^{2}}\right) \nabla\phi \cdot \nabla I$$

$$+ 2(\nabla\phi)^{2}\Delta_{h}I\right], \qquad (4.54)$$

$$\begin{split} \langle T_{ab} \rangle_{III} &= \frac{\left(\frac{1}{6} - \xi\right)^2}{m^4} \left[-\frac{1}{2} h_{ab} ({}^4R)^2 I - 12 F^0_{ab} ({}^4RI) + 8 F^1_{ab} (\phi, {}^4RI) \right. \\ &+ 2 F^2_{ab} (R, {}^4RI) \right] \\ &+ \frac{\left(\frac{1}{6} - \xi\right)}{3m^4} \left[-\frac{1}{2} h_{ab} \left(\Delta_h {}^4RI + 2 {}^4R \Delta_h \phi I + 2 {}^4R \nabla \phi \cdot \nabla I \right) \right. \\ &+ 2 {}^4R \phi_{(a} I_{b)} - 6 F^0_{ab} (2 \Delta_h \phi I + 2 \nabla \phi \cdot \nabla I + \Delta_h I) \right. \\ &+ F^1_{ab} (\phi, 2 {}^4RI + 8 \Delta_h \phi I + 8 \nabla \phi \cdot \nabla I + 4 \Delta_h I) \\ &+ F^1_{ab} ({}^4R, I) + F^2_{ab} (R, 2 \Delta_h \phi I + 2 \nabla \phi \cdot \nabla I + \Delta_h I) \right] \\ &+ \frac{1}{90m^4} \left[-\frac{1}{2} h_{ab} \left((\Delta_h {}^4R + \frac{1}{2}R^2 + 10(\Delta_h \phi)^2 + \frac{2}{r^4}) I \right) \right] \end{split}$$

$$+ (2R + 12\Delta_{h}\phi - 8(\nabla\phi)^{2} + \frac{4}{r^{2}})\nabla\phi\cdot\nabla I + 2(\nabla\phi)^{2}\Delta_{h}I)$$

$$+ \left(2R + 12\Delta_{h}\phi - 8(\nabla\phi)^{2} + \frac{4}{r^{2}}\right)\phi_{(a}I_{b)}$$

$$+ F^{0}_{ab}(-8\nabla\phi\cdot\nabla I - 4\Delta_{h}I)$$

$$+ F^{1}_{ab}(\phi, 20\Delta_{h}\phi I + 12\nabla\phi\cdot\nabla I + 4\Delta_{h}I)$$

$$+ F^{1}_{ab}(I, 2(\nabla\phi)^{2}) + F^{1}_{ab}({}^{4}R, I)$$

$$+ F^{2}_{ab}(R, RI + 2\nabla\phi\cdot\nabla I + \Delta_{h}I)], \qquad (4.55)$$

$$W_{\rm IV} = \int d^2x \sqrt{h} \,\Re_{0(1)}^{\Box - \xi^{4R}} I[J_{\ell 1}] = \int d^2x \sqrt{h} \,\frac{1}{6} (1 + r^2 B) \,I \,, \qquad (4.56)$$

$$\langle T_{ab} \rangle_{\rm IV} = \frac{1}{3} \left[-\frac{1}{2} h_{ab} \left(1 + r^2 B \right) I - 2F^0_{ab} (r^2 I) + F^1_{ab} (\phi, r^2 I) \right],$$
 (4.57)

$$W_{\rm v} = \int d^2x \sqrt{h} \frac{1}{m^2} \Re^{\Box - \xi^{4R}}_{1(1)} I[m^2 s J_{\ell 1}]$$

= $\frac{\left(\frac{1}{6} - \xi\right)}{6m^2} \int d^2x \sqrt{h} \left[{}^{4R}(I - r^2 \nabla \phi \cdot \nabla I) \right]$
+ $\frac{1}{180m^2} \int d^2x \sqrt{h} \left[\frac{2}{r^2} I + (-R - 2\Delta_h \phi + 4(\nabla \phi)^2) r^2 \nabla \phi \cdot \nabla I$
+ $(3\Delta_h \phi - 2\nabla \phi^2) r^2 \Delta_h I \right],$ (4.58)

$$\langle T_{ab} \rangle_{\vee} = \frac{\left(\frac{1}{6} - \xi\right)}{3m^2} \left[-\frac{1}{2} h_{ab} \, {}^4R(I - r^2 \nabla \phi \cdot \nabla I) - {}^4Rr^2 \phi_{(a}I_{b)} \right. \\ \left. - 6F^0_{ab}(I - r^2 \nabla \phi \cdot \nabla I) + 4F^1_{ab}(\phi, I - r^2 \nabla \phi \cdot \nabla I) \right. \\ \left. + F^2_{ab}(R, I - r^2 \nabla \phi \cdot \nabla I) \right]$$

$$+\frac{1}{90m^{2}}\left[-\frac{1}{2}h_{ab}\left(\frac{2}{r^{2}}I+(-R-2\Delta_{h}\phi+4(\nabla\phi)^{2})r^{2}\nabla\phi\cdot\nabla I\right.\right.\\\left.+\left(3\Delta_{h}\phi-2\nabla\phi^{2})r^{2}\Delta_{h}I\right)\right.\\\left.+\left(-R-2\Delta_{h}\phi+4(\nabla\phi)^{2}\right)r^{2}\phi_{(a}I_{b)}\right.\\\left.+F_{ab}^{0}(4r^{2}\nabla\phi\cdot\nabla I-2r^{2}\Delta_{h}I)\right.\\\left.+F_{ab}^{1}(\phi,-2r^{2}\nabla\phi\cdot\nabla I+3r^{2}\Delta_{h}I)\right.\\\left.+F_{ab}^{1}(I,\left(3\Delta_{h}\phi-2\nabla\phi^{2}\right)r^{2})\right.\\\left.+F_{ab}^{2}(R,-r^{2}\nabla\phi\cdot\nabla I)\right],\qquad(4.59)$$

$$W_{\rm vi} = \int d^2x \sqrt{h} \left(\Re_{0(2)}^{\Box-\xi^{4}R} - \frac{r^{2}u}{2} \Re_{1(0)}^{\Box-\xi^{4}R} \right) I[J_{\ell 2}] \\ = -\frac{\left(\frac{1}{6} - \xi\right)}{12} \int d^2x \sqrt{h} \left[(1 - r^{2}(\nabla\phi)^{2})^{4}R r^{2}I \right] \\ + \frac{1}{360} \int d^2x \sqrt{h} \left[19I + (20\Delta_{h}\phi - 40\nabla\phi^{2})r^{2}I \\ + (R\nabla\phi^{2} - 3(\Delta_{h}\phi)^{2} + 10\Delta_{h}\phi\nabla\phi^{2} - 8\nabla\phi^{4})r^{4}I \\ + (-8\Delta_{h}\phi + 6\nabla\phi^{2})r^{4}\nabla\phi \cdot \nabla I + 2\nabla\phi^{2}r^{4}\Delta_{h}I \right], \quad (4.60)$$

$$\langle T_{ab} \rangle_{\rm VI} = -\frac{\left(\frac{1}{6} - \xi\right)}{6} \left[-\frac{1}{2} h_{ab} \left(1 - r^2 \nabla \phi^2\right)^4 R r^2 I \right. \\ \left. + F^0_{ab} \left(-^4 R r^4 I - 6 (1 - r^2 \nabla \phi^2) r^2 I \right) \right. \\ \left. + 4 F^1_{ab} (\phi, (1 - r^2 \nabla \phi^2) r^2 I) + F^2_{ab} (R, (1 - r^2 \nabla \phi^2) r^2 I) \right] \right. \\ \left. + \frac{1}{180} \left[-\frac{1}{2} h_{ab} \left(19I + (20\Delta_h \phi - 40 \nabla \phi^2) r^2 I \right. \\ \left. + (R \nabla \phi^2 - 3(\Delta_h \phi)^2 + 10\Delta_h \phi \nabla \phi^2 - 8 \nabla \phi^4) r^4 I \right. \\ \left. + (-8\Delta_h \phi + 6 \nabla \phi^2) r^4 \nabla \phi \cdot \nabla I + 2 \nabla \phi^2 r^4 \Delta_h I \right) \right. \\ \left. + \left(-8\Delta_h \phi + 6 \nabla \phi^2) r^4 \phi_{(a} I_{b)} \right. \\ \left. + F^0_{ab} (-40 r^2 I + [R + 10\Delta_h \phi - 16(\nabla \phi)^2] r^4 I + 6 r^4 \nabla \phi \cdot \nabla I \right.$$

$$+ 2r^{4}\Delta_{h}I) + F_{ab}^{1}(\phi, 20r^{2}I + [-6\Delta_{h}\phi + 10(\nabla\phi)^{2}]r^{4}I - 8r^{4}\nabla\phi\cdot\nabla I) + F_{ab}^{1}(I, 2(\nabla\phi)^{2}r^{4}) + F_{ab}^{2}(R, (\nabla\phi)^{2}r^{4}I)], \qquad (4.61)$$

$$W_{\text{VII}} = -\int d^2x \sqrt{h} \left[\frac{(mr)^2 u}{2} I[\frac{1}{m^2 s} J_{\ell 2}] \right]$$

= $-\frac{m^2}{12} \int d^2x \sqrt{h} \left[(1 - r^2 \nabla \phi^2) r^2 I \right],$ (4.62)

$$\langle T_{ab} \rangle_{\rm VII} = -\frac{m^2}{6} \left[-\frac{1}{2} h_{ab} \left(1 - r^2 \nabla \phi^2 \right) - r^2 \phi_a \phi_b \right] r^2 I ,$$
 (4.63)

$$W_{\text{VIII}} = -\int d^2x \sqrt{h} \left[\frac{(mr)^2}{2} (u \Re_{0(1)}^{\Box - \xi \, 4R} + v) I[\frac{1}{m^2 s} J_{\ell 3}] \right]$$

= $-\frac{m^2}{360} \int d^2x \sqrt{h} \left[13r^2 I + (5\Delta_h \phi - 25\nabla \phi^2) r^4 I + (-2\Delta_h \phi \nabla \phi^2) r^6 I + 3r^6 \nabla \phi^2 \nabla \phi \cdot \nabla I \right], \qquad (4.64)$

$$\langle T_{ab} \rangle_{\text{VIII}} = \frac{m^2}{360} \left[h_{ab} \left(13r^2 I + (5\Delta_h \phi - 25\nabla \phi^2) r^4 I + (-2\Delta_h \phi \nabla \phi^2) r^6 I + 3r^6 \nabla \phi^2 \nabla \phi \cdot \nabla I \right) - 6\nabla \phi^2 r^6 \phi_{(a} I_{b)} + F_{ab}^0 (50r^4 I + 4\Delta_h \phi r^6 I - 6r^6 \nabla \phi \cdot \nabla I) + F_{ab}^1 (\phi, -10r^4 I + 4\nabla \phi^2 r^6 I) \right] ,$$

$$(4.65)$$

$$W_{IX} = \int d^2x \sqrt{h} \frac{(mr)^4 u^2}{8} I[\frac{1}{m^4 s^2} J_{\ell 4}]$$

= $\frac{m^4}{288} \int d^2x \sqrt{h} \left[(1 - r^2 \nabla \phi^2)^2 r^4 I \right],$ (4.66)

$$\langle T_{ab} \rangle_{\rm IX} = -\frac{m^4}{288} \left[h_{ab} \left(1 - r^2 \nabla \phi^2 \right) + 4r^2 \phi_a \phi_b \right] \left(1 - r^2 \nabla \phi^2 \right) r^4 I \,. \tag{4.67}$$

The total anomalous stress tensor in the x^a sector is

$$\Delta \langle \hat{\mathcal{T}}_{ab}^{(\ell)} \rangle = \frac{m^2}{4\pi} \sum_{i=1}^{1X} \langle T_{ab} \rangle_i \,. \tag{4.68}$$

The anomalous pressure is most easily calculated for a given geometry using (4.39). In the next section, both $\Delta \langle \hat{\mathcal{T}}_{ab}^{(\ell)} \rangle$ and $\Delta \langle \hat{\mathcal{P}}^{(\ell)} \rangle$ are computed explicitly for the $\ell = 0$ mode in the Schwarzschild geometry.

4.4 Anomalous Stress Tensor in Schwarzschild Space

In this section we calculate the anomalous contribution to the stress tensor for the $\ell = 0$ mode in the Schwarzschild geometry. This example is of particular significance to attempts to calculate the stress tensor of four-dimensional black holes using dimensionally reduced dilaton-gravity theories [32–39], discussed briefly in Section 2.2.7.

Using (4.68) and expressions (4.50)–(4.67), we can calculate the dimensionalreduction anomaly in a spacetime with given metric. In particular, for the $\ell = 0$ mode in the Euclidean Schwarzschild geometry (4.28), we find

$$\begin{aligned} \Delta \langle \hat{\mathcal{T}}_{t}^{(0)t} \rangle &= \frac{1}{180\pi m^{2}r^{6}} \left[-28rM + 54M^{2} + \left(-15m^{2}r^{4} - 30m^{4}r^{6} + 56m^{2}r^{3}M \right. \right. \\ &+ 82m^{4}r^{5}M - 108m^{2}r^{2}M^{2} - 96m^{4}r^{4}M^{2} \right) K_{0}(2mr) \\ &+ \left(56mr^{2}M + 110m^{3}r^{4}M + 20m^{5}r^{6}M - 108mrM^{2} \right) \end{aligned}$$

$$\begin{split} &-162m^3r^3M^2 - 30m^5r^5M^2)\,K_1(2mr)]\\ &+ \frac{(\xi - \frac{1}{\delta})}{3\pi m^2r^6}\left[20rM - 45M^2 + (3m^2r^4 + 6m^4r^6 - 49m^2r^3M \\ &- 26m^4r^5M + 90m^2r^2M^2 + 30m^4r^4M^2)\,K_0(2mr) \\ &+ (9m^3r^5 - 40mr^2M - 55m^3r^4M - 4m^5r^6M \\ &+ 90mrM^2 + 75m^3r^3M^2 + 8m^5r^5M^2)\,K_1(2mr)\right], (4.69) \\ \Delta \langle \hat{T}^{(0)r} \rangle &= \frac{1}{180\pi m^2r^6}\left[20rM - 42M^2 + (-15m^2r^4 - 40m^2r^3M \\ &- 38m^4r^5M + 84m^2r^2M^2 + 24m^4r^4M^2)\,K_0(2mr) \\ &+ (30m^3r^5 - 40mr^2M - 58m^3r^4M + 84mrM^2 \\ &+ 54m^3r^3M^2 + 10m^5r^5M^2)\,K_1(2mr)\right] \\ &+ \frac{(\xi - \frac{1}{\delta})}{3\pi m^2r^6}\left[-10rM + 15M^2 + (-6m^2r^4 + 29m^2r^3M \\ &+ 4m^4r^5M - 30m^2r^2M^2 - 6m^4r^4M^2)\,K_0(2mr) \\ &+ (-6m^3r^5 + 20mr^2M + 23m^3r^4M - 30mrM^2 \\ &- 21m^3r^3M^2)\,K_1(2mr)\right], \qquad (4.70) \\ \Delta \langle \hat{\mathcal{P}}^{(0)} \rangle &= \frac{1}{180\pi m^2r^6}\left[-50rM + 150M^2 + (15m^2r^4 - 30m^4r^6 \\ &+ 100m^2r^3M + 92m^4r^5M - 300m^2r^2M^2 \\ &- 108m^4r^4M^2 - 10m^6r^6M^2)\,K_0(2mr) \\ &+ (-15m^3r^5 + 100mr^2M + 142m^3r^4M + 38m^5r^5M \\ &- 300mrM^2 - 246m^3r^3M^2 - 44m^5r^5M^2)\,K_1(2mr)\right] \\ + \frac{(\xi - \frac{1}{\delta})}{3\pi m^2r^6}\left[25rM - 60M^2 + (6m^2r^4 + 6m^4r^6 - 68m^2r^3M \\ &- 28m^4r^5M + 120m^2r^2M^2 + 36m^4r^4M^2)\,K_0(2mr) \\ &+ (12m^3r^5 - 50mr^2M - 71m^3r^4M - 4m^5r^6M \\ &+ 120mrM^2 + 96m^3r^3M^2 \\ &+ 8m^5r^5M^2)\,K_1(2mr)\right]. \qquad (4.71) \end{split}$$

While these stress tensors were calculated for Euclidean Schwarzschild space,

they are also valid for Lorentzian signature.³ It is easily verified that the anomalous stress tensor is finite on both the past and future event horizons. In the $M \rightarrow 0$ limit we recover the exact results obtained earlier for the spherical decomposition of flat space, (3.82)-(3.84). Other limits of interest are $m \rightarrow 0$ and $mr \gg 1$. For massless fields we find

$$\begin{aligned} \Delta \langle \hat{\mathcal{T}}_{t}^{(0)t} \rangle &= \frac{1}{60\pi r^{2}} \left[5 \left(\gamma + \ln \mu r \right) + 9 \frac{M}{r} - 9 \frac{M^{2}}{r^{2}} \right] \\ &+ \frac{\left(\xi - \frac{1}{6} \right)}{4\pi r^{2}} \left[\left(-4 + 12 \frac{M}{r} \right) \left(\gamma + \ln \mu r \right) + 6 - 10 \frac{M}{r} - 10 \frac{M^{2}}{r^{2}} \right], \end{aligned}$$

$$(4.72)$$

$$\begin{aligned} \Delta \langle \hat{\mathcal{T}}^{(0)r}_{r} \rangle &= \frac{1}{60\pi r^{2}} \left[5 \left(\gamma + \ln \mu r \right) + 5 - 3\frac{M}{r} - 5\frac{M^{2}}{r^{2}} \right] \\ &+ \frac{\left(\xi - \frac{1}{6} \right)}{4\pi r^{2}} \left[\left(8 - 12\frac{M}{r} \right) \left(\gamma + \ln \mu r \right) - 4 + 2\frac{M}{r} + 6\frac{M^{2}}{r^{2}} \right], \end{aligned}$$

$$(4.73)$$

$$\begin{split} \Delta \langle \hat{\mathcal{P}}^{(0)} \rangle &= \frac{1}{120\pi r^2} \left[-10 \left(\gamma + \ln \mu r \right) - 5 + 14 \frac{M}{r} + 18 \frac{M^2}{r^2} \right] \\ &+ \frac{\left(\xi - \frac{1}{6} \right)}{8\pi r^2} \left[\left(-16 + 48 \frac{M}{r} \right) \left(\gamma + \ln \mu r \right) + 16 - 28 \frac{M}{r} - 32 \frac{M^2}{r^2} \right], \end{split}$$

$$(4.74)$$

where the μ appearing in the logarithms is an arbitrary parameter with units of mass. In the large mr limit, the Bessel functions in (4.69)–(4.71) are exponentially damped and may be dropped, leaving

$$\begin{split} \Delta \langle \hat{\mathcal{T}}_{t}^{(0)t} \rangle &= \frac{1}{180\pi m^{2}r^{4}} \left[-28\frac{M}{r} + 54\frac{M^{2}}{r^{2}} + 60\left(\xi - \frac{1}{6}\right)\left(20\frac{M}{r} - 45\frac{M^{2}}{r^{2}}\right) \right] , \\ \Delta \langle \hat{\mathcal{T}}_{r}^{(0)r} \rangle &= \frac{1}{180\pi m^{2}r^{4}} \left[20\frac{M}{r} - 42\frac{M^{2}}{r^{2}} + 60\left(\xi - \frac{1}{6}\right)\left(-10\frac{M}{r} + 15\frac{M^{2}}{r^{2}}\right) \right] , \end{split}$$
(4.75)
(4.76)

³One converts from Euclidean to Lorentzian signature by changing the sign of $\langle \hat{T}_{tt} \rangle$ and $\langle \hat{T}^{tt} \rangle$, but leaving $\langle \hat{T}_{t}^{t} \rangle$ and the other diagonal components of the stress tensor unchanged.

$$\Delta \langle \hat{\mathcal{P}}^{(0)} \rangle = \frac{1}{360\pi m^2 r^4} \left[-100 \frac{M}{r} + 300 \frac{M^2}{r^2} + 60 \left(\xi - \frac{1}{6} \right) \left(50 \frac{M}{r} - 120 \frac{M^2}{r^2} \right) \right]. \tag{4.77}$$

Plots of (4.75)-(4.77) versus the full expression (4.69)-(4.71) indicate that the asymptotic form is valid for all $r \ge 2M$ for approximately mM > 3.

To appreciate the significance of the anomaly, it will be convenient to analyze the massless and massive cases separately.

4.4.1 Massless Field

The calculation of the vacuum expectation value of the stress tensor for a massless scalar field in the Schwarzschild geometry has long been of interest [5–15]. In particular, in recent years numerous attempts have been made to obtain the stress tensor due to the $\ell = 0$ mode using dimensionally reduced models [32–39].

For an analysis of the dimensional reduction, let us consider the stress tensor for a field in the Hartle-Hawking state $|H\rangle$ [82]. This is the state for a black-hole spacetime which is regular on the past and future event horizons and which reduces at infinity to thermal radiation at temperature $T_H = (8\pi M)^{-1}$. It represents a black hole in (unstable) thermal equilibrium with radiation; i.e., it most closely corresponds to the thermal states considered in the previous chapter. As well, it is the state obtained by continuation from the regular Euclidean Schwarzschild geometry [82].

Anderson [11] and Anderson, Hiscock, and Samuel [12] (henceforth AHS) have developed an approximation scheme for $\langle \hat{\Phi}^2 \rangle$ and $\langle \hat{T}_{\mu}^{\nu} \rangle$ for scalar fields in four-dimensional static, spherically symmetric spacetimes. Their approach is to write the Green function as a mode sum, as discussed in Section 2.1.4, where the time dependence is of the form $e^{i\omega t}$ and the angular dependence is in terms of spherical harmonics $Y_{\ell m}(\theta, \eta)$. A WKB (high-frequency) approximation is then used to obtain the radial modes. The mode sum is evaluated for large ω and ℓ , and renormalized using point splitting [80,81]. The resulting stress tensor has been verified numerically for massless scalar fields with arbitrary coupling ξ in the Schwarzschild geometry [12]. For a field at temperature $T = \beta^{-1}$ they find

$$\langle \hat{T}_{t}^{t} \rangle_{\text{AHS}} = \frac{1}{2880\pi^{2}r^{4}f^{2}} \left[-96\pi^{4}\frac{r^{4}}{\beta^{4}} + 60\frac{M^{2}}{r^{2}} - 216\frac{M^{3}}{r^{3}} + 198\frac{M^{4}}{r^{4}} \right] + \frac{\left(\xi - \frac{1}{6}\right)}{16\pi^{2}r^{4}f^{2}} \left[8\pi^{2}\frac{M^{2}}{\beta^{2}} - 4\frac{M^{2}}{r^{2}} + 12\frac{M^{3}}{r^{3}} - 10\frac{M^{4}}{r^{4}} \right],$$
 (4.78)

$$\langle \hat{T}_{\theta}^{\theta} \rangle_{\text{AHS}} = \frac{1}{2880\pi^2 r^4 f^2} \left[32\pi^4 \frac{r^4}{\beta^4} - 8\frac{M^2}{r^2} + 24\frac{M^3}{r^3} - 18\frac{M^4}{r^4} \right] + \frac{\left(\xi - \frac{1}{6}\right)}{48\pi^2 r^4 f^2} \left[8\pi^2 \frac{Mr}{\beta^2} - 16\frac{M^2}{r^2} + 54\frac{M^3}{r^3} - 48\frac{M^4}{r^4} \right],$$
 (4.80)

where $f \equiv (1 - 2M/r)$. (In a spherical geometry the effective pressure *P* is equal to $\langle \hat{T}_{\theta}^{\theta} \rangle$.) The Hartle-Hawking state corresponds to the choice $\beta = 8\pi M$. Note that (4.78)-(4.80) is the stress tensor due to all modes ($\ell = 0, 1, 2, ...$) of the field.

Recently, Balbinot, Fabbri, and Nicolini (henceforth BFN) have used the AHS technique to calculate the stress tensor for the $\ell = 0$ mode of the dimensionally reduced massless scalar field with $\xi = 0$ in the Schwarzschild geometry [83]. For the Hartle-Hawking state they find

$$\langle \hat{\mathcal{T}}_t^t \rangle_{\rm BFN} = -\frac{1}{384\pi M^2} \left[1 + 2\frac{M}{r} + 4\frac{M^2}{r^2} + 40\frac{M^3}{r^3} \right],$$
 (4.81)

$$\langle \hat{\mathcal{T}}_{r}^{r} \rangle_{\rm BFN} = \frac{1}{384\pi M^2} \left[1 + 2\frac{M}{r} + 4\frac{M^2}{r^2} - 88\frac{M^3}{r^3} \right],$$
 (4.82)

$$\langle \hat{\mathcal{P}} \rangle_{\text{BFN}} = \frac{1}{384\pi M^2} \left[144 \frac{M^3}{r^3} \right].$$
 (4.83)

The reader is reminded of our convention of using Latin letters to denote four-

dimensional quantities (e.g., $\langle \hat{T}_t^t \rangle$) and calligraphic letters for two-dimensional quantities (e.g., $\langle \hat{T}_t^t \rangle$).

In Sections 3.2 and 3.3 we saw that in flat space the s-mode makes the dominant contribution to the stress tensor near the origin. However, the stress tensor calculated from the dimensionally reduced theory for the $\ell = 0$ mode is larger than the expected result in four dimensions for approximately $r < \beta$, and in this region the contribution of the dimensional-reduction anomaly must be taken into account to obtain the correct stress tensor for a given mode. It is interesting to analyze the situation for the Schwarzschild geometry. Does the $\ell = 0$ stress tensor $\langle \hat{T}_a{}^b \rangle_{\rm BFN}$ serve as an approximation for the full stress tensor $\langle \hat{T}_a{}^b \rangle_{\rm AHS}$ in the immediate vicinity of the black hole, and is the dimensional-reduction anomaly significant?

To test these ideas, we multiply the two-dimensional stress tensor of BFN by $(4\pi r^2)^{-1}$ as in (3.15) and compare to the stress tensor of AHS in four dimensions. The results are displayed in Figures 4.3 and 4.4. Near r = 2M, the two-dimensional stress tensor of BFN (from the $\ell = 0$ mode alone) is approximately an order of magnitude larger than and of opposite sign to the expected four-dimensional stress tensor of AHS (due to all modes). Clearly, the $\ell = 0$ contribution alone cannot serve as a useful approximation for the full stress tensor; furthermore, it seems unlikely that $\ell > 0$ modes should cancel this $\ell = 0$ result to a sufficiently high degree to restore agreement with the four-dimensional stress tensor. This indicates a dismal failure of the dimensional reduction.

The resolution of this dilemma may be that we have not taken into account the dimensional-reduction anomaly. The anomalous stress tensor is shown in Figures 4.5 and 4.6; it is of the same order as but of opposite sign to the predictions of BFN near the horizon. If the anomaly is added to the two-dimensional stress tensor and the total compared to $\langle \hat{T}_a{}^b \rangle_{AHS}$, we see that the agreement between



Figure 4.3: Comparison of the two-dimensional stress tensor of BFN for the $\ell = 0$ mode to the four-dimensional result of AHS for a massless, minimally coupled scalar field. The upper curve is $\langle \hat{T}_t^t \rangle_{AHS}$; the lower curve is $(4\pi r^2)^{-1} \langle \hat{T}_t^t \rangle_{BFN}$. This plot uses $\mu = M^{-1}$. The horizontal axis is r/M; the vertical axis is $M^4 \times$ [stress tensor].



Figure 4.4: Comparison of the two-dimensional stress tensor of BFN for the $\ell = 0$ mode to the four-dimensional result of AHS for a massless, minimally coupled scalar field. The upper curve is $\langle \hat{T}_r^T \rangle_{AHS}$; the lower curve is $(4\pi r^2)^{-1} \langle \hat{T}_r^T \rangle_{BFN}$. This plot uses $\mu = M^{-1}$. The horizontal axis is r/M; the vertical axis is $M^4 \times$ [stress tensor].



Figure 4.5: Comparison of the two-dimensional stress tensor of BFN for the $\ell = 0$ mode to the four-dimensional result of AHS when the dimensional-reduction anomaly is taken into account. From top to bottom near the horizon, the curves are $(4\pi r^2)^{-1}\Delta \langle \hat{\mathcal{T}}_t^t \rangle$, $(4\pi r^2)^{-1}[\langle \hat{\mathcal{T}}_t^t \rangle_{\text{BFN}} + \Delta \langle \hat{\mathcal{T}}_t^t \rangle]$, $\langle \hat{\mathcal{T}}_t^t \rangle_{\text{AHS}}$, $(4\pi r^2)^{-1} \langle \hat{\mathcal{T}}_t^t \rangle_{\text{BFN}}$. This plot uses $\mu = M^{-1}$. The horizontal axis is r/M; the vertical axis is $M^4 \times [\text{stress tensor}]$.



Figure 4.6: Comparison of the two-dimensional stress tensor of BFN for the $\ell = 0$ mode to the four-dimensional result of AHS when the dimensional-reduction anomaly is taken into account. From top to bottom near the horizon, the curves are $(4\pi r^2)^{-1}\Delta\langle\hat{\mathcal{T}}_r^r\rangle$, $(4\pi r^2)^{-1}[\langle\hat{\mathcal{T}}_r^r\rangle_{\text{BFN}} +$ $\Delta\langle\hat{\mathcal{T}}_r^r\rangle]$, $\langle\hat{\mathcal{T}}_r^r\rangle_{\text{AHS}}$, $(4\pi r^2)^{-1}\langle\hat{\mathcal{T}}_r^r\rangle_{\text{BFN}}$. This plot uses $\mu = M^{-1}$. The horizontal axis is r/M; the vertical axis is $M^4 \times [\text{stress tensor}]$.

the two- and four-dimensional stress tensors is much improved near the event horizon. In fact, we find the remarkable result that at r = 2M the anomalymodified stress tensor agrees *exactly* with the four-dimensional approximation of AHS.

As we move away from the horizon towards r = 3M, the match between the anomaly-corrected stress tensor and the AHS result is worse. This is likely due to the fact that the stress tensor of AHS includes the $\ell > 0$ modes, which become more significant at larger radii. It is also possible that the potential barrier, which is largely ignored by the WKB approximations of AHS and BFN, may

be of some relevance. (Note that the high frequency modes used in the WKB approximation penetrate the potential barrier with little reflection, and so it has little influence on these approximations.)

Finally, at very large distances from the black hole $(r \approx 10^2 M \text{ or greater})$, the anomaly is vanishingly small compared to the two-dimensional stress tensor, and so does not influence the asymptotic behaviour of the dimensionally reduced theory. At the same time, the geometry becomes flat and $\langle \hat{\mathcal{T}}_a{}^b \rangle_{\rm BFN}$ and $\langle \hat{\mathcal{T}}_a{}^b \rangle_{\rm AHS}$ reduce to the stress tensors for a massless scalar field at the Hartle-Hawking temperature in two and four dimensions. The results of Section 3.2 then assure us that in this regime the *s*-mode contribution to $\langle \hat{\mathcal{T}}_a{}^b \rangle_{\rm AHS}$ is precisely $(4\pi r^2)^{-1}$ times the dimensionally reduced stress tensor $\langle \hat{\mathcal{T}}_a{}^b \rangle_{\rm BFN}$.

This analysis indicates that dilaton-gravity models may indeed be used to model vacuum polarization and particle creation effects for massless fields in higher-dimensional black-hole spacetimes, but only if the dimensional-reduction anomaly is taken into account. Specifically, it appears that in order to obtain the correct expectation values in the vicinity of the event horizon from a reduced theory one must include the contributions of the dimensional-reduction anomaly. It should be noted, however, that while the improvements in the dimensionally reduced stress tensor due to the anomaly are impressive, they are still tentative. For example, the anomaly for the $\ell = 0$ mode cannot be fixed exactly due to the inherent ambiguity in low-frequency structure of the divergences for a quantum field. Furthermore, the anomaly for the massless field contains an arbitrary infrared cutoff parameter μ which has a strong influence on the anomalous stress tensor. The value $\mu \approx 1$ which was used in Figures 4.5 and 4.6 is merely that which gives the closest match between the dimensionally reduced theory and the full four-dimensional stress tensor near the horizon (though the exact agreement at r = 2M occurs for all μ). Finally, the accuracy of the two-dimensional stress tensor of BFN has not been determined; indeed, one would expect from general principles that it should also contain the cutoff parameter μ .

These difficulties could be reduced and our conclusions strengthened if the dimensionally reduced stress tensors for $\ell \geq 0$ could be calculated accurately. In this case, by combining the stress tensors for each ℓ with the corresponding anomaly and summing over even the first few ℓ a much better match with the four-dimensional stress tensor may be obtained in the vicinity of the horizon and the potential barrier. In particular, the effect of the low-frequency ambiguity in the anomaly would be lessened since the sum of the anomaly over all modes is well-defined. In any case, it is clear that the dimensional-reduction anomaly induces very significant alterations in the predictions of dilaton-gravity models for local observables in the vicinity of a black hole, and likely must be included for a proper comparison to physics in higher dimensions.

4.4.2 Large-Mass Case

For $mM \gg 1$, the Compton wavelength m^{-1} of the quantum field is small compared to the length scale M over which the geometry changes. In this case one can use the Schwinger-DeWitt expansion for the heat kernel directly as an approximate solution for the system. Taking the coincidence limit of (4.5) and integrating over s results in an expansion for the effective action in terms of $a_n/m^{2n} \approx (R/m^2)^n \approx (mM)^{-2n}$. For large mM, one can obtain an approximate effective action for the field by simply discarding the divergent terms in the Schwinger-DeWitt expansion and keeping the first finite term (the a_3 term in four dimensions, or the a_2 term in two dimensions). The stress tensor for large field mass can then be calculated from this approximate effective action.

For four-dimensional black-hole spacetimes the Schwinger-DeWitt approxi-

mation has been calculated by Frolov and Zelnikov [16]. They found

$$\begin{split} \langle \hat{T}_{t}^{t} \rangle_{\rm DS} &= A \left[\left(-285 \frac{M^{2}}{r^{2}} + 626 \frac{M^{3}}{r^{3}} \right) + 14 \left(\xi - \frac{1}{6} \right) \left(360 \frac{M^{2}}{r^{2}} - 792 \frac{M^{3}}{r^{3}} \right) \right] \,, \\ \langle \hat{T}_{r}^{r} \rangle_{\rm DS} &= A \left[\left(105 \frac{M^{2}}{r^{2}} - 154 \frac{M^{3}}{r^{3}} \right) + 14 \left(\xi - \frac{1}{6} \right) \left(-144 \frac{M^{2}}{r^{2}} + 216 \frac{M^{3}}{r^{3}} \right) \right] \,, \\ \langle \hat{T}_{\theta}^{\theta} \rangle_{\rm DS} &= A \left[\left(-315 \frac{M^{2}}{r^{2}} + 734 \frac{M^{3}}{r^{3}} \right) + 14 \left(\xi - \frac{1}{6} \right) \left(432 \frac{M^{2}}{r^{2}} - 1008 \frac{M^{3}}{r^{3}} \right) \right] \,, \\ (4.86) \end{split}$$

where $A \equiv (10080\pi^2 m^2 r^6)^{-1}$. Anderson, Hiscock, and Samuel [12] have verified numerically that (4.84)-(4.86) is a good approximation to the total stress tensor for the Hartle-Hawking state near the event horizon in four dimensions for mM > 2.

In the dimensionally reduced Schwarzschild geometry the Schwinger-DeWitt approximation for the $\ell = 0$ mode is easily calculated, and is precisely the negative of the large mr limit of the anomaly, (4.75)-(4.77). We denote this stress tensor by $\langle \hat{\mathcal{T}}_a \rangle_{\text{DS}}$.

One can compare the Schwinger-DeWitt approximations in two and four dimensions just as we compared the stress tensors for the massless field; the results are shown in Figure 4.7. As in the massless case, we see that the stress tensor predicted by dimensional reduction for one mode alone is much larger than the full stress tensor in four dimensions. Again, the cause (and cure) are clear. In four dimensions W_{DS} is constructed starting with the a_3 Schwinger-DeWitt coefficient, which is $O(R^3)$. In two dimensions the first contributions to W_{DS} come from the a_2 coefficient, which is $O(R^2)$. To reproduce the four-dimensional results, this $O(R^2)$ contribution must be removed somehow. The dimensional-reduction anomaly is the difference between the mode-decomposed subtraction terms from four dimensions (the a_0 , a_1 , and a_2 terms) and those from two dimensions (the a_0 and a_1 terms); i.e., it is essentially the negative of the mode-decomposed a_2 from four dimensions. This is why adding the anomaly to the stress tensor in the twodimensional Schwinger-DeWitt approximation exactly cancels the leading (a_2) contributions. Thus, by taking the dimensional-reduction anomaly into account, the leading mode-by-mode contributions to the stress tensor in four dimensions will be $O(R^3)$, as expected.



Figure 4.7: Comparison of the stress tensors from the Schwinger-DeWitt approximations for the $\ell = 0$ mode in two dimensions and the total stress tensor in four dimensions. The plot uses $\xi = 0$, mM = 1. From top to bottom near the event horizon, the curves are $(4\pi r^2)^{-1} \langle \hat{T}_r^{\tau} \rangle_{\text{DS}}, (4\pi r^2)^{-1} \langle \hat{T}_t^{t} \rangle_{\text{DS}}, \langle \hat{T}_r^{\tau} \rangle_{\text{DS}}, \langle \hat{T}_t^{t} \rangle_{\text{DS}}$, The horizontal axis is r/M; the vertical axis is $90\pi^2(8M)^4 \times [\text{stress tensor}]$.

From these examples it is clear that the dimensional-reduction anomaly induces very significant corrections to the contribution of modes of fixed angular momentum to the stress tensor when calculated from spherically reduced models. It is thus vital to account for the anomaly if one is to obtain the correct stress tensor for a quantum field in the Schwarzschild spacetime from dimensional reduction. (In fact, it can be shown that the sum of the anomaly over all modes is divergent, so that the total error in the four-dimensional stress tensor from ignoring the anomaly is infinite.) Note, however, that the anomalous stress tensor does not contribute to the Hawking radiation, as the flux terms $\Delta \langle \hat{T}_{tr} \rangle$ vanish. More generally, we see that the anomaly vanishes at large r in the dimensionally reduced spacetime, and so cannot affect the asymptotic behaviour of the dimensionally reduced system. This indicates that the dimensional-reduction anomaly is not associated with the negative Hawking flux from spherically reduced gravity obtained by some authors [37].

Chapter 5

Static Decompositions

In this chapter we consider the dimensional reduction of a scalar field when the internal space is flat and of dimension n = 1 or n = 2. If one of the internal-space coordinates is the Euclidean "time," and if the internal space metric is independent of this coordinate, then these spaces are static. As we saw in Chapter 2, static spaces are physically very important for their close connection to quantum field theory at finite temperature. Specifically, a quantum scalar field which is periodic in Euclidean time with period β corresponds to a thermal state at temperature β^{-1} .

In the following sections we calculate the dimensional-reduction anomaly in static spaces. For a finite-temperature field $(0 < \beta < \infty)$ we derive the anomaly in $\langle \hat{\Phi}^2 \rangle$ for a one-dimensional internal space. For zero-temperature fields $(\beta \to \infty)$ we calculate the anomalies in $\langle \hat{\Phi}^2 \rangle$ and W for both n = 1 and n = 2. We begin in Section 5.1 with the anomaly for the decomposition of flat space in Rindler coordinates, where the internal space is compact $(t \in [0, 2\pi])$ with n = 1. Sections 5.2 and 5.3 deal with noncompact $(\beta \to \infty)$ internal spaces. Finally, in Section 5.4 we derive the dimensional-reduction anomaly for the case n = 1 when the time coordinate has arbitrary periodicity β .
5.1 Rindler space

In our consideration of dimensional reductions in spherically symmetric spaces, we began with the simplest example of flat space. This system had the advantage of being exactly solvable, which aided in the physical interpretation of the dimensional-reduction anomaly.

To begin our study of dimensional reductions in static spaces, we shall again begin with the simplest nontrivial example: quantum theory in Rindler space [84]. Because Rindler space is simply flat space in non-inertial coordinates, we can again perform mode-decomposition calculations exactly. We will find results closely analogous to those for spherical decompositions in flat space.

Consider a scalar field obeying the field equation (2.55) in Euclidean Rindler space,

$$ds^2 = z^2 dt^2 + dz^2 + dx^2 + dy^2 , \qquad (5.1)$$

where $t \in [0, 2\pi]$, $z \in [0, \infty)$, $x, y \in (-\infty, \infty)$, and the points $t = 0, t = 2\pi$ are identified. This corresponds to the special case of the line element (2.53) where $x^a = (x, y, z), y^i = t, h_{ab} = \delta_{ab}, \Omega_{ij} = 1$, and $\rho e^{-2\phi} = z$. The Rindler-space line element may be obtained from the standard flat-space line element

$$ds^2 = dT^2 + dX^2 + dY^2 + dZ^2$$
(5.2)

by the coordinate transformation

$$T = z \sin(t), \quad X = x,$$

$$Z = z \cos(t), \quad Y = y,$$
(5.3)

and is clearly just flat space in polar coordinates.

Since Euclidean Rindler space is not periodic in the usual Minkowski time coordinate T of (5.3), an inertial observer will see a zero-temperature state. However, a line of fixed Rindler coordinates (x, y, z) is periodic in Rindler time with period $2\pi z$; hence a Rindler observer at fixed z will find the quantum field to be in a thermal state with temperature $(2\pi z)^{-1}$. This is a statement of the well-known Unruh effect: for two systems in relative accelerated motion, the vacuum of one system appears relative to the other as a state with thermally distributed particles at a temperature which is proportional to the acceleration [67,85].

Because the line element (5.1) is static, a quantum theory on this space may be dimensionally reduced by decomposing the field in terms of Rindler "time" modes $\cos(k[t-t'])/\sqrt{2\pi}$. We wish to calculate the anomaly associated with this dimensional reduction.

We assume that the potential V is *t*-independent and vanishes in the region of interest. Hence, the Green function for a given state in four dimensions is renormalized by subtracting the Euclidean vacuum Green function (3.63), where

$$2\sigma = (x - x')^2 + (y - y')^2 + (z - z')^2 + 2zz' [1 - \cos(t - t')] .$$
 (5.4)

Since any physical quantity must be periodic in t, we may decompose this Green function in terms of the Rindler time as an ordinary Fourier cosine series,¹

$$G_{\rm div}(X,X') = \frac{1}{2\pi\sqrt{zz'}} \sum_{k=-\infty}^{\infty} \cos(k[t-t']) G_{\rm div|k}(\vec{x},\vec{x}'), \qquad (5.5)$$

$$G_{\rm div|k}(\vec{x},\vec{x}') = \sqrt{zz'} \int_0^{2\pi} d(t-t') \cos(k[t-t']) G_{\rm div}(X,X'), \qquad (5.6)$$

where $\vec{x} = (x, y, z)$; note that $G_{\text{div}|-k} = G_{\text{div}|k}$. One may verify that $G_{\text{div}|k}(\vec{x}, \vec{x}')$ is a solution of the three-dimensional Green-function equation

$$\mathcal{F}_{k} \mathcal{G}(\vec{x}, \vec{x}') \equiv (\Delta - m^{2} - V_{k}) \mathcal{G}_{k}(\vec{x}, \vec{x}') = -\delta(x - x') \,\delta(y - y') \,\delta(z - z') \,, \quad (5.7)$$

where $\Delta = \partial_x^2 + \partial_y^2 + \partial_z^2$ and

$$V_k(z) = \frac{4k^2 - 1}{4z^2}, \qquad (5.8)$$

¹This is a special case of the decomposition for arbitrary β discussed in more detail in Section 5.4.

and which vanishes at $z = \rho e^{-2\phi} = 0$, z' = 0. [The potential (5.8) follows from (2.71) with $\lambda \to k^2$, where V is assumed to vanish in the region of interest.] We may obtain an explicit expression for $G_{\text{div}|k}(\vec{x}, \vec{x}')$ by decomposing the fourdimensional Green function as in (5.6). Using the integral representation for K_1 from Appendix C.3.1 (i.e., the heat kernel representation for G_{div}), combined with the integrals

$$\int_{0}^{2\pi} dx \cos(kx) e^{p \cos(x)} = 2\pi I_k(p)$$
(5.9)

and

$$\int_{0}^{\infty} dx \, \frac{1}{x} \, \mathrm{e}^{-ax - \frac{b}{x}} I_{k}(cx) = 2I_{k}(\sqrt{b(a+c)} - \sqrt{b(a-c)}) \\ \times K_{k}(\sqrt{b(a+c)} + \sqrt{b(a-c)})$$
(5.10)

(see, for example, 5.1.937.2 of [76] and 2.15.6.4 of [74] respectively), one can show that

$$G_{\text{div}|k}(\vec{x}, \vec{x}') = -\frac{m^2}{4\pi} \sqrt{zz'} \left[\left(I_{k+1}(\alpha_-) + \frac{k}{\alpha_-} I_k(\alpha_-) \right) K_k(\alpha_+) \left(\frac{1}{md_+} - \frac{1}{md_-} \right) - I_k(\alpha_-) \left(K_{k+1}(\alpha_+) - \frac{k}{\alpha_+} K_k(\alpha_+) \right) \left(\frac{1}{md_+} + \frac{1}{md_-} \right) \right], \quad (5.11)$$

where we have defined

$$\alpha_{\pm} = \frac{m}{2}(d_{+} \pm d_{-}) , \qquad d_{\pm} = \sqrt{(x - x')^{2} + (y - y')^{2} + (z \pm z')^{2}} . \tag{5.12}$$

Note the similarity of d_{\pm} to σ^{\pm} of (3.75).

The physical observable of interest, $\langle \hat{\Phi}^2 \rangle_{ren}$, can be calculated from the full Green function G(X, X') with a *t*-independent boundary or potential outside the region of interest using

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = \lim_{X' \to X} \left[G(X, X') - G_{\rm div}(X, X') \right] \,.$$
 (5.13)

Decomposing G in the same manner as G_{div} then allows us to write a decomposed form for $\langle \hat{\Phi}^2 \rangle_{ren}$:

$$\langle \hat{\Phi}^2(\vec{x}) \rangle_{\text{ren}} = \frac{1}{2\pi z} \lim_{\vec{x}' \to \vec{x}} \left\{ \sum_{k=-\infty}^{\infty} [\mathcal{G}_k(\vec{x}, \vec{x}') - G_{\text{div}|k}(\vec{x}, \vec{x}')] \right\}.$$
 (5.14)

On the other hand, the renormalized value of $\langle \hat{\varphi}_k^2 \rangle$ for the three-dimensional operator \mathcal{F}_k in (5.7) is obtained by subtracting from the full three-dimensional Green function $\mathcal{G}_k(\vec{x}, \vec{x}')$ not $\mathcal{G}_{\text{div}|k}(\vec{x}, \vec{x}')$, but rather the first term in the Schwinger-DeWitt expansion for \mathcal{F}_k , denoted $\mathcal{G}_{k|\text{div}}(\vec{x}, \vec{x}')$. Specifically, for each k we have

$$\langle \hat{\varphi}_k^2(\vec{x}) \rangle_{\text{ren}} = \lim_{\vec{x}' \to \vec{x}} \left[\mathcal{G}_k(\vec{x}, \vec{x}') - \mathcal{G}_{k|\text{div}}(\vec{x}, \vec{x}') \right] , \qquad (5.15)$$

where²

$$\mathcal{G}_{k}^{\text{div}}(\vec{x}, \vec{x}') = \left(\frac{m}{4\pi^{2}\sqrt{2\sigma}}\right)^{\frac{1}{2}} K_{\frac{1}{2}}(m\sqrt{2\sigma}) .$$
 (5.16)

Comparing (5.14) to (5.15) we find

$$\langle \hat{\Phi}^2(\vec{x}) \rangle_{\rm ren} = \frac{1}{2\pi z} \sum_{k=-\infty}^{\infty} \left[\langle \hat{\varphi}_k^2(\vec{x}) \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_k^2(\vec{x}) \rangle \right] , \qquad (5.17)$$

where for each k the anomaly is

$$\begin{aligned} \Delta \langle \hat{\varphi}_{k}^{2}(\vec{x}) \rangle &= \lim_{\vec{x}' \to \vec{x}} \left[\mathcal{G}_{k|\operatorname{div}}(\vec{x}, \vec{x}') - G_{\operatorname{div}|k}(\vec{x}, \vec{x}') \right] \end{aligned} (5.18) \\ &= \frac{m}{4\pi} \left[-1 + mz \left(I_{k+1}(mz) K_{k+1}(mz) + I_{k}(mz) K_{k}(mz) \right) + k \left(I_{k}(mz) K_{k+1}(mz) - I_{k+1}(mz) K_{k}(mz) \right) \right] . \end{aligned} (5.19)$$

Once again, the anomaly is seen as the difference in the subtraction terms used to renormalize the higher- and lower-dimensional theories. It is easily shown that (5.19) falls off as $O(z^{-1})$ for large mz, and diverges as $O(z^{-1})$ when $mz \to 0$ for $k \neq 0$. For k = 0 the anomaly is finite as $mz \to 0$.

The divergence of the anomaly as $z \to 0$ is familiar from the spherical decompositions, and arises for the same reason. As in the spherical decomposition of flat space, the dimensionally reduced theory obeys Dirichlet boundary conditions (i.e. the Green function vanishes) at $z = \rho e^{-2\phi} = 0$. We therefore expect divergences in the three-dimensional renormalized expectation values at z = 0

²Equation (5.16) can be obtained from the Schwinger-DeWitt expansion in three dimensions [41].

due to the vacuum polarization produced by this boundary. These are cancelled by the dimensional-reduction anomaly, leaving finite expectation values in four dimensions, as expected for flat space.

These results are qualitatively very similar to those from the spherical decomposition of flat spacetime in Section 3.3. This should not be surprising, considering that the Euclidean Rindler space (5.1) is simply flat space in polar coordinates. As a result, the calculations of this section amount to a one-dimensional "spherical" decomposition of flat space.

To be precise, recall that mode decompositions take advantage of a continuous symmetry of the manifold. For flat space, there are three distinct sets of continuous symmetries (the Poincare group): translations, rotations, and boosts. The translations are fairly trivial; one can get the dimensional-reduction anomaly for these by taking $\phi \rightarrow 0$ in the results of the next section. Rotations were covered in Chapter 3. Translations in Rindler time are the boosts, which mix the Cartesian time and space coordinates. Of course, in Euclidean space they are just another set of rotations. As a result, the decompositions of this section are the one-dimensional counterparts of the two-dimensional decompositions of Section 3.3.

5.2 The Dimensional-Reduction Anomaly in $\langle \hat{\Phi}^2 angle$

In the previous section we examined the dimensional-reduction anomaly for the decomposition of a field in flat space into Rindler time modes. There the internal space was compact and of dimension 1. In the next two sections we consider general static spaces with internal spaces which are noncompact and of dimension 1 or 2. While qualitatively different from Rindler space, we shall see in Section 5.4

how the anomalies in these two examples may be related.

5.2.1 (1+3) Reduction

We begin with the case n = 1 and write the metric (2.53) in the form

$$ds^{2} = g_{\mu\nu} dX^{\mu} dX^{\nu} = e^{-4\phi(x)} dt^{2} + h_{ab}(x) dx^{a} dx^{b}, \qquad (5.20)$$

where $a, b \in \{1, 2, 3\}$. We assume that $t \in (-\infty, \infty)$, corresponding to a zerotemperature state. The scalar field operator is again taken to be $F = \Box - m^2 - V$, where the potential V is independent of Euclidean "time" t. It is easy to see that the operator Δ_{Ω} for the metric (5.20) is $\partial^2/\partial t^2$. Hence, the mode decomposition in terms of its eigenvalues is simply the standard Fourier transform with

$$Y_{\omega}(t) = \frac{\exp(-i\omega t)}{\sqrt{2\pi}}.$$
 (5.21)

In the language of Section 2.2 we have

$$\lambda = \omega^2, \qquad \sum_{\lambda,\varrho} = \int_{-\infty}^{\infty} d\omega.$$
 (5.22)

For example, the orthogonality and completeness relations become

$$\int_{-\infty}^{\infty} \frac{dt}{2\pi} e^{-i(\omega - \omega')t} = \delta(\omega - \omega'), \qquad (5.23)$$

$$\int_{-\infty}^{\infty} \frac{d\omega}{2\pi} e^{-i\omega(t-t')} = \delta(t-t').$$
(5.24)

The bare $\langle \hat{\Phi}^2 \rangle$ is obtained from the coincidence limit of the Green function,

$$\langle \hat{\Phi}^2(X) \rangle = \lim_{X' \to X} G(X, X') \,. \tag{5.25}$$

We saw in the previous chapter that for a general four-dimensional space the divergences of the Green function in the coincidence limit come from the first two terms of the Schwinger-DeWitt expansion of the heat kernel. From (4.11) and (4.12),

$$G_{\rm div}(t,x;t',x') = \int_0^\infty ds \, \frac{1}{(4\pi s)^2} \exp\left\{-m^2 s - \frac{2\sigma + \epsilon^2}{4s}\right\} \left[\Re_0^{\Box - V} + s\Re_1^{\Box - V}\right] \,.$$
(5.26)

Here $\sigma = \sigma(X, X')$ is one-half of the square of the geodesic distance between points $X = (t, \vec{x})$ and $X' = (t', \vec{x}')$, $\Re_n^{\Box - V}$ is defined as in (4.13), D(X, X') is the Van Vleck determinant (4.6), and the first few Schwinger-DeWitt coefficients for arbitrary potential V in the coincidence limit $X' \to X$ are

$$a_0^{\Box - V} = 1, \qquad (5.27)$$

$$a_1^{\Box - V} = \frac{1}{6} {}^4 R - V , \qquad (5.28)$$

$$a_{2}^{\Box-V} = \frac{1}{180} \left[{}^{4}R_{\alpha\beta\gamma\delta} {}^{4}R^{\alpha\beta\gamma\delta} - {}^{4}R_{\alpha\beta} {}^{4}R^{\alpha\beta} \right] + \frac{1}{2} \left(\frac{1}{6} {}^{4}R - V \right)^{2} + \frac{1}{30} \Box {}^{4}R - \frac{1}{6} \Box V .$$
 (5.29)

Expansions of σ and the $\Re_n^{\Box-V}$ for the metric (5.20) with x = x' and $e^{-2\phi}(t - t')$ small are given in Appendix C.1. We also include a cut-off parameter ϵ in the exponent of the integrand in (5.26) which ensures convergence of the integral for small *s* with σ vanishing. The anomaly is independent of this regularization parameter, and it is useful for displaying intermediate results. (This is the way the calculations of the anomaly were done in Chapter 4; ϵ did not appear explicitly since only final results were displayed.)

Our purpose is to compare the divergences of four- and three-dimensional theories related by a Fourier time transform. Unfortunately, it is not possible to evaluate the Fourier transform of (5.26) exactly for general h_{ab} , as σ and the $\Re_n^{\Box-V}$ are known in the general case only for infinitesimal point separations. Our response is the same as that in Chapter 4: we make point splitting in the *t*-direction, expand all *t*-dependent quantities in powers of the curvature, and truncate all expressions at first order in the curvature (two derivatives of the

dilaton or metric). Denoting $\tau \equiv e^{-2\phi}(t-t')$ and putting x = x' we have up to first order in the curvature

$$2\sigma(t,x;t',x) = \tau^2 - \frac{1}{3}(\nabla\phi)^2\tau^4, \qquad (5.30)$$

$$\Re_0^{\Box - V} = 1 + \frac{1}{6} \Box \phi \tau^2, \qquad (5.31)$$

$$\Re_1^{\Box-V} = \frac{1}{6} {}^4R - V. \qquad (5.32)$$

See Appendix C.1 for details. Note that in this case the internal space is not compact, and so there is no need to make our expansions periodic in (t - t').

Substituting (5.30) into (5.26), expanding the exponent, and keeping in the exponent only terms which are quadratic in τ , we get

$$G_{\rm div}(t,x;t',x) = \int_0^\infty ds \; \frac{\mathrm{e}^{-m^2 s - (\tau^2 + \epsilon^2)/4s}}{(4\pi s)^2} \left[\left(1 + \frac{(\nabla \phi)^2}{12s} \tau^4 \right) \Re_0^{\Box - V} + s \Re_1^{\Box - V} \right] \,. \tag{5.33}$$

The integral over the parameter s can be taken with the following result (see Appendix C.3.1):

$$G_{\text{div}}(t,x;t',x) = \frac{1}{8\pi^2} \left[\left(\frac{1}{6} {}^4R - V \right) \mathbf{K}_0(z) + m^2 \left(1 + \frac{1}{6} \Box \phi \tau^2 \right) \mathbf{K}_1(z) + m^4 \frac{(\nabla \phi)^2}{12} \tau^4 \mathbf{K}_2(z) \right].$$
(5.34)

Here $z \equiv m\sqrt{\tau^2 + \epsilon^2}$,

$$\mathbf{K}_{\nu}(z) = \left(\frac{2}{z}\right)^{\nu} K_{\nu}(z) , \qquad (5.35)$$

and the K_{ν} are modified Bessel functions. For example, putting $\epsilon = 0$ and expanding G_{div}^{F} in a Laurent series in τ we get

$$G_{div}(t,x;t',x) = \frac{1}{4\pi^{2}\tau^{2}} - \frac{1}{8\pi^{2}} \left[\frac{1}{6} {}^{4}R - V - m^{2} \right] \left\{ \gamma + \frac{1}{2} \ln \left(\frac{m^{2}\tau^{2}}{4} \right) \right\} - \frac{m^{2}}{16\pi^{2}} + \frac{(\nabla\phi)^{2}}{12\pi^{2}} + \frac{\Box\phi}{24\pi^{2}} + \dots$$
(5.36)

Here γ is the Euler constant, and the dots denote terms of higher order in τ . The terms displayed are just the usual DeWitt-Schwinger expansion for the divergent parts of G for the metric (5.20).

The renormalized value of $\langle \hat{\Phi}^2 \rangle$ can be written in the form

$$\langle \hat{\Phi}^2(t,x) \rangle_{\text{ren}} = \lim_{\epsilon \to 0} \lim_{t \to t' \to 0} \left[G(t,x;t',x) - G_{\text{div}}(t,x;t',x) \right] \,. \tag{5.37}$$

Taking the limit $\epsilon \to 0$ in this expression is a trivial operation since the difference in the square brackets is already a finite quantity.

Let us now analyze what happens when we mode decompose G_{div} and compare to the corresponding divergent terms from the three-dimensional theory. Following (2.86) and (5.21)-(5.24), the Fourier time-transform pair is defined as

$$G_{\rm div}(x;x'|\omega) = e^{-(\phi+\phi')} \int_{-\infty}^{\infty} d(t-t') e^{i\omega(t-t')} G_{\rm div}(t,x;t',x'), \qquad (5.38)$$

$$G_{\rm div}(t,x;t',x') = e^{(\phi+\phi')} \int_{-\infty}^{\infty} \frac{d\omega}{2\pi} e^{-i\omega(t-t')} G_{\rm div}(x;x'|\omega) \,. \tag{5.39}$$

Since $G_{div}(t, x; t', x')$ depends only on the difference t-t', the function $G_{div}(x; x'|\omega)$ does not depend on t and t'. Calculating the integral in (5.38) using (5.34) and (C.57) we obtain

$$G_{\rm div}(x;x|\omega) = \frac{1}{4\pi} \left[\frac{1}{\epsilon} - \mu + \frac{1}{2\mu} \left(\frac{1}{6} {}^4R - V \right) + \frac{m^2 \Box \phi}{6\mu^3} + \frac{m^4 (\nabla \phi)^2}{2\mu^5} + O(\epsilon) \right],$$
(5.40)

where $\mu \equiv \sqrt{m^2 + e^{4\phi}\omega^2}$. Meanwhile, the operator \mathcal{F}_{ω} which determines the reduced equation of motion (2.72) is

$$\mathcal{F}_{\omega} = \Delta_h - m^2 - V_{\omega}[\phi], \qquad (5.41)$$

where

$$V_{\omega}[\phi] = V + \omega^2 e^{4\phi} + (\nabla \phi)^2 - \Delta_h \phi \,. \tag{5.42}$$

The Schwinger-DeWitt expansion of the heat kernel [41] for the operator \mathcal{F}_{ω} in three dimensions is

$$\mathcal{K}_{\omega}(x,x|s) = \frac{1}{(4\pi s)^{\frac{3}{2}}} \exp\left\{-m^2 s\right\} \left[1 + s\left(\frac{1}{6}R - V_{\omega}\right) + \cdots\right] \,. \tag{5.43}$$

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

The divergent part of the Green function for \mathcal{F}_{ω} is generated by the first term in this expansion,

$$\mathcal{G}_{\omega|\text{div}}(x;x) = \int_0^\infty ds \, \frac{1}{(4\pi s)^{\frac{3}{2}}} \exp\left\{-m^2 s - \frac{\epsilon^2}{4s}\right\} = \frac{1}{4\pi} \left[\frac{1}{\epsilon} - m + O(\epsilon)\right] \,. \tag{5.44}$$

Hence, if we start with a three-dimensional theory with the field equation

$$\mathcal{F}_{\omega}\,\hat{\varphi}_{\omega}(x) = 0\,, \qquad (5.45)$$

we will obtain for the renormalized value of $\langle \hat{\varphi}_{\omega}^2 \rangle$ the representation

$$\langle \hat{\varphi}^2_{\omega}(x) \rangle_{\text{ren}} = \lim_{\epsilon \to 0} \left[\mathcal{G}_{\omega}(x;x) - \mathcal{G}_{\omega|\text{div}}(x;x) \right]$$
 (5.46)

By comparing (5.37) with (5.46) we can get the following relation between $\langle \hat{\Phi}^2 \rangle$ in the four- and three-dimensional theories:

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = e^{2\phi} \int_{-\infty}^{\infty} \frac{d\omega}{2\pi} \left[\langle \hat{\varphi}_{\omega}^2 \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_{\omega}^2 \rangle \right] , \qquad (5.47)$$

where the anomalous term is

$$\Delta \langle \hat{\varphi}_{\omega}^{2} \rangle = \lim_{\epsilon \to 0} \left[\mathcal{G}_{\omega | \operatorname{div}}(x; x) - G_{\operatorname{div}}(x; x | \omega) \right]$$
(5.48)

$$= \frac{1}{4\pi} \left[\mu - m - \frac{1}{2\mu} \left(\frac{1}{6} {}^{4}R - V \right) - \frac{m^{2} \Box \phi}{6\mu^{3}} - \frac{m^{4} (\nabla \phi)^{2}}{2\mu^{5}} \right]. \quad (5.49)$$

Since $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle$ does not vanish we have another example of the dimensional-reduction anomaly.

It is useful to note that for large frequencies $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle$ is proportional to ω . Thus, the integral over ω of the anomaly diverges as $\omega \to \pm \infty$. Since the renormalized $\langle \hat{\Phi}^2 \rangle$ in four dimensions should be finite, this implies that the integral of $\langle \hat{\varphi}_{\omega}^2 \rangle_{\text{ren}}$ over ω also diverges. Thus, if one attempted to calculate $\langle \hat{\Phi}^2 \rangle_{\text{ren}}$ in four dimensions by summing over the $\langle \hat{\varphi}_{\omega}^2 \rangle_{\text{ren}}$, the result would diverge. Similar results held for the spherical reductions considered in Chapters 3 and 4.

5.2.2 (2+2) Reduction

Let us discuss now the dimensional-reduction anomaly for $\langle \hat{\Phi}^2 \rangle$ for the case when the metric of the internal space is flat and two-dimensional; that is, the spacetime metric is of the form

$$ds^{2} = e^{-2\phi(x)} (dt_{0}^{2} + dt_{1}^{2}) + h_{AB}(x) dx^{A} dx^{B}, \qquad (5.50)$$

where $A, B \in \{2, 3\}$. First of all, it should be noticed that this metric is a special case of the metric (5.20) when the dilaton field does not depend on one of the coordinates x^a . Equation (5.50) can be obtained from (5.20) by rescaling the dilaton field $\phi \rightarrow \phi/2$ and putting

$$h_{ab}dx^{a}dx^{b} = e^{-2\phi(x)} dt_{1}^{2} + h_{AB}(x)dx^{A}dx^{B}.$$
 (5.51)

For the divergent part of the four-dimensional Green function expanded to first order in the curvature we have an expression similar to (5.34),

$$G_{div}(\mathbf{t}, x; \mathbf{t}', x) = \frac{1}{8\pi^2} \left[m^2 \mathbf{K}_1(z) + \frac{(\nabla \phi)^2}{48} m^4 \tau^4 \mathbf{K}_2(z) + \frac{\Box \phi}{12} m^2 \tau^2 \mathbf{K}_1(z) + \left(\frac{1}{6} {}^4 R - V\right) \mathbf{K}_0(z) \right].$$
(5.52)

See Appendix C.2. Here ${}^{4}R$ refers to the curvature of the full four-dimensional space with metric $g_{\mu\nu}$ of (5.50), and we define

$$z = m\sqrt{\tau^2 + \epsilon^2}, \quad \tau^2 = e^{-2\phi} \mathbf{t}^2 \equiv e^{-2\phi} (t_0^2 + t_1^2).$$
 (5.53)

To obtain the mode decomposition of the divergent part of the Green function we make a Fourier transform similar to (5.38)

$$G_{\rm div}(x;x'|p) = e^{-(\phi+\phi')} \int_{-\infty}^{\infty} d(\mathbf{t}-\mathbf{t}') e^{i\mathbf{p}(\mathbf{t}-\mathbf{t}')} G_{\rm div}(\mathbf{t},x;\mathbf{t}',x'), \qquad (5.54)$$

$$G_{\rm div}(\mathbf{t}, x; \mathbf{t}', x') = e^{(\phi + \phi')} \int_{-\infty}^{\infty} \frac{d\mathbf{p}}{(2\pi)^2} e^{-i\mathbf{p}(\mathbf{t} - \mathbf{t}')} G_{\rm div}(x; x'|p) .$$
(5.55)

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Here we use the vector notations $\mathbf{p} = (p_0, p_1)$ and $\mathbf{pt} = p_0 t_0 + p_1 t_1$. We also denote $p^2 = \mathbf{p}^2$. Since the function $G_{div}(\mathbf{t}, x; \mathbf{t}', x')$ depends only on the difference $\mathbf{t} - \mathbf{t}'$, its Fourier transform depends only on p. Calculating the integrals we obtain

$$G_{div}(x;x'|p) = \frac{1}{2\pi} \left[-\left\{ \gamma + \frac{1}{2} \ln\left(\frac{\mu^2 \epsilon^2}{4}\right) \right\} + \frac{1}{2\mu^2} \left(\frac{1}{6} {}^4R - V\right) + \frac{m^2}{6\mu^4} \Box \phi + \frac{m^4}{3\mu^6} (\nabla \phi)^2 \right] + O(\epsilon), \quad (5.56)$$

where

$$\mu = \sqrt{m^2 + p^2 e^{2\phi}} \,. \tag{5.57}$$

The adopted mode expansion into plane waves $\exp(i\mathbf{pt})/2\pi$ reduces the initial system for the four-dimensional operator $F = \Box - m^2 - V$ a two-dimensional system with a dilaton-dependent potential. The corresponding wave operator \mathcal{F}_p is

$$\mathcal{F}_p = \Delta_h - m^2 - V_p[\phi], \qquad (5.58)$$

where

$$V_{p}[\phi] = V + p^{2} e^{2\phi} + (\nabla \phi)^{2} - \Delta_{h} \phi.$$
 (5.59)

The divergent part of the two-dimensional Green function for the operator \mathcal{F}_p can be obtained from the first term of the Schwinger-DeWitt expansion for this operator, (4.10) with $V_{\ell} \to V_p$. With the regularization parameter ϵ inserted,

$$\mathcal{G}_{p|\operatorname{div}}(x;x) = \int_0^\infty ds \, \frac{1}{4\pi s} \exp\left\{-m^2 s - \frac{\epsilon^2}{4s}\right\} = -\frac{1}{2\pi} \left\{\gamma + \frac{1}{2} \ln\left(\frac{m^2 \epsilon^2}{4}\right)\right\} + O(\epsilon) \,. \tag{5.60}$$

We define the four- and two-dimensional renormalized values $\langle \hat{\Phi}^2(x) \rangle_{\text{ren}}$ and $\langle \hat{\varphi}_p^2(x) \rangle_{\text{ren}}$ by expressions similar to (5.37) and (5.46) respectively. By comparing these definitions, and using relations (5.56) and (5.60), we obtain the representation

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = e^{2\phi} \int \frac{d\mathbf{p}}{(2\pi)^2} \left[\langle \hat{\varphi}_p^2 \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_p^2 \rangle \right] , \qquad (5.61)$$

where the dimensional-reduction anomaly $\Delta \langle \hat{\varphi}_{p}^{2} \rangle$ is

$$\Delta \langle \hat{\varphi}_{p}^{2} \rangle = \frac{1}{4\pi} \left[\ln \left(\frac{\mu^{2}}{m^{2}} \right) - \frac{1}{\mu^{2}} \left(\frac{1}{6} {}^{4}R - V \right) - \frac{m^{2}}{3\mu^{4}} \Box \phi - \frac{2m^{4}}{3\mu^{6}} (\nabla \phi)^{2} \right] .$$
(5.62)

Compare this to the anomaly for the (1+3) reduction, (5.49).

5.3 The Dimensional-Reduction Anomaly in the Effective Action

5.3.1 (1+3) Reduction

For the static spacetime (5.20), the calculation of the anomaly in the effective action W, where

$$W[g] = \int dX \sqrt{g} L , \qquad (5.63)$$

proceeds analogously to the calculation of the anomaly in $\langle \hat{\Phi}^2 \rangle$. To analyze the divergent part of the effective action we introduce first a point-split version of the effective Lagrangian³ L. Using the Schwinger-DeWitt expansion for the heat kernel, we have for the divergent part of L

$$L_{\rm div}(t,x;t',x) = -\frac{1}{2} \int_0^\infty \frac{ds}{s} \; \frac{e^{-m^2 s - (\tau^2 + \epsilon^2)/4s}}{(4\pi s)^2} \left[\Re_0^{\Box - V} + s \Re_1^{\Box - V} + s^2 \Re_2^{\Box - V} \right] \; .$$
(5.64)

As earlier, the points are split in the t direction. Since the internal space is homogeneous, the point-split Lagrangian depends on t and t' only in the combination t - t'.

Faced with the same problem as before, we expand σ and the \Re_n^{D-V} in terms of $\tau \equiv e^{-2\phi}(t-t')$ for x = x', this time truncating at second order in the curvature

³It is more convenient to work with L than W for noncompact internal spaces, as the spacetime volume integral (5.63) contains an infinite volume factor $V_{\Omega} \rightarrow \infty$.

(four derivatives of the metric or dilaton). Writing

$$2\sigma(t, x; t', x) = \tau^{2} + u\tau^{4} + v\tau^{6} + \cdots,$$

$$\Re_{n}^{\Box - V}(t, x; t', x) = \Re_{n(0)}^{\Box - V} + \Re_{n(2)}^{\Box - V} \tau^{2} + \Re_{n(4)}^{\Box - V} \tau^{4} + \cdots, \qquad (5.65)$$

we have $\Re_{0(0)}^{\Box-V} = 1$, while u, v, and the other $\Re_{n(k)}^{\Box-V}$ may be found in Appendix C.1. Inserting and truncating at $O(R^2)$ gives

$$L_{div}(t, x; t', x) = -\frac{1}{(4\pi)^2} \left[m^4 \mathbf{K}_2(z) - \frac{1}{4} (u\tau^4 + v\tau^6) m^6 \mathbf{K}_3(z) + \frac{1}{32} u^2 \tau^8 m^8 \mathbf{K}_4(z) + \Re_{0(2)}^{\Box - V} \tau^2 m^4 \left(\mathbf{K}_2(z) - \frac{1}{4} u\tau^4 m^2 \mathbf{K}_3(z) \right) + \Re_{0(4)}^{\Box - V} \tau^4 m^4 \mathbf{K}_2(z) + \Re_{1(0)}^{\Box - V} m^2 \left(\mathbf{K}_1(z) - \frac{1}{4} u\tau^4 m^2 \mathbf{K}_2(z) \right) + \Re_{1(2)}^{\Box - V} \tau^2 m^2 \mathbf{K}_1(z) + \Re_{2(0)}^{\Box - V} \mathbf{K}_0(z) \right].$$
(5.66)

The function $\mathbf{K}_{\nu}(z)$ is defined by (5.35).

We define the Fourier transform of L_{div} as in (5.38). (This follows from using the heat-kernel decomposition formula (2.92) and the mode (5.21) from the previous section.) Evaluating the transform as before yields

$$\begin{split} L_{\rm div}(x|\omega) &= -\frac{1}{4\pi} \left\{ \frac{1}{\epsilon^3} + \frac{1}{2\epsilon} \left[-3u + 2\Re_{0(2)}^{\Box-V} + \Re_{1(0)}^{\Box-V} - \mu^2 \right] \\ &+ \frac{1}{3}\mu^3 - \frac{1}{2}\mu \left(-3u + 2\Re_{0(2)}^{\Box-V} + \Re_{1(0)}^{\Box-V} \right) \\ &+ \frac{5u\omega^2 e^{4\phi}}{2\mu} + \frac{m^2 u\omega^2 e^{4\phi}}{2\mu^3} - \frac{15m^6 v}{2\mu^7} + \frac{105m^8 u^2}{8\mu^9} \\ &+ \Re_{0(2)}^{\Box-V} \left[-\frac{\omega^2 e^{4\phi}}{\mu} - \frac{15m^6 u}{2\mu^7} \right] + \Re_{0(4)}^{\Box-V} \left[\frac{3m^4}{\mu^5} \right] \\ &+ \Re_{1(0)}^{\Box-V} \left[-\frac{3m^4 u}{4\mu^5} \right] + \Re_{1(2)}^{\Box-V} \left[\frac{m^2}{2\mu^3} \right] + \Re_{2(0)}^{\Box-V} \left[\frac{1}{4\mu} \right] + O(\epsilon) \right\} . (5.67) \end{split}$$

Meanwhile, for the three-dimensional theory with the field operator \mathcal{F}_{ω} given by (5.41) we have

$$\mathcal{L}_{\omega|\text{div}}(x) = -\frac{1}{2} \int_0^\infty \frac{ds}{s} \frac{1}{(4\pi s)^{\frac{3}{2}}} \exp\left\{-m^2 s - \frac{\epsilon^2}{4s}\right\} \left[1 + s\left(\frac{1}{6}R - V_{\omega}\right)\right]$$

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

$$= -\frac{1}{4\pi} \left[\frac{1}{\epsilon^3} + \frac{1}{2\epsilon} \left(\frac{1}{6} R - V_\omega - m^2 \right) + \frac{m^3}{3} - \frac{m}{2} \left(\frac{1}{6} R - V_\omega \right) \right] + O(\epsilon) .$$
(5.68)

Here V_{ω} is the effective potential of the three-dimensional system, given by (5.42), and R is the scalar curvature calculated from the three-metric h_{ab} . It is related to the four-dimensional curvature ${}^{4}R$ via

$${}^{4}R[g] = R[h] + 4\Box\phi = R[h] + 4\Delta_{h}\phi - 8(\nabla\phi)^{2}.$$
(5.69)

The renormalized effective Lagrangians in four and three dimensions are obtained by subtracting from the exact effective action its divergent part, as given by (5.66) and (5.68) respectively. By comparing these divergent parts using (5.38), (5.39) and (5.67) we can write

$$L_{\rm ren}(x) = e^{2\phi} \int_{-\infty}^{\infty} \frac{d\omega}{2\pi} \left[\mathcal{L}_{\omega|\rm ren}(x) + \Delta \mathcal{L}_{\omega}(x) \right] , \qquad (5.70)$$

where $\Delta \mathcal{L}_{\omega}$ is the term representing the dimensional-reduction anomaly of the renormalized effective Lagrangian,

$$\Delta \mathcal{L}_{\omega} = \frac{1}{4\pi} \left\{ \frac{1}{3} (\mu^{3} - m^{3}) - \frac{1}{2} (\mu - m) \left(\frac{1}{6} R - V - (\nabla \phi)^{2} + \Delta_{h} \phi \right) - \frac{1}{2} m \omega^{2} e^{4\phi} \right. \\ \left. + \mu \left(\frac{5}{2} u - \Re_{0(2)}^{\Box - V} \right) + \frac{1}{\mu} \left(-2m^{2}u + m^{2} \Re_{0(2)}^{\Box - V} + \frac{1}{4} \Re_{2(0)}^{\Box - V} \right) \right. \\ \left. + \frac{m^{2}}{\mu^{3}} \left(-\frac{m^{2}}{2} u + \frac{1}{2} \Re_{1(2)}^{\Box - V} \right) + \frac{m^{4}}{\mu^{5}} \left(3 \Re_{0(4)}^{\Box - V} - \frac{3}{4} u \Re_{1(0)}^{\Box - V} \right) \right. \\ \left. + \frac{m^{6}}{\mu^{7}} \left(-\frac{15}{2} v - \frac{15}{2} u \Re_{0(2)}^{\Box - V} \right) + \frac{m^{8}}{\mu^{9}} \left(\frac{105}{8} u^{2} \right) \right\}.$$
(5.71)

5.3.2 (2+2) Reduction

As we already mentioned, the (2 + 2) reduction is a special case of the "static" spacetime reduction. The calculation of the dimensional-reduction anomaly is very similar to the calculations of the previous subsection – straightforward but

quite involved. We do not reproduce the details of these calculations here but simply give the final results.

The mode decomposition of the renormalized effective Lagrangian for the operator $F = \Box - m^2 - V$ has the form

$$L_{\rm ren} = e^{2\phi} \int \frac{d\mathbf{p}}{(2\pi)^2} \left[\mathcal{L}_{p|\rm ren} + \Delta \mathcal{L}_p \right] , \qquad (5.72)$$

where the dimensional-reduction anomaly $\Delta \mathcal{L}_p$ is

$$\Delta \mathcal{L}_{p} = = \frac{1}{8\pi} \left[-p^{2} e^{2\phi} + \left(\frac{1}{6} R - V + \Delta \phi - (\nabla \phi)^{2} - m^{2} - p^{2} e^{2\phi} \right) \ln \left(\frac{m^{2}}{\mu^{2}} \right) + \left(12u - 4\Re_{0(2)}^{\Box - V} \right) + \frac{1}{\mu^{2}} \left(\Re_{2(0)}^{\Box - V} - 8m^{2}u + 4m^{2} \Re_{0(2)}^{\Box - V} \right) + \frac{m^{2}}{\mu^{4}} \left(-4m^{2}u + 4\Re_{1(2)}^{\Box - V} \right) + \frac{m^{4}}{\mu^{6}} \left(-8u \Re_{1(0)}^{\Box - V} + 32 \Re_{0(4)}^{\Box - V} \right) + \frac{m^{6}}{\mu^{8}} \left(-96v - 96u \Re_{0(2)}^{\Box - V} \right) + \frac{m^{8}}{\mu^{10}} \left(192u^{2} \right) \right].$$
(5.73)

Recall that μ is given by relation (5.57).

In the next chapter we will use these anomalies to derive analytic approximations for $\langle \hat{\Phi}^2 \rangle$ and W at zero temperature. First, however, we will consider an approach to mode decompositions in static spaces with arbitrary periodicity β in an attempt to unify the Rindler-space results of Section 5.1 with these zero-temperature anomalies.

5.4 The Dimensional-Reduction Anomaly in $\langle \hat{\Phi}^2 \rangle$ at Finite Temperature

In this section we determine the dimensional-reduction anomaly in $\langle \hat{\Phi}^2 \rangle$ for the decomposition of a scalar field in a static space where the Euclidean time coordinate has an arbitrary period β . In the context of thermal quantum field theory, this corresponds to a field at temperature $T = \beta^{-1}$. Although we consider only the anomaly in $\langle \hat{\Phi}^2 \rangle$, the techniques used here are readily applied to the calculation of the anomaly in W.

Consider the line element (5.20),

$$ds^{2} = e^{-4\phi(x^{c})}dt^{2} + h_{ab}(x^{c})dx^{a}dx^{b}, \qquad (5.74)$$

where we now choose $t \in [0, \beta]$ with t = 0 and $t = \beta$ identified. Rindler space (5.1) corresponds to $\beta = 2\pi$, $e^{-2\phi} = x^3 = z$ with $h_{ab} = \delta_{ab}$, while the zero-temperature case (5.20) corresponds to $\beta \to \infty$.

Following the steps of the previous sections, we split the points in the t-direction only, and expand all geometric quantities in powers of $(t - t')^2$. Since the internal space of (5.74) is periodic, as for the spherical decompositions, we follow the technique of Section 4.1 and make our expansions periodic as well. Guided by the Rindler space example, in which the t-dependence of σ was in the form $(1 - \cos(t - t'))$ with $t \in [0, 2\pi]$, we define an effective angular coordinate λ by

$$\lambda \equiv 2\pi \frac{(t-t')}{\beta} \in [0, 2\pi].$$
(5.75)

We then replace all $(t - t')^2$ by

$$(t-t')^{2} = \left(\frac{\beta}{2\pi}\right)^{2} \left[2(1-\cos\lambda) + \frac{1}{3}(1-\cos\lambda)^{2} + \frac{4}{45}(1-\cos\lambda)^{3} + \cdots\right],$$
(5.76)

truncating at the lowest order that gives the correct flat-space limit. With this technique our expansions (5.30)–(5.32) for σ and the $\Re_n^{\Box-V}$ become

$$2\sigma = 2L^{2}(1-\cos\lambda) + L^{2}\left[\frac{1}{3} - \frac{4}{3}(\nabla\phi)^{2}L^{2}\right](1-\cos\lambda)^{2}, \quad (5.77)$$

$$\Re_0^{\Box - V} = 1 + \frac{L^2}{3} \Box \phi \left(1 - \cos \lambda \right), \qquad (5.78)$$

$$\Re_1^{\Box-V} = \frac{1}{6} {}^4R - V, \qquad (5.79)$$

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

where for convenience we define

$$\mathbf{L} \equiv \frac{\beta \mathrm{e}^{-2\phi}}{2\pi} \,, \tag{5.80}$$

which is the effective radius of a line of constant t. The local temperature is then given by $T_{loc} = (2\pi L)^{-1}$. It is easily verified that (5.77)-(5.79) reduce to the previously derived results for both Rindler space and zero-temperature static space.

The mode decomposition is similar to that for the Rindler-space example:

$$K(t, x; t', x'|s) = \frac{e^{\phi + \phi'}}{\beta} \sum_{n = -\infty}^{\infty} e^{-in\frac{2\pi}{\beta}(t - t')} K_n(x, x'|s), \qquad (5.81)$$

$$K_n(x,x'|s) = e^{-(\phi+\phi')} \int_0^\beta d(t-t') e^{in\frac{2\pi}{\beta}(t-t')} K(t,x;t',x'|s), \qquad (5.82)$$

where we use

$$Y_{\lambda\varrho}(y) \to \frac{e^{-in\frac{2\pi}{\beta}t}}{\sqrt{\beta}}, \qquad \lambda \to \left(n\frac{2\pi}{\beta}\right)^2, \qquad \sum_{\lambda,\varrho} \to \sum_{n=-\infty}^{\infty}.$$
 (5.83)

The renormalized values of the square of the field operator will then be related via

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = \frac{e^{2\phi}}{\beta} \sum_{n=-\infty}^{\infty} \left[\langle \hat{\varphi}_n^2 \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_n^2 \rangle \right] \,. \tag{5.84}$$

Note also that we can return to the zero-temperature $(\beta \rightarrow \infty)$ formalism with the replacements

$$n\frac{2\pi}{\beta} \to \omega$$
, $\frac{1}{\beta} \sum_{n=-\infty}^{\infty} \to \frac{1}{2\pi} \int_{-\infty}^{\infty} d\omega$, (5.85)

$$\int_0^\beta d(t-t') \to \int_{-\infty}^\infty d(t-t') \,. \tag{5.86}$$

By inserting the expansions (5.77)-(5.79) into the expression (4.12) for the divergent part of the heat kernel in four dimensions and performing the mode

decomposition (5.82), we obtain

$$K_{div|n}(x,x|s) = L \frac{e^{-m^2 s}}{(4\pi s)^2} \left[J_{n0} + s \left(\frac{1}{6} {}^4 R - V \right) J_{n0} + \frac{L^2}{3} \Box \phi J_{n1} - \frac{L^2}{12s} (1 - 4(\nabla \phi)^2 L^2) J_{n2} \right], \qquad (5.87)$$

where

$$J_{nq} \equiv \int_0^{2\pi} d\lambda \,\mathrm{e}^{in\lambda} \,\mathrm{e}^{p(1-\cos\lambda)} (1-\cos\lambda)^q \,, \tag{5.88}$$

and $p \equiv -L^2/2s$. These integrals are evaluated by taking derivatives with respect to p of the q = 0 result (5.9) used in the mode decomposition of Rindler space:

$$J_{n0} = 2\pi (-1)^n e^p I_n(p) , \qquad (5.89)$$

$$J_{nq} = \frac{d^{q}}{dp^{q}} J_{n0} \,. \tag{5.90}$$

Equation (5.87) is to be compared to the subtraction term for the dimensionally reduced theory, consisting of the first term of the Schwinger-DeWitt expansion (5.43) in three dimensions:

$$\mathcal{K}_{n|\text{div}} = \frac{e^{-m^2 s}}{(4\pi s)^{\frac{3}{2}}} \,. \tag{5.91}$$

The anomaly in $\langle \hat{\Phi}^2 \rangle$ can now be obtained by integrating the difference of (5.91) and (5.87) over s:

$$\Delta \langle \hat{\varphi}_n^2 \rangle = \int_0^\infty ds \left[\mathcal{K}_{n|\operatorname{div}}(x, x|s) - K_{\operatorname{div}|n}(x, x|s) \right] \,. \tag{5.92}$$

Employing the formula (5.10) from the Rindler example, one can show with much tedious algebra that the anomaly is given by

$$\Delta \langle \hat{\varphi}_{n}^{2} \rangle = \frac{m}{4\pi} \left[-1 + n \left(I_{n}(mL) K_{n+1}(mL) - I_{n+1}(mL) K_{n}(mL) \right) + mL \left(I_{n}(mL) K_{n}(mL) + I_{n+1}(mL) K_{n+1}(mL) \right) \right] - \frac{L}{4\pi} \left(\frac{1}{6} {}^{4}R - V \right) I_{n}(mL) K_{n}(mL)$$

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

$$+ \frac{L}{6\pi} \Box \phi \left[nI_n(mL)K_n(mL) - \frac{1}{2}mL \left(I_n(mL)K_{n+1}(mL) - I_{n+1}(mL)K_n(mL) \right) \right] \\ + \frac{1}{12\pi L} (1 - 4(\nabla \phi)^2 L^2) \left[(n^2 - n)I_n(mL)K_n(mL) + \frac{n - 1}{2}mL \left(I_{n+1}(mL)K_n(mL) - I_n(mL)K_{n+1}(mL) \right) + \frac{1}{2} (mL)^2 \left(I_n(mL)K_n(mL) - I_{n+1}(mL)K_{n+1}(mL) \right) \right] .$$
(5.93)

Using uniform asymptotic expansions of the Bessel functions, one can verify that in the L $\rightarrow \infty$ ($\beta \rightarrow \infty$, $T \rightarrow 0$) limit (5.93) reproduces the zero-temperature result (5.49).⁴ As well, the special case of $\beta = 2\pi$ with line element (5.1) reproduces the exact Rindler-space anomaly, (5.19).

In the following chapter we turn our attention to making "practical" use of these dimensional-reduction anomaly results. First, however, a few comments regarding the dimensional-reduction anomaly and the choice of quantum state are in order. The anomaly is determined by the divergent subtraction terms of the D- and (D - n)-dimensional theories, and the mode decomposition. The divergences are local, depending on the geometry and potential, and so are state-independent. The mode decomposition, however, is sensitive to the global behaviour of the internal space (at least for low frequencies), and so may be affected by the choice of state. In particular, if the time direction lies in the internal space, the mode decomposition will be sensitive to the temperature; this means that the dimensional-reduction anomaly for static spaces will be state-dependent. By contrast, for the spherical reductions considered previously, the decomposition was over spatial directions, and so the dimensional-reduction anomalies of Chapters 3 and 4 are state-independent.

⁴In taking the zero-temperature limit it is important to hold the frequency fixed. One does this by replacing n by $\omega\beta/2\pi$ and holding ω constant as $\beta \to \infty$.

In either case, it is important to recognize the distinction between the low- and high-frequency parts of the dimensional-reduction anomaly. The high-frequency part of the anomaly is responsible for the divergences in the D-dimensional theory. It is purely local, and so is universal and unambiguous. By contrast, the low-frequency part of the anomaly is sensitive to the global behaviour of the system. As a result, it is inherently somewhat ambiguous in the sense that different assumptions for the large-distance behaviour of the system will alter the anomaly at low frequencies. In particular, the assumption of periodicity used in the spherical decompositions of Chapter 4 and in the static decompositions of this section do influence the anomaly. While they are perhaps the most natural choices for determining the anomaly for compact internal spaces, they are not unique.

Chapter 6

Analytic Approximations for $\langle \hat{\Phi}^2 angle$ and W

In the preceding chapter we examined mode decompositions and the dimensionalreduction anomaly in static spaces. In this chapter we will see how these results may be used to derive a new type of analytic approximation for renormalized expectation values in static spaces. At the present time, this scheme has only been applied to zero-temperature fields; its generalization to fields at arbitrary temperature is still underway.

6.1 Analytic Approximation for $\langle \hat{\Phi}^2 \rangle$

In Section 5.2.1 we derived the dimensional-reduction anomaly in $\langle \hat{\Phi}^2 \rangle$ for a field at zero-temperature in the static space (5.20),

$$\Delta \langle \hat{\varphi}_{\omega}^{2} \rangle = \frac{1}{4\pi} \left[\mu - m - \frac{1}{2\mu} \left(\frac{1}{6} {}^{4}R - V \right) - \frac{m^{2} \Box \phi}{6\mu^{3}} - \frac{m^{4} (\nabla \phi)^{2}}{2\mu^{5}} \right], \quad (6.1)$$

where the four- and three-dimensional theories are related via

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = e^{2\phi} \int_{-\infty}^{\infty} \frac{d\omega}{2\pi} \left[\langle \hat{\varphi}_{\omega}^2 \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_{\omega}^2 \rangle \right] \,. \tag{6.2}$$

The anomalous term $\Delta \langle \hat{\varphi}^2_{\omega} \rangle$ is finite and can be written in the form

$$\Delta \langle \hat{\varphi}_{\omega}^{2} \rangle = \Delta \langle \hat{\varphi}_{\omega}^{2} \rangle^{\sharp} + \Delta \langle \hat{\varphi}_{\omega}^{2} \rangle^{\flat} , \qquad (6.3)$$

where¹

$$\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\sharp} = \frac{1}{4\pi} \left[\bar{\omega} + \frac{m^2}{2\bar{\omega}} - m - \frac{1}{2\bar{\omega}} \left(\frac{1}{6} {}^4R - V \right) \right], \qquad (6.4)$$

$$\Delta \langle \hat{\varphi}_{\omega}^{2} \rangle^{\flat} = \frac{1}{4\pi} \left[\mu - \bar{\omega} - \frac{m^{2}}{2\bar{\omega}} - \frac{1}{2} \left(\frac{1}{6} {}^{4}R - V \right) \left(\frac{1}{\mu} - \frac{1}{\bar{\omega}} \right) - \frac{m^{2} \Box \phi}{6\mu^{3}} - \frac{m^{4} (\nabla \phi)^{2}}{2\mu^{5}} \right]$$
(6.5)

Here and later we use the notation $\bar{\omega} = e^{2\phi}\omega$. The quantity $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\sharp}$ is that part of the anomaly which dominates at high frequencies $(\omega \to \infty)$; it consists of all terms of $O(\omega^{-1})$ and higher in the large- ω expansion of $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle$. These are the terms which diverge in the integration over ω , and hence which lead to the divergences in the four-dimensional Green function as $t - t' \to 0$. The part of the anomaly that remains when $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\sharp}$ is subtracted off is denoted by $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\flat}$. It is of $O(\omega^{-2})$ for high frequencies, so the inverse Fourier transform of $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\flat}$ is finite as $t - t' \to 0$.

Let us examine the inverse Fourier transform of $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\flat}$. Defining

$$\langle \hat{\Phi}^2 \rangle_{\text{approx}} = \int_0^\infty \frac{d\bar{\omega}}{\pi} \Delta \langle \hat{\varphi}_{\omega}^2 \rangle^\flat , \qquad (6.6)$$

and performing the integration (see Appendix C.3.2), we obtain

$$\langle \hat{\Phi}^2 \rangle_{\text{approx}} = \frac{m^2}{16\pi^2} + \frac{1}{16\pi^2} \left(\frac{1}{6} {}^4R - V - m^2 \right) \ln \frac{m^2 \mathrm{e}^{-4\phi}}{4\eta^2} - \frac{\Box \phi}{24\pi^2} - \frac{(\nabla \phi)^2}{12\pi^2} \,. \tag{6.7}$$

The parameter η is a low-frequency cut-off which is required to make the integral convergent. It corresponds to a well-known ambiguity in the renormalization prescription. This ambiguity is absent for a conformally invariant theory, when m = 0 and $V = \frac{1}{6} {}^{4}R$.

¹The symbol # (*sharp*) is borrowed from musical notation to denote the *high-frequency* part of the anomaly; the remainder is labelled with \flat (*flat*).

CHAPTER 6. ANALYTIC APPROXIMATIONS FOR $\langle \hat{\Phi}^2 \rangle$ AND W 116

The reason that (6.7) is of interest (and the justification for the name $\langle \hat{\Phi}^2 \rangle_{approx}$) is that in certain special cases it reproduces previously known approximation schemes for quantum fields in curved spacetimes. In particular, for a massless, conformally coupled field (6.7) reduces to the zero-temperature limit of the Killing approximation of Frolov and Zelnikov [10] for static spacetimes. This scheme is based on the assumption that one can approximate the expectation value of interest $\langle \hat{\Phi}^2 \rangle$ or $\langle \hat{T}_{\mu\nu} \rangle$ by a tensor which is a local function of the curvatures, the Killing vector for the static spacetime, and their covariant derivatives. Imposing physical properties, such as the covariant conservation of $\langle \hat{T}_{\mu\nu} \rangle$, then uniquely determines the approximate tensor up to a few arbitrary constants which depend on the choice of state. For the further specialization to static Einstein spaces the Killing approximation contains the approximations of Page [8], and Page, Brown, and Ottewill [9], and so our result (6.7) is equivalent to the zero-temperature limit of these approximations as well. Finally, for the special case of static spherically symmetric spacetimes (6.7) also reduces to the zero-temperature limit of the WKB approximation of Anderson [11]. (The corresponding result for the stress tensor, derived by Anderson, Hiscock, and Samuel [12], was discussed in Section 4.4, where it was denoted $\langle \hat{T}_{\mu\nu} \rangle_{AHS}$.)

The agreement between our $\langle \hat{\Phi}^2 \rangle_{approx}$ and the results of Anderson is particularly interesting when we recall that (6.7) is constructed using the low-frequency part of the dimensional-reduction anomaly, $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\flat}$. We are thus led to the seemingly paradoxical conclusion that there is a connection between the low-frequency part of the anomaly and the high-frequency behaviour of the field.

A more careful examination of how $\langle \hat{\Phi}^2 \rangle_{approx}$ is constructed resolves this mystery. Note that the renormalization terms $\mathcal{G}_{\omega|div}$ (5.44) from the dimensionally reduced theory do not contribute² to $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\flat}$. The analytic approximation is

²This is because of their dependence on the frequency, ω . Specifically, the renormalization terms from the dimensionally reduced theory are polynomials in ω^2 [ω only enters $\mathcal{K}_{\omega|div}$

therefore constructed using only the renormalization terms $G_{div|\omega}$ (5.40) from four dimensions. Schematically, we have

$$\langle \hat{\Phi}^2(x) \rangle_{\text{approx}} = \int_0^\infty \frac{d\bar{\omega}}{\pi} \left[\Delta \langle \hat{\varphi}^2_{\omega}(x) \rangle - \Delta \langle \hat{\varphi}^2_{\omega}(x) \rangle^{\sharp} \right]$$

=
$$\int_0^\infty \frac{d\bar{\omega}}{\pi} \left[G^{\sharp}(x;x|\omega) - G_{\text{div}}(x;x|\omega) \right]$$
(6.8)

$$= \lim_{t'\to t} \left[G^{\sharp}(t,x;t',x) - G_{div}(t,x;t',x) \right] .$$
 (6.9)

Here $G^{\sharp}(t, x; t', x)$ denotes the high-frequency part of the Green function, obtained by Fourier transforming the Green function, keeping only the terms of $O(\omega^{-1})$ and higher in the large- ω limit, and then inverting the Fourier transform. Viewed in this manner, we see that the analytic approximation is simply the expression resulting from using a high-frequency approximation for the Green function, renormalized in the standard manner in four dimensions (using (5.36)). This is why our technique reproduces the WKB approximation of Anderson.

There are several points to note regarding our approximation. First, it is unambiguous since it is constructed using the high-frequency behaviour of the Green function, which is determined by the ultraviolet divergences of the theory. In particular, G^{\sharp} is completely independent of the particular ansatz used for the Green function for large point separations in the Fourier transform. Different choices change only the low-frequency behaviour of the Green function, which is discarded anyway in the high-frequency limit. This means that we can construct G^{\sharp} using any ansatz for the Green function which contains all of its divergences for the given point-splitting; the Schwinger-DeWitt expansion is merely the most convenient and general such ansatz.

Similarly, the low-frequency part of G_{div} used for renormalization in four dimensions is also unimportant, as it is not affected by the subtraction in (6.8) and through the induced potential V_{ω} , given by (5.42)]. Thus, they are unaltered in the highfrequency expansions and so are removed identically when we subtract the high-frequency limit, $\Delta \langle \hat{\varphi}_{\omega}^2 \rangle^{\sharp}$. is just returned identically in four dimensions when the mode sum is done. For example, note that the terms in (5.36) which are finite as $\tau \to 0$ all appear identically in the analytic approximation (6.7) [except for a sign reversal, since (5.36) is subtracted from G^{\sharp} in the renormalization]. As a result, these finite terms need not be mode-decomposed in the first place. We could just as easily have begun with the Laurent series (5.36) for G, and only Fourier-transformed those terms which diverge as $t \to 0$. Removing the high-frequency limit and performing the inverse Fourier transform would then yield our analytic approximation (6.7).

The only knowledge used to construct $\langle \hat{\Phi}^2 \rangle_{approx}$ is the structure of the divergences of the physical theory and the static nature of the space. Since the divergences are known for arbitrary spaces and field parameters via the Schwinger-DeWitt expansion, this method can (in theory) be applied to any spacetime with a continuous symmetry. This is a distinct advantage over the WKB approximations used by Anderson, Hiscock, and Samuel and by Balbinot, Fabbri, and Nicolini, which require explicit mode solutions for the entire field $\hat{\Phi}$, not just for the temporal behaviour. Our approach does not presume detailed knowledge of the spacetime; in particular, we do not need to assume any symmetries other than being static. Furthermore, since our approximation is constructed (in the massless case) using only covariant objects from the static space (i.e., the curvature, the Killing vector, and their covariant derivatives), it is clear that it must reduce to the Killing approximation of Frolov and Zelnikov and the approximations of Page, Brown, and Ottewill for massless conformally coupled fields. Our high-frequency approximation is not restricted to conformal fields, however, and hence relation (6.7) can be considered as an extension of these approximations to the general case when the spacetime is static, but not necessary spherically symmetric, and the field equation includes an arbitrary mass and potential V. This approximation should be valid whenever the high-frequency limit of the Green function is appropriate for modelling the full Green function.

Finally, we note that our approximation for $\langle \hat{\Phi}^2 \rangle$ is the same if we were to use the anomaly from the (2 + 2) reduction in Section 5.2.2. In this case, the square of the field operators for a (2+2) decomposition of the space (5.50) were shown to be related by (5.61),

$$\langle \hat{\Phi}^2 \rangle_{\rm ren} = e^{2\phi} \int \frac{d\mathbf{p}}{(2\pi)^2} \left[\langle \hat{\varphi}_p^2 \rangle_{\rm ren} + \Delta \langle \hat{\varphi}_p^2 \rangle \right] , \qquad (6.10)$$

where the dimensional-reduction anomaly $\Delta \langle \hat{\varphi}_p^2 \rangle$ is

$$\Delta \langle \hat{\varphi}_{p}^{2} \rangle = \frac{1}{4\pi} \left[\ln \left(\frac{\mu^{2}}{m^{2}} \right) - \frac{1}{\mu^{2}} \left(\frac{1}{6} {}^{4}R - V \right) - \frac{m^{2}}{3\mu^{4}} \Box \phi - \frac{2m^{4}}{3\mu^{6}} (\nabla \phi)^{2} \right] .$$
(6.11)

The part of $\Delta \langle \hat{\varphi}_p^2 \rangle$ which dominates at large "momentum" p and which is responsible for the divergences of the four-dimensional Green function in the coincidence limit $\mathbf{t} - \mathbf{t}' \to 0$ is

$$\Delta \langle \hat{\varphi}_{p}^{2} \rangle^{\sharp} = \frac{1}{4\pi} \left[\ln \frac{\bar{p}^{2}}{m^{2}} - \frac{1}{\bar{p}^{2}} \left(\frac{1}{6} {}^{4}R - V - m^{2} \right) \right] , \qquad (6.12)$$

where we define $\bar{p} = p e^{\phi}$. Defining the sub-leading part $\Delta \langle \hat{\varphi}_p^2 \rangle^{\flat}$ of the anomaly and $\langle \hat{\Phi}^2 \rangle_{approx}$ by relations similar to (6.3) and (6.6), respectively, we obtain an expression for $\langle \hat{\Phi}^2 \rangle_{approx}$ which is identical to (6.7); see Appendix C.3.3. One can expect this result, since the (2+2) reduction may be considered as a special case of the (1+3) reduction.

Thus confident as to the physical meaning of our new approximation scheme, let us apply it to the effective action and the stress tensor.

6.2 Analytic Approximation for the Effective Action

Our approximation procedure is easily applied to the effective action. From (5.70), the effective Lagrangians in four and three dimensions are related by

$$L_{\rm ren}(x) = e^{2\phi} \int_{-\infty}^{\infty} \frac{d\omega}{2\pi} \left[\mathcal{L}_{\omega|\rm ren}(x) + \Delta \mathcal{L}_{\omega}(x) \right] , \qquad (6.13)$$

where $\Delta \mathcal{L}_{\omega}$ is the dimensional-reduction anomaly (5.71),

$$\Delta \mathcal{L}_{\omega} = \frac{1}{4\pi} \left\{ \frac{1}{3} (\mu^{3} - m^{3}) - \frac{1}{2} (\mu - m) \left(\frac{1}{6} R - V - (\nabla \phi)^{2} + \Delta_{h} \phi \right) - \frac{1}{2} m \bar{\omega}^{2} + \mu \left(\frac{5}{2} u - \Re_{0(2)}^{\Box - V} \right) + \frac{1}{\mu} \left(-2m^{2}u + m^{2} \Re_{0(2)}^{\Box - V} + \frac{1}{4} \Re_{2(0)}^{\Box - V} \right) + \frac{m^{2}}{\mu^{3}} \left(-\frac{m^{2}}{2} u + \frac{1}{2} \Re_{1(2)}^{\Box - V} \right) + \frac{m^{4}}{\mu^{5}} \left(3 \Re_{0(4)}^{\Box - V} - \frac{3}{4} u \Re_{1(0)}^{\Box - V} \right) + \frac{m^{6}}{\mu^{7}} \left(-\frac{15}{2} v - \frac{15}{2} u \Re_{0(2)}^{\Box - V} \right) + \frac{m^{8}}{\mu^{9}} \left(\frac{105}{8} u^{2} \right) \right\}.$$
 (6.14)

As earlier, we write

$$\Delta \mathcal{L}_{\omega} = \Delta \mathcal{L}_{\omega}^{\sharp} + \Delta \mathcal{L}_{\omega}^{\flat} , \qquad (6.15)$$

where $\Delta \mathcal{L}_{\omega}^{\sharp}$ is the part of the anomaly which dominates at high frequencies $(\omega \to \infty)$,

$$\Delta \mathcal{L}_{\omega}^{\sharp} = \frac{1}{4\pi} \left\{ \frac{1}{3} \bar{\omega}^{3} - \frac{1}{2} m \bar{\omega}^{2} + \bar{\omega} \left[\frac{m^{2}}{2} - \frac{1}{2} \left(\frac{1}{6} R - V - (\nabla \phi)^{2} + \Delta_{h} \phi \right) \right. \\ \left. + \frac{5u}{2} - \Re_{0(2)}^{\Box - V} \right] + \left[\frac{m}{2} \left(\frac{1}{6} R - V - (\nabla \phi)^{2} + \Delta_{h} \phi \right) - \frac{m^{3}}{3} \right] \\ \left. + \frac{1}{\bar{\omega}} \left[\frac{3m^{4}}{8} - \frac{m^{2}}{4} \left(\frac{1}{6} R - V - (\nabla \phi)^{2} + \Delta_{h} \phi + 3u - 2 \Re_{0(2)}^{\Box - V} \right) \right. \\ \left. + \frac{1}{4} \Re_{2(0)}^{\Box - V} \right] \right\}.$$
(6.16)

By subtracting this large- ω limit from the anomaly (6.14) and making the inverse Fourier transform, we can construct an approximate effective Lagrangian for the four-dimensional theory:

$$L_{\rm approx} = \int_0^\infty \frac{d\bar{\omega}}{\pi} \Delta \mathcal{L}_{\omega}^{\flat} \,. \tag{6.17}$$

Performing the ω -integration (see Appendix C.3.2) gives

$$L_{\text{approx}} = \frac{3m^4}{128\pi^2} + \frac{m^2}{32\pi^2} \left[-\frac{1}{6}R + V + (\nabla\phi)^2 - \Delta_h\phi + u - 2\Re_{0(2)}^{\Box-V} \right] + \frac{1}{8\pi^2} \left[\Re_{1(2)}^{\Box-V} + 4\Re_{0(4)}^{\Box-V} - u\Re_{1(0)}^{\Box-V} - 8u\Re_{0(2)}^{\Box-V} - 8v + 12u^2 \right] + \frac{1}{32\pi^2} \left[-\frac{m^4}{2} + m^2 \left(\frac{1}{6}R - V - (\nabla\phi)^2 + \Delta_h\phi + 3u - 2\Re_{0(2)}^{\Box-V} \right) - \Re_{2(0)}^{\Box-V} \right] \ln \left(\frac{m^2 e^{-4\phi}}{4\eta^2} \right).$$
(6.18)

The effective action corresponding to this Lagrangian may be simplified considerably using integration by parts. Substituting for the u, v, and $\Re_{n(k)}^{\Box-V}$ from Appendix C.1 and neglecting surface terms, one can show that the effective action for the important case $V = \xi^4 R$ may be written as

$$W_{approx} = \int d^{4}x \sqrt{g} \left\{ -\frac{1}{64\pi^{2}} \ln\left(\frac{m^{2}e^{-4\phi}}{4\eta^{2}}\right) \left[m^{4} + \frac{1}{90} \left({}^{4}R_{\alpha\beta\gamma\delta} {}^{4}R^{\alpha\beta\gamma\delta} - {}^{4}R_{\alpha\beta} {}^{4}R^{\alpha\beta\gamma\delta} + \Box {}^{4}R\right) \right] + \frac{3m^{4}}{128\pi^{2}} - \frac{m^{2}(\nabla\phi)^{2}}{24\pi^{2}} + \frac{1}{360\pi^{2}} \left[{}^{4}R_{\alpha\beta} \phi^{\alpha} \phi^{\beta} - \frac{3}{2} (\Box\phi)^{2} - 4\Box\phi(\nabla\phi)^{2} - 4(\nabla\phi)^{4} \right] + \frac{(\xi - \frac{1}{6})}{32\pi^{2}} \left[m^{2} {}^{4}R - \frac{4}{3} {}^{4}R(\nabla\phi)^{2} + \ln\left(\frac{m^{2}e^{-4\phi}}{4\eta^{2}}\right) \left(-m^{2} {}^{4}R + \frac{1}{6}\Box {}^{4}R\right) \right] - \frac{(\xi - \frac{1}{6})^{2}}{64\pi^{2}} \left[\ln\left(\frac{m^{2}e^{-4\phi}}{4\eta^{2}}\right) ({}^{4}R)^{2} \right] \right\}.$$
(6.19)

Note that (6.19) has been written entirely in terms of four-dimensional quantities. The stress tensor $\langle \hat{T}_{\mu\nu} \rangle_{approx}$ resulting from the variation of this action with respect to the metric is displayed in Section 6.3. It can be shown that in the special case of a static, spherically symmetric spacetime $\langle \hat{T}_{\mu\nu} \rangle_{approx}$ coincides with the analytic approximation of Anderson, Hiscock, and Samuel [12] for the zero-temperature case. Furthermore, in the massless, conformally coupled limit (6.19) coincides with the zero-temperature Killing approximation [10] and the zero-temperature Page-Brown-Ottewill approximation [8, 9].

6.3 Analytic Approximation for $\langle \hat{\mathbf{T}}_{\mu\nu} \rangle$

Using the result (6.19) for the approximate effective action for a scalar field in a static four-dimensional spacetime with potential $V = \xi^4 R$, one can calculate the resulting stress tensor. For convenience, we split the action into pieces according to its dependence on the field mass m and coupling ξ as follows:

$$W_{\text{approx}} = W_{m^4}^0 + W_{m^2}^0 + W_{m^0}^0 + \left(\xi - \frac{1}{6}\right) \left[W_{m^2}^1 + W_{m^0}^1\right] + \left(\xi - \frac{1}{6}\right)^2 W_{m^0}^2; \quad (6.20)$$

where

$$W_{m^{4}}^{0} = \frac{m^{4}}{32\pi^{2}} \int d^{4}x \sqrt{g} \left\{ \frac{3}{4} - \frac{1}{2} \ln\left(\frac{m^{2}\chi^{2}}{4\eta^{2}}\right) \right\}, \qquad (6.21)$$

$$W_{m^2}^0 = \frac{m^2}{32\pi^2} \int d^4x \sqrt{g} \left\{ -\frac{4}{3} (\nabla \phi)^2 \right\} , \qquad (6.22)$$

$$W_{m^{0}}^{0} = \frac{1}{720\pi^{2}} \int d^{4}x \sqrt{g} \left\{ \left[2R_{\alpha\beta}\phi^{\alpha}\phi^{\beta} - 3(\Box\phi)^{2} - 8\Box\phi(\nabla\phi)^{2} - 8(\nabla\phi)^{4} \right] - \frac{1}{8} \ln\left(\frac{m^{2}\chi^{2}}{4\eta^{2}}\right) \left(R_{\alpha\beta\gamma\delta}R^{\alpha\beta\gamma\delta} - R_{\alpha\beta}R_{\alpha\beta} + \Box R \right) \right\}, \quad (6.23)$$

$$W_{m^2}^1 = \frac{m^2}{32\pi^2} \int d^4x \sqrt{g} \left\{ R - R \ln\left(\frac{m^2\chi^2}{4\eta^2}\right) \right\}, \qquad (6.24)$$

$$W_{m^0}^1 = \frac{1}{32\pi^2} \int d^4x \sqrt{g} \left\{ -\frac{4}{3} R (\nabla \phi)^2 + \frac{1}{6} \ln \left(\frac{m^2 \chi^2}{4\eta^2} \right) \Box R \right\} , \qquad (6.25)$$

$$W_{m^0}^2 = \frac{1}{32\pi^2} \int d^4x \sqrt{g} \left\{ -\frac{1}{2} R^2 \ln\left(\frac{m^2 \chi^2}{4\eta^2}\right) \right\} .$$
 (6.26)

Here and for the remainder of this section we assume all curvatures and derivatives to be four-dimensional; i.e., calculated using the metric $g_{\mu\nu}$. For example, multiple four-dimensional covariant derivatives of ϕ are represented by $\phi_{\alpha\beta} \equiv \phi_{;\alpha\beta}$, etc. Also, χ^{μ} is the Killing vector of the space (5.20), with $\chi^2 = e^{-4\phi}$. Using

$$\langle \hat{T}_{\mu\nu} \rangle = \frac{2}{\sqrt{g}} \frac{\delta W}{\delta g^{\mu\nu}} , \qquad (6.27)$$

the stress tensor due to each part of the effective action is found to be (in Euclidean signature)

$$\langle \hat{T}_{\mu\nu} \rangle_{m^4}^0 = \frac{m^4}{32\pi^2} \left\{ \frac{\chi_{\mu}\chi_{\nu}}{\chi^2} + g_{\mu\nu} \left[\frac{1}{2} \ln \left(\frac{m^2 e^{-4\phi}}{4\eta^2} \right) - \frac{3}{4} \right] \right\}, \qquad (6.28)$$

$$\langle \hat{T}_{\mu\nu} \rangle_{m^2}^0 = \frac{m^4}{24\pi^2} \left\{ \frac{\chi_{\mu}\chi_{\nu}}{\chi^2} \Box \phi + g_{\mu\nu} (\nabla \phi)^2 - 2\phi_{\mu}\phi_{\nu} \right\}, \qquad (6.29)$$

$$\begin{split} \langle \hat{T}_{\mu\nu} \rangle_{\mathfrak{m}^{0}}^{0} &= -\frac{1}{2880\pi^{2}} \frac{1}{2} \ln \left(\frac{m^{2}\chi^{2}}{4\eta^{2}} \right) \left[12(R_{\mu\alpha\nu\beta} - \frac{1}{4}g_{\mu\nu}R_{\alpha\beta})R^{\alpha\beta} \right. \\ &\left. - 4R(R_{\mu\nu} - \frac{1}{4}g_{\mu\nu}R) + 6\Box R_{\mu\nu} - 2R_{;\mu\nu} - g_{\mu\nu}\Box R \right] \\ &+ \frac{1}{2880\pi^{2}} \frac{\chi_{\mu}\chi_{\nu}}{\chi^{2}} \left[\left(R^{\alpha\beta\gamma\delta}R_{\alpha\beta\gamma\delta} - R^{\alpha\beta}R_{\alpha\beta} + \Box R \right) \right. \\ &\left. + 4\left(-R^{;\alpha}\phi_{\alpha} - 2R^{\alpha\beta}\phi_{\alpha\beta} - 3\Box^{2}\phi - 4\Box((\nabla\phi)^{2}) + 8(\Box\phi)^{2} \right. \\ &\left. + 8\nabla\phi\cdot\nabla(\Box\phi) + 16\Box\phi(\nabla\phi)^{2} + 16\nabla\phi\cdot\nabla((\nabla\phi)^{2}) \right) \right] \\ &+ \frac{1}{720\pi^{2}}g_{\mu\nu} \left[\left. -R^{\alpha\beta}\phi_{\alpha\beta} - \frac{1}{2}R^{;\alpha}\phi_{\alpha} + \Box^{2}\phi - 2R^{\alpha\beta}\phi_{\alpha}\phi_{\beta} \right. \\ &\left. + 2(\phi^{\alpha}\phi^{\beta})_{;\alpha\beta} - 3(\Box\phi)^{2} - 6\nabla\phi\cdot\nabla(\Box\phi) - 8\nabla\phi\cdot\nabla((\nabla\phi)^{2}) \right. \\ &\left. + 8(\nabla\phi)^{4} \right] \\ &+ \frac{1}{720\pi^{2}} \left[4R_{\mu\alpha\nu\beta}\phi^{\alpha\beta} + \frac{1}{2}\left(R_{\alpha\mu}\phi_{\nu}^{\ \alpha} + R_{\alpha\nu}\phi_{\mu}^{\ \alpha} \right) \right. \\ &\left. + 3\left(R_{\mu\nu;\alpha} - R_{\mu\alpha;\nu} - R_{\nu\alpha;\mu} \right)\phi^{\alpha} \right. \\ &\left. - \left(\Box\phi \right)_{;\mu\nu} + 4\left(R_{\alpha\mu}\phi_{\nu} + R_{\alpha\nu}\phi_{\mu} \right)\phi^{\alpha} + 2\Box(\phi_{\mu}\phi_{\nu}) \right. \\ &\left. - 2\left(\left(\phi_{\mu}\phi_{\alpha} \right)_{\nu}^{\ \alpha} + \left(\phi_{\nu}(\nabla\phi)^{2} \right)_{;\mu} \right) - 16(\phi_{\mu}\phi_{\nu})\Box\phi \right. \end{split}$$

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

$$-32(\phi_{\mu}\phi_{\nu})(\nabla\phi)^{2} \bigg] , \qquad (6.30)$$

$$\langle \hat{T}_{\mu\nu} \rangle_{m^{2}}^{1} = \frac{m^{2}}{16\pi^{2}} \left\{ \left(R_{\mu\nu} - \frac{1}{2} g_{\mu\nu} R \right) \left[1 - \ln \left(\frac{m^{2} \chi^{2}}{4\eta^{2}} \right) \right] + \frac{\chi_{\mu} \chi_{\nu}}{\chi^{2}} R + 4 g_{\mu\nu} \Box \phi - 4 \phi_{\mu\nu} \right\},$$

$$(6.31)$$

$$\langle \hat{T}_{\mu\nu} \rangle_{m^{0}}^{1} = \frac{1}{24\pi^{2}} \left\{ \frac{\chi_{\mu}\chi_{\nu}}{\chi^{2}} \left[R^{;\alpha}\phi_{\alpha} + R\Box\phi - \frac{1}{4}\Box R \right] - 2 \left(R_{\mu\nu} - \frac{1}{2}g_{\mu\nu}R \right) (\nabla\phi)^{2} - R_{\mu\nu}\Box\phi - g_{\mu\nu}\Box \left(\Box\phi + 2(\nabla\phi)^{2}\right) + \left(\Box\phi + 2(\nabla\phi)^{2}\right)_{;\mu\nu} - 2R\phi_{\mu}\phi_{\nu} + \frac{1}{2} \left(R_{;\mu}\phi_{\nu} + R_{;\nu}\phi_{\mu} - g_{\mu\nu}R^{;\alpha}\phi_{\alpha} \right) \right\},$$
(6.32)

$$\langle \hat{T}_{\mu\nu} \rangle_{m^{0}}^{2} = \frac{1}{32\pi^{2}} \left\{ \ln\left(\frac{m^{2}\chi^{2}}{4\eta^{2}}\right) \left[\frac{1}{2}g_{\mu\nu}R^{2} + 2R_{;\mu\nu} - 2g_{\mu\nu}\Box R - 2R_{\mu\nu}R\right] + \frac{\chi_{\mu}\chi_{\nu}}{\chi^{2}}R^{2} - 8\left(R_{;\mu}\phi_{\nu} + R_{;\nu}\phi_{\mu} + R\phi_{\mu\nu}\right) + 8g_{\mu\nu}\left(R\Box\phi + R^{;\alpha}\phi_{\alpha}\right) \right\}.$$

$$(6.33)$$

As noted before, in the special case of a static, spherically symmetric spacetime this stress-energy tensor coincides with the analytic approximation of Anderson *et al* [12] for the zero-temperature case. Furthermore, in the massless, conformally coupled limit, it coincides with the zero-temperature Killing approximation [10], and the zero-temperature Page-Brown-Ottewill approximation [8,9]. Our high-frequency approximation can in fact be considered as an extension of these schemes to the general case when the spacetime is static, but not necessary spherically symmetric, and the field equation includes an arbitrary mass and potential V.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Chapter 7

Discussion and Future Work

In the presence of a continuous spacetime symmetry, the field equation can be solved by decomposition of the field into harmonics. This effectively replaces the single field in the physical theory by a collection of lower-dimensional fields. One can then express quantities of interest, such as the Green function and the effective action, as sums of the corresponding objects in a dimensionally reduced theory. Due to the presence of ultraviolet divergences, however, these decompositions have only formal meaning, as the renormalization violates the exact form of such representations. As a result, the expression for the renormalized expectation value of the object in the physical spacetime can be obtained by summing the contributions of corresponding lower-dimensional quantities only if additional anomalous terms are added to each of the modes. We call this effect the *dimensional-reduction anomaly*.

The anomaly may have several sources. The dimensional reduction may change the global properties of the spacetime, inducing additional boundary conditions which the dimensionally reduced field must satisfy. This effect was seen in the decomposition of flat space into spherical modes (Section 3.3), and into Rindler-time modes (Section 5.1). Because of its nonlocal nature, this effect cannot be eliminated by additional finite renormalization of the dimensionally reduced theory. In addition to this "global" contribution to the dimensionalreduction anomaly, there also exist "local" contributions due to the curvature and the potential induced by the dimensional reduction. The corresponding anomalous terms are local invariants constructed from the curvature, the dilaton field, and their covariant derivatives. Comparison to the Schwinger-DeWitt expansion shows that these additional anomalous terms also cannot be eliminated by additional finite renormalizations. The dimensional-reduction anomaly is therefore not merely an artifact of renormalization ambiguity, and cannot be ignored in the analysis of quantum fields via dimensional reduction. Failure to account for the anomaly will lead to incorrect predictions for the contributions of fixed modes to expectation values in the physical theory. Furthermore, when summed over all modes, the renormalized expectation values of the dimensionally reduced theory will in general diverge if the anomaly is not included. Naïve dimensional reduction and renormalization (ignoring the anomaly) cannot yield finite results for the physical theory.

We have explicitly demonstrated the importance of the dimensional-reduction anomaly for the study of the stress tensor of a quantum field in a black-hole spacetime using dimensional reduction. The stress tensors predicted by twodimensional dilaton-gravity models for the $\ell = 0$ mode are qualitatively very different near the event horizon from the stress tensor in four dimensions. The anomaly appears to correct the predictions of the dimensionally reduced theory at the event horizon, while leaving the asymptotic behaviour from individual modes unaffected.

We have also demonstrated how mode decompositions can be used to obtain a very simple and general high-frequency approximation scheme for renormalized expectation values of quantum fields in static spaces. This scheme requires only the structure of the divergences in the theory and minimal assumptions about the spacetime (the existence of a Killing vector). It can therefore be applied to fields with arbitrary parameters in a wide class of spacetimes.

There are a number of topics presented in this thesis which merit further investigation. In dilaton-gravity theories, the contribution of the $\ell > 0$ modes is of interest, particularly near the potential barrier at $r \approx 3M$. The high-frequency approximation scheme developed in Chapter 6 may be useful here for confirming the WKB approximation of Balbinot *et al* for the $\ell = 0$ mode, and for deriving the dimensionally reduced stress tensors for $\ell > 0$. Better approximations for the stress tensor in black-hole spacetimes would aid in the accurate modelling of the back-reaction on the geometry and the subsequent evaporation process.

The dimensional-reduction anomaly could also be of relevance to Kaluza-Klein theories [86, 87], in which the assumption of extra 'hidden' dimensions is used in attempts to explain the physical properties of our four-dimensional universe in a more natural manner. Such theories arise, for example, as a natural consequence of the low-energy limit of string theory [88]. The dimensionalreduction anomaly implies that one might expect additional quantitative and even qualitative differences between the higher-dimensional quantum field theory and the effective four-dimensional behaviour of matter in our universe, and this possibility has yet to be fully explored [89].

The generalization of our approximation scheme to include finite-temperature effects would be of great value. In that case, one could regard our scheme as a proper extension of the Killing and WKB techniques to general static spaces with nonconformal fields.

Also important is the testing of our approximation scheme to establish its range of validity. While in the special cases of conformally invariant fields and Schwarzschild spacetimes it reduces to known approximation methods (Killing and WKB respectively), one does not expect this high-frequency approximation to be useful in all cases, such as for fields of large mass. A determination of the criteria for the validity of this scheme is required if it is to be applied to other systems.
Appendix A

Quantum Fields in Flat Spacetime at Finite Temperature

As an exercise we derive the renormalized expectation values of the square of the field operator and the stress tensor for a scalar field in four- and two-dimensional flat spacetimes at finite temperature. These are useful for interpreting the contribution of s mode alone, derived in Section 3.2.

The total renormalized values of $\langle \hat{\Phi}^2 \rangle^{\beta}$ and $\langle \hat{T}_{\mu\nu} \rangle^{\beta}$ for a scalar field in flat spacetime are easily calculated using (3.18), (3.19) of Section 3.2. Decomposing the field $\hat{\Phi}$ in terms of the standard plane-wave modes

$$\Phi_{\vec{k}} \equiv \frac{\mathrm{e}^{ik_{\alpha}x^{\alpha}}}{\sqrt{2\omega(2\pi)^{n-1}}},\tag{A.1}$$

where $k^0 = \omega_{\vec{k}} \equiv \sqrt{k^2 + m^2}$, $k \equiv |\vec{k}|$, and each of the components of the vector \vec{k} range over $(-\infty, \infty)$, we find the renormalized Hadamard Green function to be

$$G_{\beta}^{(1)}(X,X') = \int_{-\infty}^{\infty} d\vec{k} \frac{2}{\mathrm{e}^{\beta\omega_{\vec{k}}} - 1} \left(\Phi_{\vec{k}} \overline{\Phi}_{\vec{k}}' + \overline{\Phi}_{\vec{k}} \Phi_{\vec{k}}' \right) \,. \tag{A.2}$$

A.1 Four dimensions

In four dimensions we have

$$\int_{-\infty}^{\infty} d\vec{k} = 4\pi \int_{0}^{\infty} dk \, k^2 \,. \tag{A.3}$$

Using the modes (A.1) with (3.18), we obtain

$$\langle \hat{\Phi}^2 \rangle^{\beta} = \frac{1}{2\pi^2} \int_0^\infty dk \, \frac{1}{e^{\beta\sqrt{k^2 + m^2}} - 1} \frac{k^2}{\sqrt{k^2 + m^2}}.$$
 (A.4)

Note that this is equal to the expression for the contribution of the s-mode alone, (3.30), evaluated at r = 0. For a massless field we find

$$\langle \hat{\Phi}^2 \rangle_{m=0}^{\beta} = \frac{1}{2\pi^2} \int_0^{\infty} d\omega \, \frac{\omega}{\mathrm{e}^{\beta\omega} - 1} = \frac{1}{12\beta^2} \,, \tag{A.5}$$

while in the massive case we have

$$\langle \hat{\Phi}^2 \rangle^{\beta} = \frac{m^2}{2\pi^2} \sum_{n=1}^{\infty} \frac{K_1(nm\beta)}{nm\beta}$$
 (A.6)

$$\stackrel{m\beta \gg 1}{=} \frac{1}{2\pi^2} \frac{m}{\beta} K_1(m\beta) \tag{A.7}$$

$$\stackrel{m\beta \ll 1}{=} \frac{1}{12\beta^2} - \frac{m}{4\pi\beta} + \frac{m^2}{16\pi^2} - \frac{m^2}{8\pi^2} \left(\gamma + \ln\frac{m\beta}{4\pi}\right) \,. \tag{A.8}$$

Similarly, for the total stress tensor we have

$$\langle \hat{T}_{\mu}^{\nu} \rangle^{\beta} = \frac{1}{(2\pi)^3} \int_{-\infty}^{\infty} d\vec{k} \; \frac{1}{\mathrm{e}^{\beta\sqrt{k^2 + m^2}} - 1} \; \frac{k_{\mu}k^{\nu}}{\sqrt{k^2 + m^2}} \,.$$
 (A.9)

Note that $\langle \hat{T}_{\mu}{}^{\nu} \rangle^{\beta}$ is diagonal, independent of ξ , and also traceless for m = 0. In the massless case we need only compute the time-time component,

$$\langle \hat{T}_{tt} \rangle^{\beta} = \frac{1}{2\pi^2} \int_0^\infty d\omega \, \frac{\omega^3}{\mathrm{e}^{\beta\omega} - 1} = \frac{\pi^2}{30} T^4 \,,$$
 (A.10)

and the total stress tensor is then¹

$$\langle \hat{T}_{\mu}^{\ \nu} \rangle^{\beta} = \frac{\pi^2}{30} T^4 \begin{pmatrix} -1 & 0 & 0 & 0 \\ 0 & \frac{1}{3} & 0 & 0 \\ 0 & 0 & \frac{1}{3} & 0 \\ 0 & 0 & 0 & \frac{1}{3} \end{pmatrix} .$$
(A.11)

In the massive case let us consider first the low temperature limit, $m\beta \gg 1$. The trace of the stress tensor is given by

$$\langle \hat{T}_{\nu}^{\nu} \rangle_{m\beta \gg 1} = \frac{-m^2}{2\pi^2} \int_0^\infty dk \, \frac{k^2}{\sqrt{k^2 + m^2}} e^{-\beta\sqrt{k^2 + m^2}} = \frac{1}{(2\pi)^2} \left[-\frac{2m^3}{\beta} K_1(m\beta) \right] .$$
 (A.12)

Combining this with the time-time component

$$\langle \hat{T}_{tt} \rangle_{m\beta \gg 1} = \frac{1}{2\pi^2} \int_0^\infty dk \, k^2 \sqrt{k^2 + m^2} \, \mathrm{e}^{-\beta \sqrt{k^2 + m^2}} \\ = \frac{1}{(2\pi)^2} \left[\frac{2m^3}{\beta} K_1(m\beta) + \frac{6m^2}{\beta^2} K_0(m\beta) + \frac{12m}{\beta^3} K_1(m\beta) \right]$$
(A.13)

yields the total stress tensor

¹These calculations have been performed in Cartesian coordinates; however, in transforming to spherical coordinates any second-rank tensor of the form $T^{\nu}_{\mu} = \text{diag}(a, b, b, b)$ has the same components in both spherical and Cartesian coordinates.

The high-temperature limit can be obtained from the low-temperature result by substituting $\beta \rightarrow n\beta$ and summing over n from 1 to ∞ . Using the sums (3.39)-(3.41), we obtain

A.2 Two dimensions

Repeating these calculations for two-dimensional space, where

$$\int_{-\infty}^{\infty} d\vec{k} = 2 \int_{0}^{\infty} dk , \qquad (A.16)$$

we find that $\langle \hat{\Phi}^2 \rangle^{\beta}$ is divergent for a massless field, while in the massive case we have

$$\langle \hat{\Phi}^2 \rangle^{\beta} \stackrel{m\beta \gg 1}{=} \frac{1}{\pi} K_0(m\beta),$$
 (A.17)

$$\stackrel{m\beta \ll 1}{=} \frac{1}{2m\beta} + \frac{1}{2\pi} \left(\gamma + \ln \frac{m\beta}{4\pi} \right) . \tag{A.18}$$

The stress tensor is

$$\langle \hat{T}_{\mu}^{\nu} \rangle^{\beta} \stackrel{m=0}{=} \frac{\pi^2}{6\beta^2} \begin{pmatrix} -1 & 0 \\ 0 & 1 \end{pmatrix},$$
 (A.19)

$$\overset{m\beta \gg 1}{=} \frac{m^2}{\pi} \left[\begin{pmatrix} -1 & 0 \\ 0 & 1 \end{pmatrix} \frac{K_1(m\beta)}{m\beta} + \begin{pmatrix} -1 & 0 \\ 0 & 0 \end{pmatrix} K_0(m\beta) \right], \quad (A.20)$$

$$\overset{m\beta \ll 1}{=} \begin{pmatrix} -1 & 0 \\ 0 & 1 \end{pmatrix} \left\{ \frac{\pi}{6\beta^2} - \frac{m}{2\beta} - \frac{m^2}{4\pi} \left(\gamma + \ln \frac{m\beta}{4\pi} \right) + \frac{m^2}{8\pi} \right\}$$

$$+ \begin{pmatrix} -1 & 0 \\ 0 & 0 \end{pmatrix} \left\{ \frac{m}{2\beta} + \frac{m^2}{2\pi} \left(\gamma + \ln \frac{m\beta}{4\pi} \right) \right\}. \quad (A.21)$$

Note that the $r \to \infty$ limit of the contribution of the s mode alone in four dimensions, derived in Section 3.2, are $(4\pi r^2)^{-1}$ times these.

Appendix B

Spherical Decompositions

B.1 Spherical Decomposition of Curvatures

Consider a line element of the form

$$ds^{2} = g_{\mu\nu} dX^{\mu} dX^{\nu} = h_{ab} dx^{a} dx^{b} + \rho^{2} e^{-2\phi} \omega_{ij} dy^{i} dy^{j} , \qquad (B.1)$$

where $h_{ab} = h_{ab}(x^c)$ is an arbitrary two-dimensional metric and $\omega_{ij} = \omega_{ij}(y^k)$ is the metric of a two-sphere. The dilaton ϕ is a function of the x^a only, and ρ is a constant with the dimensions of length. The radius of a two-sphere of fixed x^a is $r = \rho e^{-\phi}$.

We wish to decompose our field theory in terms of modes on the two-sphere. This requires rewriting four-dimensional geometric quantities like the curvatures in terms of the corresponding curvatures for the metric h.

Our notational conventions are as follows: four-dimensional covariant derivatives are denoted by $()_{;a}$, while \Box is understood to represent the d'Alembertian with respect to g. Meanwhile, ∇ , $()_{|a}$ and Δ_{h} are the two-dimensional covariant derivatives and d'Alembertian calculated using the metric h_{ab} . For the dilaton ϕ we shall understand ϕ_{a} , ϕ_{ab} , etc. to denote multiple two-dimensional covariant derivatives of ϕ . For example, the four-dimensional d'Alembertian of an angle-independent scalar S decomposes to

$$\Box S = \Delta_h S - 2\nabla \phi \cdot \nabla S \,. \tag{B.2}$$

In particular,

$$\Box \phi = \Delta_h \phi - 2(\nabla \phi)^2. \tag{B.3}$$

For the given line element, the nonvanishing Christoffel symbols are

$${}^{4}\Gamma^{a}_{bc}[g] = {}^{2}\Gamma^{a}_{bc}[h], \qquad (B.4)$$

$${}^{4}\Gamma^{a}_{ij}[g] = \phi^{a}g_{ij}, \qquad (B.5)$$

$${}^{4}\Gamma^{i}_{ja}[g] = -\phi_a \delta^{i}_j, \qquad (B.6)$$

$${}^{4}\Gamma^{k}_{ij}[g] = {}^{2}\Gamma^{k}_{ij}[\omega]. \qquad (B.7)$$

Selecting coordinates (θ, η) on the two-spheres, where

$$\omega_{ij}dy^i dy^j = d\theta^2 + \sin^2\theta \, d\eta^2 \,, \tag{B.8}$$

one finds

$${}^{4}\Gamma^{\theta}_{\eta\eta}[g] = -\sin\theta\cos\theta , \qquad {}^{4}\Gamma^{\eta}_{\eta\theta}[g] = \frac{\cos\theta}{\sin\theta} . \tag{B.9}$$

For convenience, we define the following commonly occurring functions of the dilaton field:

$$A = 1 - r^2 (\nabla \phi)^2, \qquad (B.10)$$

$$B = \Delta_h \phi - 2(\nabla \phi)^2, \qquad (B.11)$$

$$T_{ab} = \phi_{ab} - \phi_a \phi_b , \qquad (B.12)$$

$$T = h^{ab}T_{ab} = \Delta_h \phi - (\nabla \phi)^2. \qquad (B.13)$$

Since the two-sphere metric has constant curvature, ${}^{2}R[\omega] = 2$, explicit reference to it may be dropped. Henceforth we shall assume all curvatures to be with respect to the two-dimensional metric h_{ab} unless explicitly labelled otherwise. Using this notation, one can show that the only nonvanishing components of the four-dimensional curvatures are

$${}^{4}R_{abcd}[g] = \frac{1}{2} R \left(h_{ac} h_{bd} - h_{ad} h_{bc} \right), \qquad (B.14)$$

$${}^{4}R_{aibj}[g] = g_{ij}T_{ab},$$
 (B.15)

$${}^{4}R_{ijkm}[g] = \frac{A}{r^{2}}(g_{ik}g_{jm} - g_{im}g_{jk}), \qquad (B.16)$$

$${}^{4}R_{ab}[g] = \frac{1}{2} R h_{ab} + 2T_{ab}, \qquad (B.17)$$

$${}^{4}R_{ij}[g] = g_{ij}\left[\frac{1}{r^{2}} + B\right], \qquad (B.18)$$

$${}^{4}R[g] = R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}}, \qquad (B.19)$$

while the only nonvanishing ${}^4\!R_{\alpha\beta;\gamma}$ are

$${}^{4}R_{ab;c}[g] = \frac{1}{2}h_{ab}R_{|c} + 2T_{ab|c}, \qquad (B.20)$$

$${}^{4}R_{am;n}[g] = g_{mn} \left[\left(-\frac{1}{2}R + \frac{1}{r^{2}} + B \right) \phi_{a} - 2T_{ab} \phi^{b} \right], \qquad (B.21)$$

$${}^{4}R_{mn;a}[g] = g_{mn} \left(\frac{1}{r^{2}} + B\right)_{|a}.$$
(B.22)

Also,

$${}^{4}R_{mn;ab}[g] = g_{mn} \left(\frac{1}{r^{2}} + B\right)_{|ab}, \qquad (B.23)$$

$${}^{4}R_{mn;jk}[g] = -(g_{km}g_{nj} + g_{kn}g_{mj}) \left[\left(-\frac{1}{2}R + \frac{1}{r^{2}} + B \right) (\nabla \phi)^{2} - 2T_{ab}\phi^{a}\phi^{b} \right] - g_{jk}g_{mn} \left(\frac{1}{r^{2}} + B \right)_{|a}\phi^{a}, \qquad (B.24)$$

$$\Box^{4}R_{mn}[g] = g_{mn}\left\{ \left[\Delta_{h} - 2\nabla\phi \cdot \nabla\right] \left(\frac{1}{r^{2}} + B\right) + R(\nabla\phi)^{2} - 2\left(\frac{1}{r^{2}} + B\right)(\nabla\phi)^{2} + 4T_{ab}\phi^{a}\phi^{b} \right\}, \qquad (B.25)$$

$${}^{4}R_{;a}[g] = \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}}\right)_{|a}, \qquad (B.26)$$

$${}^{4}R_{;ab}[g] = \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}}\right)_{|ab}, \qquad (B.27)$$

$${}^{4}R_{;mn}[g] = -g_{mn}\left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}}\right)_{|a}\phi^{a}, \qquad (B.28)$$

$$\Box^{4}R[g] = [\Delta_{h} - 2\nabla\phi \cdot \nabla] \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}} \right).$$
(B.29)

B.2 Small-Distance Expansions

To perform the decomposition into spherical harmonics it is necessary to know the behaviour of σ and the $D^{\frac{1}{2}}a_n$ for X and X' separated along the two-spheres. Without loss of generality we take the points to be split in the θ direction only, with angular separation $\lambda = \theta - \theta'$. Our procedure will be to calculate the desired quantities first as expansions in powers of λ^2 , and then to convert them to expansions in powers of $(1-\cos \lambda)$ for use in the mode-decomposition calculations.

We take as our ansatz for the geodetic interval σ

$$2\sigma(x,y;x',y') = (\tilde{r}\lambda)^2 + U(\tilde{x})(\tilde{r}\lambda)^4 + V(\tilde{x})(\tilde{r}\lambda)^6 + \cdots, \qquad (B.30)$$

where $\tilde{x} \equiv \frac{1}{2}(x + x')$. Taking the derivative of $\tilde{\sigma}$ with respect to each of the coordinates and requiring $\sigma = \frac{1}{2}g^{\alpha\beta}\sigma_{\alpha}\sigma_{\beta}$ in the coincidence limit, one can show that

$$U(x) = -\frac{1}{12} (\nabla \phi)^2, \qquad (B.31)$$

$$V(x) = \frac{1}{90} (\nabla \phi)^4 - \frac{1}{120} \phi^a \phi^b \phi_{ab} , \qquad (B.32)$$

and

$$(\sigma^{\theta})^{2} = \lambda^{2} \left[1 - \frac{1}{3} r^{2} (\nabla \phi)^{2} \lambda^{2} + r^{4} \left(\frac{17}{180} (\nabla \phi)^{4} - \frac{1}{20} \phi^{a} \phi^{b} \phi_{ab} \right) \lambda^{4} + \cdots \right] ,$$
(B.33)

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

137

$$\sigma^{\eta} = 0, \qquad (B.34)$$

$$\sigma^{a} = -\frac{1}{2}\phi^{a}(r\lambda)^{2} + \left[-\frac{1}{24}\phi^{ab}\phi_{b} + \frac{1}{12}(\nabla\phi)^{2}\phi^{a}\right](r\lambda)^{4} + \cdots .$$
(B.35)

The expansion (B.30) for σ can be converted into one in terms of $(1 - \cos \lambda)$ using

$$\lambda^{2} = 2(1-z) + \frac{1}{3}(1-z)^{2} + \frac{4}{45}(1-z)^{3} + \cdots, \qquad (B.36)$$

where $z \equiv \cos \lambda$. Defining the functions u(x), v(x) by

$$2\sigma(x,y;x,y') = 2r^2 \left[(1-z) + u(x)(1-z)^2 + v(x)(1-z)^3 + \cdots \right], \quad (B.37)$$

we obtain

$$u(x) = \frac{1}{6} [1 - r^2 (\nabla \phi)^2], \qquad (B.38)$$

$$v(x) = \frac{2}{45} \left[1 - \frac{5}{4} r^2 (\nabla \phi)^2 + r^4 (\nabla \phi)^4 - \frac{3}{8} r^4 \nabla \phi \cdot \nabla [(\nabla \phi)^2] \right]. \quad (B.39)$$

Combining (B.33-B.36) with the results of Appendix B.1 and the shortdistance expansions of [80, 81], one can derive expansions for the $D^{\frac{1}{2}}a_n$ in powers of (1-z). Writing

$$D^{\frac{1}{2}}a_{n}^{\Box-\xi^{4}R} = \Re_{n}^{\Box-\xi^{4}R} = \Re_{n(0)}^{\Box-\xi^{4}R} + \Re_{n(1)}^{\Box-\xi^{4}R}(1-z) + \Re_{n(2)}^{\Box-\xi^{4}R}(1-z)^{2} + \cdots , \quad (B.40)$$

it can be shown that

$$\Re_{0(0)}^{\Box-\xi^{4}R} = 1, \qquad (B.41)$$

$$\Re_{0(1)}^{\Box-\xi^{4}R} = \frac{1}{6}(1+r^{2}B), \qquad (B.42)$$

$$\Re_{0(2)}^{\Box-\xi^{4}R} = \frac{1}{90}A^{2} + \frac{1}{72}(1+r^{2}B)^{2} + \frac{1}{36}(1+r^{2}B)(1-4r^{2}(\nabla\phi)^{2}) + \frac{r^{4}}{180} \left[\frac{3}{2}R(\nabla\phi)^{2} + 6T_{ab}\phi^{a}\phi^{b} + 2T_{ab}T^{ab} + 12\left(\frac{1}{r^{2}} + B\right)(\nabla\phi)^{2} + 6\left(\frac{1}{r^{2}} + B\right)_{|a}\phi^{a}\right], \qquad (B.43)$$

$$\Re_{1(0)}^{\Box-\xi^{4}R} = \left(\frac{1}{6} - \xi\right) \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}}\right), \qquad (B.44)$$

$$\begin{aligned} \Re_{1(1)}^{\Box-\xi^{4}R} &= \frac{1}{6} \left(\frac{1}{6} - \xi \right) \left[(1+r^{2}B) + r^{2}\nabla\phi \cdot \nabla \right] \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}} \right) \\ &+ \frac{r^{2}}{180} \left[RT + 3R(\nabla\phi)^{2} + 8T_{ab}T^{ab} + 12T_{ab}\phi^{a}\phi^{b} \right. \\ &+ \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}} \right)_{|a}\phi^{a} \\ &+ 3[\Delta_{h} - 2\nabla\phi \cdot \nabla] \left(\frac{1}{r^{2}} + B \right) - 6 \left(\frac{1}{r^{2}} + B \right) (\nabla\phi)^{2} \right] \\ &+ \frac{1}{90r^{2}} \left(2A^{2} + A(1+r^{2}B) - 2(1+r^{2}B)^{2} \right) , \end{aligned}$$
(B.45)

$$\begin{aligned} \Re_{2(0)}^{\Box-\xi^{4}R} &= \frac{1}{2} \left(\frac{1}{6} - \xi \right)^{2} \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}} \right)^{2} \\ &+ \frac{1}{6} \left(\frac{1}{6} - \xi \right) \left[\Delta_{h} - 2\nabla\phi \cdot \nabla \right] \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}} \right) \\ &+ \frac{1}{180} \left[\left[\Delta_{h} - 2\nabla\phi \cdot \nabla \right] \left(R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} + \frac{2}{r^{2}} \right) \\ &+ \frac{1}{2}R^{2} - 2RT + 4T_{ab}T^{ab} + \frac{4}{r^{4}}A^{2} - \frac{2}{r^{4}}(1 + r^{2}B)^{2} \right] . \end{aligned}$$
(B.46)

It is easily verified that for flat spacetime each of the $\Re_{n(k)}^{\Box-\xi^{4}R}$ vanishes, except for $\Re_{0(0)}^{\Box-\xi^{4}R}$.

Appendix C

Static Decompositions

These appendices contain formulae required for the calculations in Chapters 5 and 6, which deal with mode decompositions in static spaces.

C.1 Small-Distance Expansions for (1+3) Reductions

The line element for a static space may be written in the form

$$ds^{2} = e^{-4\phi(x)}dt^{2} + h_{ab}(x)dx^{a}dx^{b} .$$
 (C.1)

For this static metric h_{ab} is an induced 3-metric and $n^{\alpha} = e^{2\phi(x)}\delta_t^{\alpha}$ is a unit vector normal to the surfaces of constant t. The extrinsic curvature K_{ab} on these surfaces vanishes. The nonvanishing Christoffel symbols are

$${}^{4}\Gamma^{a}_{bc}[g] = \Gamma^{a}_{bc}[h], \quad {}^{4}\Gamma^{a}_{00} = 2e^{-4\phi}\phi^{;a}, \quad {}^{4}\Gamma^{0}_{0a} = -2\phi_{;a} \quad . \tag{C.2}$$

Because we will be using some quantities defined in terms of the full fourdimensional metric g and others defined in terms of the three-dimensional metric h, some conventions on notation are in order. Henceforth four-dimensional curvatures and covariant derivatives will be denoted by ${}^{4}R_{\cdots}$ and ()_i respectively, while \Box is understood to represent the d'Alembertian with respect to g. All other curvatures and covariant derivatives are understood to be calculated using the three-metric h. In particular, ∇ and ()_{|a} are three-dimensional covariant derivatives, and Δ_h is the three-dimensional d'Alembertian. For the dilaton ϕ we shall understand ϕ_a , ϕ_{ab} , etc. to denote multiple three-dimensional covariant derivatives of ϕ .

With these conventions we have, for example,

$$\Box \phi = \Delta_h \phi - 2(\nabla \phi)^2 . \tag{C.3}$$

It is convenient to define the following three-dimensional tensor which occurs naturally in the 1 + 3 reduction:

$$T_{ab} = 2\left[\phi_{ab} - 2\phi_a\phi_b\right], \qquad (C.4)$$

$$T = T_a^a = 2 \left[\Delta_h \phi - 2 (\nabla \phi)^2 \right] = 2 \Box \phi$$
. (C.5)

In terms of T_{ab} the only nonvanishing components of the four-dimensional curvatures are

$${}^{4}R_{abcd} = R_{abcd}, \quad {}^{4}R_{0a0b} = e^{-4\phi}T_{ab}, \quad (C.6)$$

$${}^{4}\!R_{ab} = R_{ab} + T_{ab} , \quad {}^{4}\!R_{00} = e^{-4\phi}T , \quad {}^{4}\!R = R + 2T .$$
 (C.7)

We shall also need the following expressions for ${}^{4}R_{\alpha\beta;\gamma}$, ${}^{4}R_{\alpha\beta;\gamma\delta}$, ${}^{4}R_{;\alpha}$, and ${}^{4}R_{;\alpha\beta}$:

$${}^{4}R_{ab;c} = (R_{ab} + T_{ab})_{|c} , \qquad (C.8)$$

$${}^{4}R_{a0;0} = -2e^{-4\phi} \left[\left(R_{ab} + T_{ab} \right) \phi^{b} - T\phi_{a} \right] , \qquad (C.9)$$

$${}^{4}R_{00;c} = e^{-4\phi}T_{|c}, \qquad (C.10)$$

$${}^{4}R_{ab;cd} = (R_{ab} + T_{ab})_{|cd}, \qquad (C.11)$$

$${}^{4}R_{ab;00} = -2e^{-4\phi} \left[(R_{ab} + T_{ab})_{|c} \phi^{c} - 4\phi_{a}\phi_{b}T + 2\phi_{a} (R_{bc} + T_{bc}) \phi^{c} + 2\phi_{b} (R_{ac} + T_{ac}) \phi^{c} \right], \qquad (C.12)$$

$${}^{4}R_{a0;c0} = -2e^{-4\phi} \left[(R_{ab} + T_{ab})_{|c} \phi^{b} + 2\phi^{b} (R_{ab} + T_{ab}) \phi_{c} - 2\phi_{a}\phi_{c}T - \phi_{a}T_{|c} \right], \qquad (C.13)$$

$${}^{4}R_{a0;0d} = -2e^{-4\phi} \left[\left(R_{ab} + T_{ab} \right) \phi^{b} - T\phi_{a} \right]_{|d} , \qquad (C.14)$$

$${}^{4}R_{00;cd} = e^{-4\phi}T_{|cd}, \qquad (C.15)$$

$${}^{4}R_{00;00} = 2e^{-8\phi} \left[4 \left(R_{ab} + T_{ab} \right) \phi^{a} \phi^{b} - 4T \phi^{a} \phi_{a} - T_{|a} \phi^{a} \right] , \qquad (C.16)$$

$${}^{4}R_{;a} = (R+2T)_{|a}, \qquad (C.17)$$

$${}^{4}R_{;ab} = (R+2T)_{|ab} , \qquad (C.18)$$

$${}^{4}R_{;00} = -2e^{-4\phi} \left(R + 2T\right)_{|a} \phi^{a}.$$
(C.19)

For the mode decomposition (Fourier time transform) we need to know the behaviour of the two-point functions σ and $D^{1/2}a_n^{\Box-V}$ for points $X^{\alpha} = (t, x)$ and $X'^{\alpha} = (t', x)$ where $\tau \equiv e^{-2\phi}(t-t')$ is small. Using the procedure of Appendix B.2 one can easily show that

$$2\sigma(t,x;t',x) = \tau^2 - \frac{1}{3}\phi^a\phi_a\tau^4 + \frac{1}{45}\left[8(\phi^a\phi_a)^2 - 3\phi^a\phi^b\phi_{ab}\right]\tau^6 + \cdots, \quad (C.20)$$

$$\sigma_{a}(t, x; t', x) = -\phi_{a}\tau^{2} + \frac{1}{6} \left[\phi^{b}\phi_{b}\phi_{a} - \phi^{b}\phi_{ba} \right] \tau^{4} + \cdots, \qquad (C.21)$$

$$\sigma_t(t, x; t', x) = e^{-2\phi} \tau \left[1 - \frac{2}{3} \phi^a \phi_a \tau^2 + \frac{1}{15} \left[8(\phi^a \phi_a)^2 - 3\phi^a \phi^b \phi_{ab} \right] \tau^4 \right] + \cdots .$$
(C.22)

Combining the above expressions with the results of [80,81] for small- σ^{α} expansions of the $D^{1/2}a_n$, it is easily shown that for the operator $\Box - V$, where the potential function V is independent of t, the first three $D^{1/2}a_n^{\Box-V}$ to $O(R^2)$ are

$$D^{1/2}a_{0}^{\Box-V} = 1 + \frac{1}{12}T\tau^{2} + \frac{1}{360} \left[6\left(R_{ab} + T_{ab}\right)\phi^{a}\phi^{b} - 16T\phi^{a}\phi_{a} + 6T_{|a}\phi^{a} + \frac{5}{4}T^{2} + T^{ab}T_{ab} \right]\tau^{4}, \qquad (C.23)$$

$$D^{1/2}a_{1}^{\Box-V} = \left(\frac{1}{6}R - V + \frac{1}{3}T\right) + \left\{\frac{1}{2}\left(\frac{1}{6}R - V + \frac{1}{3}T\right)_{|a}\phi^{a} + \left(\frac{1}{3}V_{|a}\phi^{a} - \frac{1}{12}VT\right) + \frac{1}{360}\left[3\Delta_{h}T + 6T^{2} + 6T^{ab}T_{ab} + 5RT + 2R^{ab}T_{ab} + 24\left(R_{ab} + T_{ab}\right)\phi^{a}\phi^{b} - 24T\phi^{a}\phi_{a} - 18R_{|a}\phi^{a} - 42T_{|a}\phi^{a}\right]\right\}\tau^{2},$$
(C.24)

$$D^{1/2}a_{2}^{\Box-V} = \frac{1}{2} \left(\frac{1}{6}R - V + \frac{1}{3}T \right)^{2} + \frac{1}{30} \left[\Delta_{h}(R + 2T) - 2(R + 2T)_{|a}\phi^{a} \right] + \frac{1}{180} \left[R_{abcd}R^{abcd} + 4T^{ab}T_{ab} - (R^{ab} + T^{ab})(R_{ab} + T_{ab}) - T^{2} \right] - \frac{1}{6} \left[\Delta_{h}V - 2V_{|a}\phi^{a} \right].$$
(C.25)

C.2 Small-Distance Expansions for (2+2) Reductions

Consider a spacetime with the line element

$$ds^{2} = e^{-2\phi(x)} (dt_{0}^{2} + dt_{1}^{2}) + h_{AB}(x) dx^{A} dx^{B}.$$
 (C.26)

In this case we will be using some quantities defined in terms of the full fourdimensional metric $g_{\mu\nu}$, and others in terms of the two-dimensional metric h_{AB} (A, B = 2, 3).¹ In analogy to the (1 + 3)-splitting case, four-dimensional curvatures and covariant derivatives will be denoted by ${}^{4}R_{...}$ and ();_a respectively, while \Box is understood to represent the d'Alembertian with respect to $g_{\mu\nu}$. All other curvatures and covariant derivatives are understood to be calculated using the two-metric h_{AB} . In particular, ∇ and ()_{|A} are two-dimensional covariant

¹We use uppercase Latin indices for the n = 2 decomposition to avoid confusion with the n = 1 case.

derivatives, and Δ_h is the two-dimensional d'Alembertian. For the dilaton ϕ we shall understand ϕ_A , ϕ_{AB} , etc. to denote multiple two-dimensional covariant derivatives of ϕ . For example, with these conventions the four-dimensional d'Alembertian of a y-independent scalar S decomposes to

$$\Box S = \Delta_h S - 2\nabla \phi \cdot \nabla S . \tag{C.27}$$

In particular,

$$\Box \phi = \Delta_h \phi - 2(\nabla \phi)^2 = -\frac{1}{2} e^{2\phi} \Delta_h e^{-2\phi} . \qquad (C.28)$$

For the given line element, the nonvanishing Christoffel symbols are (i, j = 0, 1)

$$\Gamma^{A}_{BC}[g] = \Gamma^{A}_{BC}[h] = \frac{1}{2}h^{AD}(h_{DB,C} + h_{CD,B} - h_{BC,D}), \qquad (C.29)$$

$$\Gamma^{A}_{ij}[g] = \phi^{A} e^{-2\phi} \eta_{ij} = \phi^{A} g_{ij}, \qquad (C.30)$$

$$\Gamma^i_{jA}[g] = -\phi_A \delta^i_j = -\phi_A g^i_j, \qquad (C.31)$$

$$\Gamma_{Aij}[g] = -\Gamma_{ijA}[g] = \phi_A e^{-2\phi} \eta_{ij} = \phi_A g_{ij}.$$
 (C.32)

Meanwhile, the only nonvanishing components of the four-dimensional curvatures are

$${}^{4}R_{ABCD}[g] = \frac{1}{2}R\left(g_{AC}g_{BD} - g_{AD}g_{BC}\right), \qquad (C.33)$$

$${}^{4}R_{AmBn}[g] = g_{mn}[\phi_{AB} - \phi_{A}\phi_{B}], \qquad (C.34)$$

$${}^{4}R_{ijkm}[g] = -\phi_{A}\phi^{A}(g_{ik}g_{jm} - g_{im}g_{jk}), \qquad (C.35)$$

$${}^{4}R_{AB}[g] = \frac{1}{2}Rg_{AB} + 2[\phi_{AB} - \phi_{A}\phi_{B}], \qquad (C.36)$$

$${}^{4}R_{mn}[g] = g_{mn}[\phi^{A}_{A} - 2\phi^{A}\phi_{A}], \qquad (C.37)$$

$${}^{4}R[g] = R + 4\phi_{A}^{A} - 6\phi^{A}\phi_{A}.$$
 (C.38)

We shall also need the following components of ${}^{4}R_{\alpha\beta;\gamma}$, ${}^{4}R_{\alpha\beta;\gamma\delta}$, ${}^{4}R_{;\alpha}$, and ${}^{4}R_{;\alpha\beta}$:

$${}^{4}R_{AB;C}[g] = \frac{1}{2}g_{AB}R_{,C} + 2[\phi_{AB} - \phi_{A}\phi_{B}]_{|C}, \qquad (C.39)$$

$${}^{4}R_{Am;n}[g] = g_{mn} \left[-\frac{1}{2} R \phi_A + \phi_A \Delta_h \phi - 2 \phi^B \phi_{BA} \right], \qquad (C.40)$$

$${}^{4}R_{mn;A}[g] = g_{mn}[\Delta_{h}\phi - 2(\nabla\phi)^{2}]_{|A}, \qquad (C.41)$$

$${}^{4}R_{mn;AB}[g] = g_{mn}[\Delta_{h}\phi - 2(\nabla\phi)^{2}]_{|AB}, \qquad (C.42)$$

$${}^{4}R_{ij;km}[g] = (g_{ik}g_{jm} + g_{im}g_{jk}) \left[\frac{1}{2}R(\nabla\phi)^{2} - (\nabla\phi)^{2}\Delta_{h}\phi + 2\phi^{A}\phi^{B}\phi_{AB}\right]$$

$$- (g_{ij}g_{km}) \left[\Delta_{h}\phi - 2(\nabla\phi)^{2}\right]_{|A}\phi^{A}, \qquad (C.43)$$

$${}^{4}R_{;A}[g] = \left[R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2}\right]_{,A}, \qquad (C.44)$$

$${}^{4}R_{;AB}[g] = \left[R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2}\right]_{|AB}, \qquad (C.45)$$

$${}^{4}R_{;mn}[g] = -g_{mn} \left[R + 4\Delta_{h}\phi - 6(\nabla\phi)^{2} \right]_{|A} \phi^{A} .$$
 (C.46)

Note again that operators and curvatures are with respect to the two-dimensional metric h_{AB} unless explicitly labelled otherwise.

We now write out the expansions of $\sigma(\mathbf{t}, x; \mathbf{t}', x)$ and the $D^{1/2}a_n(\mathbf{t}, x; \mathbf{t}', x)$ for small separations. Defining $\tau^i \equiv e^{-\phi}(t^i - t'^i)$, we have to second order in the curvature

$$2\sigma(\mathbf{t}, x; \mathbf{t}', x) = \tau^2 - \frac{1}{12} (\nabla \phi)^2 \tau^4 + \frac{1}{360} \left[4 (\nabla \phi)^4 - 3\phi^A \phi^B \phi_{AB} \right] \tau^6, \qquad (C.47)$$

$$\sigma^{i}(\mathbf{t}, x; \mathbf{t}', x) = e^{\phi} \tau^{i} \left[1 - \frac{1}{6} \phi_{A} \phi^{A} \tau^{2} + \frac{1}{120} \left[4(\phi_{A} \phi^{A})^{2} - 3\phi^{A} \phi^{B} \phi_{AB} \right] \tau^{4} \right], (C.48)$$

$$\sigma^{a}(\mathbf{t}, x; \mathbf{t}', x) = -\frac{1}{2}\phi^{A}\tau^{2} - \frac{1}{24}\phi_{B}\left(\phi^{BA} - 2\phi^{B}\phi^{A}\right)\tau^{4} + \frac{1}{720}\left[-12(\phi^{B}\phi_{B})^{2}\phi^{A} + 8\phi^{B}\phi_{B}\phi_{C}\phi^{CA} + 9\phi^{A}\phi^{B}\phi^{C}\phi_{BC} - 3\phi^{B}\phi_{BC}\phi^{CA} - \frac{3}{2}\phi_{B}\phi_{C}\phi^{BCA}\right]\tau^{6}.$$
(C.49)

Combining these expressions with the results of [80, 81], it is easily shown that for the operator $\Box - V$, where the potential function V is independent of t^i , the first three $D^{1/2}a_n^{\Box-V}$ are

$$D^{1/2}a_{0}^{\Box-V} = 1 + \frac{1}{12} \left[\Delta_{h}\phi - 2(\nabla\phi)^{2} \right] \tau^{2} + \frac{1}{1440} \left[3R(\nabla\phi)^{2} + 48(\nabla\phi)^{4} - 36(\nabla\phi)^{2}\Delta_{h}\phi - 44\phi^{A}\phi^{B}\phi_{AB} + 5(\Delta_{h}\phi)^{2} + 4\phi^{AB}\phi_{AB} + 12\phi^{A}(\Delta_{h}\phi)_{A} \right] \tau^{4} + O(\tau^{6}), \qquad (C.50)$$

$$D^{1/2}a_{1}^{\Box-V} = \left[\frac{1}{6}R - V + \frac{2}{3}\phi_{A}^{A} - \phi^{A}\phi_{A}\right] + \frac{1}{2}\left[-\frac{1}{6}V_{A}\phi^{A} - \frac{1}{6}V[\Delta_{h}\phi - 2(\nabla\phi)^{2}] + \frac{1}{30}R_{A}\phi^{A} + R[\frac{1}{30}\Delta_{h}\phi - \frac{2}{45}(\nabla\phi)^{2}] + \frac{1}{180}\left(60(\nabla\phi)^{4} - 62(\nabla\phi)^{2}\Delta_{h}\phi - 52\phi^{A}\phi^{B}\phi_{AB} + 16(\Delta_{h}\phi)^{2} - 4\phi^{AB}\phi_{AB} + 18\phi^{A}(\Delta_{h}\phi)_{A} - 12\phi^{A}\Delta_{h}(\phi_{A}) + 3\Delta_{h}^{2}\phi\right]\tau^{2} + O(\tau^{4}), \quad (C.51)$$

$$D^{1/2}a_{2}^{\Box-V} = \frac{1}{2} \left[\frac{1}{6}R - V + \frac{2}{3}\phi_{A}^{A} - \phi^{A}\phi_{A} \right]^{2} - \frac{1}{6}\Delta_{h}V + \frac{1}{3}V_{,A}\phi^{A} + \frac{1}{30}\Delta_{h}R - \frac{1}{15}R_{,A}\phi^{A} + \frac{1}{180} \left[\frac{1}{2}R^{2} - 2R[\Delta_{h}\phi - (\nabla\phi)^{2}] + 8(\nabla\phi)^{2}\Delta_{h}\phi + 136\phi^{A}\phi^{B}\phi_{AB} - 2(\Delta_{h}\phi)^{2} - 68\phi^{AB}\phi_{AB} - 72\phi^{A}\Delta_{h}(\phi_{A}) - 48\phi^{A}(\Delta_{h}\phi)_{A} + 24\Delta_{h}^{2}\phi \right] + O(\tau^{2}).$$
(C.52)

C.3 Useful Formulae

C.3.1 The Modified Bessel Function K_{ν}

The modified Bessel functions $K_{\nu}(z)$ may be defined via the integral

$$\int_{0}^{\infty} dx \, x^{-1-\nu} \exp\left\{-x - \frac{z^{2}}{4x}\right\} = 2\left(\frac{2}{z}\right)^{\nu} K_{\nu}(z) \,. \tag{C.53}$$

It may be shown that $K_{-\nu}(z) = K_{\nu}(z)$. Furthermore, for $\nu > 0$ the $K_{\nu}(z)$ obey the differential relation

$$\left(-\frac{1}{z}\frac{d}{dz}\right)^{n} z^{\nu} K_{\nu}(z) = z^{\nu-n} K_{\nu-n}(z)$$
(C.54)

In particular, for $z = m\sqrt{\tau^2 + \epsilon^2}$ one can easily show that

$$\frac{1}{z^n}K_n(z) = \left(-\frac{1}{m^2\epsilon}\frac{d}{d\epsilon}\right)^n K_0(z) \tag{C.55}$$

Combining (C.55) with integral (6.677) of [76],

$$\int_{-\infty}^{\infty} d\tau \, \cos(\bar{\omega}\tau) \, K_0(m\sqrt{\epsilon^2 + \tau^2}) = \frac{\pi}{\sqrt{m^2 + \bar{\omega}^2}} \exp(-\epsilon\sqrt{m^2 + \bar{\omega}^2}) \,, \qquad (C.56)$$

allows us to evaluate the (1 + 3)-splitting Fourier transforms of Sections 5.2.1, 5.3.1 as follows:

$$\int_{-\infty}^{\infty} d\tau \, \cos(\bar{\omega}\tau) \, \tau^{2k} \, \frac{1}{(m\sqrt{\tau^2 + \epsilon^2})^n} K_n(m\sqrt{\epsilon^2 + \tau^2}) = \\ = (-1)^k \frac{d^{(2k)}}{d\bar{\omega}^{(2k)}} \left(-\frac{1}{m^2\epsilon} \frac{d}{d\epsilon}\right)^n \frac{\pi}{\sqrt{m^2 + \bar{\omega}^2}} \exp(-\epsilon\sqrt{m^2 + \bar{\omega}^2}) \,. \tag{C.57}$$

For convenience, we have used the notation $\bar{\omega} = e^{2\phi}\omega$ introduced in Section 6.1.

C.3.2 Integrals of μ^n for the (1+3) Reduction

For $\mu = \sqrt{m^2 + x^2}$ it is easily shown that for large $\bar{\omega}$

$$\int_0^{\omega} dx \mu^3 = \frac{1}{4} \bar{\omega}^4 + \frac{3}{4} m^2 \bar{\omega}^2 + \frac{9}{32} m^4 + \frac{3}{8} m^4 \ln \frac{2\bar{\omega}}{m} , \qquad (C.58)$$

$$\int_{0}^{\bar{\omega}} dx \mu = \frac{1}{2} \bar{\omega}^{2} + \frac{1}{4} m^{2} + \frac{1}{2} m^{2} \ln \frac{2\bar{\omega}}{m} , \qquad (C.59)$$

$$\int_0^{\bar{\omega}} dx \frac{1}{\mu} = \ln \frac{2\bar{\omega}}{m} . \qquad (C.60)$$

In addition, for $n \ge 1$,

$$\int_0^\infty dx \mu^{-(2n+1)} = \frac{1}{m^{2n}} \frac{2^{n-1} (n-1)!}{(2n-1)!!} \,. \tag{C.61}$$

See for example (2.271) of [76]. These results are sufficient to perform the sums over modes in Sections 6.1 and 6.2.

C.3.3 Formulae for the (2+2) Reduction

The Fourier transforms of Sections 5.2.2 and 5.3.2 were computed before performing the *s*-integration by expanding all *z*-dependent quantities for small curvatures and using

$$\int_{-\infty}^{\infty} d^2 z \exp\left\{-\frac{1}{4s}z^2 + i\bar{p}z\right\} z^{2n} = (4\pi s) \mathrm{e}^{-\bar{p}^2 s} (4s)^n I_n , \qquad (\mathrm{C.62})$$

where $\bar{p} = e^{\phi}p$ and

$$I_0 = 1$$
, (C.63)

$$I_1 = 1 - \bar{p}^2 s , \qquad (C.64)$$

$$I_2 = 2 - 4\bar{p}^2 s + \bar{p}^4 s^2 , \qquad (C.65)$$

$$I_3 = 6 - 18\bar{p}^2s + 9\bar{p}^4s^2 - \bar{p}^6s^3 , \qquad (C.66)$$

$$I_4 = 24 - 96\bar{p}^2s + 72\bar{p}^4s^2 - 16\bar{p}^6s^3 + \bar{p}^8s^4 .$$
 (C.67)

For the summation over modes referred to in Sections 6.1 and 6.2 one may use the integrals

$$\int_0^{\bar{p}} dx \, x \, \ln \frac{\mu^2}{m^2} = \frac{1}{2} (\bar{p}^2 + m^2) \ln \frac{\bar{p}^2 + m^2}{m^2} - \frac{1}{2} \bar{p}^2 , \qquad (C.68)$$

$$\int_0^{\bar{p}} dx \, x \, \mu^{-2} = \frac{1}{2} \ln \frac{\bar{p}^2 + m^2}{m^2} \,, \qquad (C.69)$$

and for n > 1,

$$\int_0^\infty dx \, x \, \mu^{-2n} = \frac{1}{2(n-1)m^{2(n-1)}} \,, \tag{C.70}$$

where $\mu = \sqrt{m^2 + x^2}$.

Bibliography

- J. S. Dowker, Ann. Phys. (NY) 62, 361 (1971); R. Critchley, Ph. D. thesis, University of Manchester, 1976 (unpublished); J. S. Dowker and R. Critchley, Phys. Rev. D 15, 1484 (1977); T. S. Bunch and P. C. W. Davies, Proc. R. Soc. London A 357, 381 (1977); T. S. Bunch, Phys. Rev. D 18, 1844 (1978).
- [2] A. di Sessa, J. Math. Phys. (NY) 15, 1892 (1974); S. A. Fulling, L. Parker, and B. L. Hu, Phys. Rev. D 10, 3905 (1974), erratum *ibid* 11, 1714; C. M. Sommerfield, Ann. Phys. (NY) 84, 285 (1974); T. S. Bunch, Ph. D. thesis, King's College, London, 1977 (unpublished); T. S. Bunch, Phys. Rev. D 18, 1844 (1978); P. C. W. Davies and S. A. Fulling, Proc. R. Soc. London A 354, 59 (1977); T. S. Bunch, S. M. Christensen, and S. A. Fulling, Phys. Rev. D 18, 4435 (1978); V. M. Frolov, A. A. Grib, and V. M. Mostepanenko, Theor. and Math. Phys. 33, 869 (1977) and Phys. Lett. A 65, 282 (1978); D. M. Chitre and J. B. Hartle, Phys. Rev. D 16, 251 (1977).
- [3] J. S. Dowker and R. Critchley, Phys. Rev. D 13, 224 (1976), and Phys. Rev. D 13, 3224 (1976); G. Grensing, J. Phys. A: Gen. Phys. 10, 1687 (1977);
 N. D. Birrell, J. Phys. A: Gen. Phys. 12, 337 (1979); S. A. Fulling, Ph. D. thesis, Princeton University, 1972 (unpublished); C. Schomblond and P. Spindel, Ann. Inst. Henri Poincare A 25, 67 (1976); R. Critchley, Ph. D. thesis, University of Manchester, 1976 (unpublished); T. S. Bunch, Ph. D.

thesis, King's College, London, 1977 (unpublished); T. S. Bunch and P. C.
W. Davies, Proc. R. Soc. London A 360, 117 (1978); G. W. Gibbons and S.
W. Hawking, Phys. Lett. A 60, 385 (1977).

- [4] P. Candelas and J. S. Dowker, Phys. Rev. D 19, 2902 (1979).
- [5] S. M. Christensen and S. A. Fulling, Phys. Rev. D 15, 2088 (1977).
- [6] P. Candelas, Phys. Rev. D 21, 2185 (1980).
- [7] V. P. Frolov, Phys. Rev. D 26, 954 (1982) and in Quantum Gravity, proceedings of the Second Seminar on Quantum Gravity, Moscow, USSR, edited by M. A. Markov and P. C. West (Plenum, New York, 1984), p. 303; M. S. Fawcett and B. F. Whiting, in Quantum Structure of Space and Time, proceedings of the Nuffield Workshop, London, England, 1981, edited by M. J. Duff and C. J. Isham (Cambridge University Press, Cambridge, 1982); P. Candelas and K. W. Howard, Phys. Rev. D 29, 1618 (1984); P. Candelas and B. P. Jensen, Phys. Rev. D 33, 1596 (1986); M. S. Fawcett, Commun. Math. Phys. 89, 103 (1983); K. W. Howard and P. Candelas, Phys. Rev. Lett. 53, 403 (1984); K. W. Howard, Phys. Rev. D 30, 2532 (1984).
- [8] D. N. Page, Phys. Rev. D 25, 1499 (1982).
- [9] M. R. Brown and A. C. Ottewill, Phys. Rev. D 31, 2514 (1985); M. R. Brown,
 A. C. Ottewill, and D. N. Page, Phys. Rev. D 33, 2840 (1986).
- [10] V. P. Frolov and A. I. Zel'nikov, Phys. Rev. D 35, 3031 (1987).
- [11] P. R. Anderson, Phys. Rev. D 41, 1152 (1990).
- [12] P. R. Anderson, W. A. Hiscock, and D. A. Samuel, Phys. Rev. D 51, 4337 (1995).

- [13] S. Sushkov, Phys. Rev. D 62, 064007 (2000).
- [14] T. Zannias, Phys. Rev. D 30, 1161 (1984).
- [15] V. P. Frolov and K. S. Thorne, Phys. Rev. D 39, 2125 (1989).
- [16] V. P. Frolov and A. I. Zel'nikov, Phys. Lett. B 115, 372 (1982); *ibid* 123, 197 (1983); Phys. Rev. D 29, 1057 (1984).
- [17] P. R. Anderson, Phys. Rev. D 39, 3785 (1989).
- [18] B. E. Taylor, W. A. Hiscock, and P. R. Anderson, Phys. Rev. D 55, 6116 (1997).
- [19] T. Elster, Class. Quantum Grav. 1, 43 (1984); A. I. Zel'nikov and V. P. Frolov, Physical Effects in the Gravitational Field of Black Holes, proceedings of the Lebedev Physics Institute of the Academy of Sciences of the USSR, Series 169, edited by N. G. Basov (Nova Science, New York, 1987), p. 171; B. P. Jensen and A. Ottewill, Phys. Rev. D 39, 1130 (1989).
- [20] P. Thomi, B. Isaak and P. Hajicek, Phys. Rev. D 30, 1168 (1984).
- [21] G. Mandal, A. Sengupta, and S.R. Wadia, Mod. Phys. Lett. A 6, 1685 (1991); J.G. Russo, L. Susskind and L. Thorlacius, Phys. Lett. B 292, 13 (1992); T. Banks, A. Dabholkar, M. Douglas and M. O'Loughlin, Phys. Rev D 45, 3607 (1992); S.P. deAlwis, Phys. Lett. B 289, 278 (1992); S.D. Odintsov and I.L. Shapiro, Phys. Lett. B 263, 183 (1991); E. Elizalde, P. Fosalba-Vela, S. Naftulin, and S.D. Odintsov, Phys. Lett. B 352, 235 (1995); S. Elitzur, A. Forge and E. Rabinovici, Nucl. Phys. B 359, 581 (1991); E. Witten, Phys. Rev. D 44, 314 (1991); C. G. Callan, S. B. Giddings, J. A. Harvey, and A. Strominger, Phys. Rev. D 45, 1005 (1992).

- [22] V.P. Frolov, Phys. Rev. D 46, 5383 (1992); J.G. Russo and A.A. Tseytlin, Nucl. Phys. B 382, 259 (1992); S.D. Odintsov and I.J. Shapiro, Phys. Lett. B 263, 183 (1991) and Mod. Phys. Lett. A 7 437 (1992).
- [23] C. Teitelboim, Phys. Lett. B 126, 41 (1983); R. Jackiw, Quantum Theory of Gravity, ed. S. Christensen p. 403. (Hilger, Bristol, 1984).
- [24] K.C.K. Chan and R.B. Mann, Class. Quant. Grav. 12, 1609 (1995); J.S.F.
 Chan and R.B. Mann, Class. Quant. Grav. 12, 351 (1995); R.B. Mann and
 S.F. Ross, Phys. Rev. D 47, 3312 (1993); R.B. Mann, M.S. Morris, and S.F.
 Ross, Class. Quant. Grav. 10, 1477 (1993).
- [25] R.B. Mann, A. Shiekh and L. Tarasov, Nucl. Phys. B 341, 134 (1990); D. Banks and M. O'Loughlin, Nucl. Phys. B 362, 649 (1991); H.J. Schmidt, J. Math. Phys. 32, 1562 (1991); R.B. Mann, Phys. Rev. D 47, 4438 (1993); D. Louis-Martinez, J. Gegenberg and G. Kunstatter, Phys. Lett. B 321, 193 (1994); D. Louis-Martinez and G. Kunstatter, Phys. Rev. D 49, 5227 (1994); E. d'Hoker, D.Z. Freedman and R. Jackiw, Phys. Rev. D. 28, 2583 (1983); R. Jackiw, Nucl. Phys. B 252, 343 (1985); D. Cangemi and R. Jackiw, Phys. Rev. Lett 69, 233 (1991).
- [26] J.S. Lemos and P.M. Sa, Phys. Rev. D 49, 2897 (1994).
- [27] S. Mignemi, Phys. Rev. D 50, 4733 (1994).
- [28] A. Fabbri and J.G. Russo, Phys. Rev. D 53, 6995 (1996).
- [29] M.O. Katanaev, W. Kummer and H. Liebl, Phys. Rev. D 53, 5609 (1996);
 M.O. Katanaev, W. Kummer and H. Liebl, Nucl. Phys. B 486, 353 (1997);
 J.D. Christensen and R.B. Mann, Class. Quant. Grav. 9, 1769 (1992).

- [30] R.B. Mann and T. Ohta, Phys. Rev. D 55, 4723 (1997) and Class. Quant. Grav. 14, 1529 (1997).
- [31] J.S.F. Chan and R.B. Mann, Phys. Rev. D 50, 7376 (1994).
- [32] V. Mukhanov, A. Wipf, and A. Zelnikov, Phys. Lett. B 332, 283 (1994).
- [33] R. Bousso and S. W. Hawking, Phys. Rev. D 56, 7788 (1997).
- [34] W. Kummer, H. Liebl and D.V. Vassilevich, Mod. Phys. Lett. A 12, 2683 (1997).
- [35] T. Chiba and M. Siino, Mod. Phys. Lett. A 12, 709 (1997); S. Ichinose, Phys. Rev. D 57, 6224 (1998).
- [36] W. Kummer, H. Liebl, D. V. Vassilevich, Phys. Rev. D 58, 108501 (1998);
 W. Kummer and D. V. Vassilevich, Annalen Phys. 8, 801 (1999), and Phys. Rev. D 60, 084021 (1999).
- [37] R. Balbinot and A. Fabbri, Phys. Rev. D 59, 044031 (1999), and Phys. Lett.
 B 459, 112 (1999).
- [38] S. Nojiri and S. D. Odintsov, Mod. Phys. Lett. A 12, 2083 (1997), and Phys. Rev. D 57, 2363 (1998); A. Mikovic and V. Radovanovic, Class. Quant. Grav. 15, 827 (1998); F. C. Lombardo, F. D. Mazzitelli, and J. G. Russo, Phys. Rev. D 59, 064007 (1999).
- [39] S. Nojiri and S. D. Odintsov, Phys. Lett. B 463, 57 (1999).
- [40] E. Kleist and H. Moi, Genitourinary Medicine 69, 322 (1993).
- [41] B. S. DeWitt, Dynamical Theory of Groups and Fields (Gordon and Breach, New York, 1965).

- [42] B. S. DeWitt, Phys. Rep. C 19, 297 (1975).
- [43] F. A. Berezin, The Method of Second Quantization (Academic Press, New York, 1966).
- [44] N. D. Birrell and P. C. W. Davies, Quantum fields in curved space (Cambridge University Press, Cambridge, 1982).
- [45] R. M. Wald, Quantum Field Theory in Curved Spacetime and Black Hole Thermodynamics (University of Chicago Press, Chicago, 1994).
- [46] G. W. Gibbons and S. W. Hawking, Phys. Rev. D 15, 2752 (1977).
- [47] D. Bailin and A. Love, Introduction to Gauge Field Theory (Institute of Physics Publishing, Bristol, 1993).
- [48] J. Schwinger, Proc. Nat. Acad. Sci. USA 37, 452 (1951).
- [49] P. Candelas and D. J. Raine, Phys. Rev. D 15, 1494 (1977).
- [50] T. S. Bunch and L. Parker, Phys. Rev. D 20, 2499 (1979).
- [51] J. Schwinger, Phys. Rev. 82, 664 (1951).
- [52] R. Wald, Commun. Math. Phys. 70, 221 (1979).
- [53] See, for example, J. I. Kapusta, Finite Temperature Field Theory (Cambridge University Press, Cambridge, 1989); M. LeBellac, Thermal Field Theory (Cambridge University Press, Cambridge, 1996); A. Das, Finite Temperature Field Theory (World Scientific, 1997).
- [54] T. Matsubara, Prog. Theor. Phys. 14, 351 (1955).
- [55] C. Bloch, Nuc. Phys. 7, 451 (1958).

- [56] R. Kubo, J. Phys. Soc. Japan 12, 570 (1957).
- [57] P. Martin and J. Schwinger, Phys. Rev. 115, 1342 (1959).
- [58] R. Camporesi, Phys. Rep. 196, 1 (1990).
- [59] For a detailed account of dimensional reduction in non-relativistic quantum mechanics, see S. Tanimura and T. Iwai, J. Math. Phys. 41, 1814 (2000).
- [60] See, for example, Chapter 10, Lemma 6.3 of S. Helgason, Differential Geometry and Symmetric Spaces (Academic Press, New York, 1962).
- [61] K.V. Kuchař, J. D. Romano and M. Varadarajan, Phys. Rev. D 55, 795 (1997); D. Cangemi and R. Jackiw, Phys. Lett. B 337, 271 (1994); D. Cangemi and R. Jackiw and D. Zwiebach, Ann. Phys. 245, 408 (1996).
- [62] R.B. Mann, Phys. Lett. B 294, 310 (1994).
- [63] D. Grumiller, W. Kummer, and D. V. Vassilevich, Nucl. Phys. B 580, 438 (2000).
- [64] D. N. Page, Phys. Rev. D 13, 198 (1976), and *ibid* 14, 3260 (1976); W. A. Hiscock and L. Weems, Phys. Rev. D 41, 1142 (1990); J. M. Bardeen, Phys. Rev. Lett. 46, 382 (1981); R. Balbinot, Class. Quantum Grav. 1, 573 (1984); J. W. York, Jr., Phys. Rev. D 31, 775 (1985); V. P. Frolov and G. A. Vilkovisky, Phys. Lett. B 106, 307 (1981), and in *Proceedings of the Second Marcel Grossmann Meeting on General Relativity*, edited by R. Ruffini (North Holland, Amsterdam, 1982), p.455.
- [65] See, for example, S. P. De Alwis and D. A. MacIntire, Phys. Lett. B 344, 110 (1995); L. Thorlacius, Nucl. Phys. Proc. Suppl. 41, 245 (1995). T. Banks, Nucl. Phys. Proc. Suppl. 41, 21 (1995); P. T. Landsberg and R. B. Mann, Class. Quant. Grav. 10, 2373 (1993).

- [66] S.W. Hawking, Commun. Math. Phys. 43, 199 (1975).
- [67] W.G. Unruh, Phys. Rev. D 14, 870 (1976).
- [68] S.M. Christensen and S.A. Fulling, Phys. Rev. D 15, 2083 (1977); P. Candelas, Phys. Rev. D 21, 2185 (1980).
- [69] E. Elizalde, L. Vanzo, and S. Zerbini, Comm. Math. Phys. 194, 613 (1998).
- [70] J. S. Dowker, On the relevance of the multiplicative anomaly, hepth/9803200.
- [71] T. S. Evans, Phys. Lett. B 457, 127 (1999).
- [72] C. W. Misner, K. S. Thorne, and J. A. Wheeler, *Gravitation* (W. H. Freeman, San Francisco, 1973).
- [73] R. Balbinot, A. Fabbri, and V. P. Frolov, private communication.
- [74] A. P. Prudnikov, Yu. A. Brychkov, and O. I. Marichev, Integrals and Series, Vol. 1 Elementary Functions, and Vol. 2 Special Functions (Gordon and Breach, London, 1986).
- [75] See, for example, the text by Birrell and Davies [44]. Note, however, that the argument used by Birrell and Davies to show that the stress tensor will not diverge for the special case of a conformally invariant field with plane boundary does not hold for our dimensionally reduced theory because the corresponding stress tensor is not conserved [see equation (2.97) of Section 4.39].
- [76] I. S. Gradshteyn and I. M. Ryzhik, Tables of Integrals, Series, and Products (Academic, New York, 1994).
- [77] J. L. Synge, *Relativity: the general theory* (North Holland, Amsterdam, 1971).

- [78] J. H. Van Vleck, Proc. Nat. Acad. Sci. USA 14, 178 (1928).
- [79] C. Morette, Phys. Rev. 81, 848 (1951).
- [80] S. M. Christensen, Phys. Rev. D 14, 2490 (1976).
- [81] S. M. Christensen, Phys. Rev. D 17, 946 (1978).
- [82] J. B. Hartle and S. W. Hawking, Phys. Rev. D 13, 2188 (1976).
- [83] R. Balbinot, A. Fabbri, and P. Nicolini (unpublished).
- [84] W. Rindler, Am. J. Phys. 34, 1174 (1966).
- [85] S. A. Fulling, Phys. Rev. D 7, 2850 (1973); P. C. W. Davies, J. Phys. A 8, 609 (1975).
- [86] Th. Kaluza, Sitzungsber. d. Preuss. Akad. D. Wiss., 966 (1921); O. Klein,
 Zeitsch. f. Phys. 37, 895 (1926).
- [87] J. M. Overduin and P. S. Wesson, Phys. Rep. 283, 303 (1997).
- [88] M. B. Green, J. H. Schwarz and E. Witten, Superstring Theory. Vol. 1: Introduction, and Superstring Theory. Vol. 2: Loop Amplitudes, Anomalies And Phenomenology, (Cambridge University Press, Cambridge, 1987).
- [89] See, for example, M. J. Duff and D. J. Toms, Ref.TH.3248-CERN (1981).